ABSTRACT

| Title of Dissertation: | Cause and Effect of Threshold-Voltage Instability on the Reliability of Silicon-Carbide MOSFETs | | |
|------------------------|---|--|--|
| | Aivars Lelis, Doctor of Philosophy, 2011 | | |
| Directed by: | Professor Neil Goldsman | | |
| | Department of Electrical and Computer Engineering and associated with Reliability Engineering | | |

A significant instability of the threshold voltage (V_T) in silicon carbide (SiC) MOSFETs in response to gate-bias and *ON*-state current stressing was discovered and examined as a function of bias, temperature, and time. It was determined that the likely mechanism causing this effect is the charging and discharging of gate-oxide traps, located close to the interface of the SiC conducting channel, via a direct tunneling mechanism. High-temperature reverse-bias induced leakage current in the *OFF*-state was identified as a potential failure mode.

A simultaneous two-way tunneling model was developed, based on an existing one-way tunneling model, to simulate the time-dependent and field-dependent charging and discharging of the near-interfacial oxide traps in response to an applied gate-bias stress. The simulations successfully matched experimental results, both with respect to measurement time and to bias-stress time as a function of gate bias.

Experimental results were presented, showing that the V_T instability increases with both increasing gate-bias-stress time and bias-stress magnitude. The measurement conditions, including gate-ramp speed and direction, were shown to have a significant influence on the measured result, with a 20- μ s measurement revealing instabilities three times greater than those at standard 1-s measurement speeds, whereas 1-ks measurements showed shifts only half as large. High-temperature bias stressing was found to cause even more significant increases in the V_T instability. *ON*-state current stressing was found to also increase the V_T instability, due to self-heating effects.

 V_T shifts as large as 2 V were reported, with the number of calculated oxide traps switching charge state varying between 1×10^{11} and 8×10^{11} cm⁻², depending on processing, stress, and measurement conditions. The standard post-oxidation NO anneal was shown to reduce the number of active oxide traps by about 70 percent.

The dominant oxide trap was identified as an E'-center type defect—a weak Si-Si bond due to an oxygen vacancy which has been broken during processing or subsequent device stressing. The large increase in bias-stress induced V_T instability at temperatures above 100 °C was explained by an increase in the number of active E'-center type defects.

Existing reliability qualification standards based on silicon device technology are inadequate for SiC MOSFETs and need to be revised, with particular attention paid to the measurement conditions.

CAUSE AND EFFECT OF THRESHOLD-VOLTAGE INSTABILITY ON THE RELIABILITY OF SILICON CARBIDE MOSFETS

by

Aivars J. Lelis

Dissertation submitted to the Faculty of the Graduate School of the University of Maryland, College Park in partial fulfillment of the requirements for the degree of Doctor of Philosophy 2011

Advisory Committee:

Professor Neil Goldsman, Chair Professor Aris Christou Professor Patrick McCluskey Professor Pamela Abshire Professor Martin Peckerar ©Copyright by

Aivars J. Lelis

Dedication

Mammai un Tētim.

Acknowledgments

There are many people to thank to whom I am indebted:

First off, I would like to thank by advisor, Prof. Neil Goldsman, for all his support and encouragement over the past number of years.

This work would not have been possible at all without the strong programmatic support of Mr. Skip Scozzie at the U.S. Army Research Laboratory. I am also indebted to Skip and Dr. Jim McGarrity for recruiting me into the world of SiC in the first place. In more recent times I have also had the strong managerial support at ARL of Mr. Bruce Geil and Dr. Paul Barnes. I thank them all.

I also thank by original mentor, Dr. Tim Oldham, who along with Dr. F. Barry McLean and Mr. Ed Boesch from the old Harry Diamond radiation-effects team started me on my way in the study of charge traps in semiconductor devices. I have also benefited greatly from many discussions with Prof. Pat Lenahan at Penn State.

I am also indebted to Dr. John Suehle at NIST and his graduate student, Mr. Moshe Gurfinkel, for their invaluable assistance in performing the very fast *I-V* measurements.

But most of all, I am indebted to all the laboratory assistance at ARL, first from Mr. Gabriel Lopez, and more recently—and especially—from Mr. Dan Habersat and Dr. Ron Green. I also thank Dr. Lauren Boteler for all her help with formatting this document.

I also thank all my colleagues at ARL and elsewhere, along with my family and friends, for all their encouragement. And finally, a million thanks to my number-one fan, rab.

Table of Contents

| 1 | Intro | oduction | 1 |
|---------------|------------|--|----------|
| | 1.1 | Research goals and thesis outline | 1 |
| | 1.2 | Key contributions | 3 |
| | 1.3 | Why SiC MOSFETs? | 4 |
| 2 | Bac | ground | 8 |
| - | 2 1 | Review of the different types of interfacial charge | 0 |
| | 2.2 | Other work related to oxide charge trapping in SiC MOS | 12 |
| | 2.3 | Review of threshold-voltage instability in irradiated Si MOS | 15 |
| | 2.4 | Review of magnetic resonance work and the HDL oxide hole tran model | 17 |
| | 2.5 | Summary | 21 |
| 2 | Calc | wlation of Ovida Field and Thrashold Voltage Shift Caused by Interfacial | |
| $\frac{5}{C}$ | Calc | culation of Oxide Field and Threshold- voltage Shift Caused by Interfactar | าา |
| C | narge | Introduction: Calculation of algoritic notantial in the gate oxide and the SiC | LL |
| | J.I | Introduction. Calculation of electric potential in the gate oxide and the SIC | าา |
| | | Coloulation of the Elet hand voltage. V | 22 |
| | 5.2 2.2 | Calculation of V_{e} and V_{e} in the absence of interfacial charge | 23 |
| | 5.5 2.4 | Valuation of v_{SiC} and v_{OX} in the absence of interfactor charge | 20 |
| | 5.4 2.5 | Voltage shift due to interface traps | 29 |
| | 3.J | offect on the oxide field | , 24 |
| | | Effect of interfacial charge on threshold valtage and the gate avide field | 54 40 |
| | 3.0 2.7 | Effect of interfactal charge of uneshold voltage and the gate oxide field | 42 |
| | 5.7 2.9 | Numerical simulation of electric potential and field in the gate exide region f | 4/ |
| | 3.0 | Numerical simulation of electric potential and neight in the gate oxide region i | 52 |
| | | Summery of calculations | 50 |
| | 5.9 | Summary of calculations | 20 |
| 4 | Basi | c Experimental Results of Threshold-Voltage Instability Effect | 60 |
| | 4.1 | Introduction | 60 |
| | 4.2 | The device sample sets | 60 |
| | 4.3 | Experimental procedure | 64 |
| | 4.4 | Basic experimental effect of gate-bias stress induced threshold-voltage | |
| | instabi | lity | 67 |
| | 4.5 | Effect of bias-stress magnitude | 71 |
| | 4.6 | Effect of the Post-oxidation NO Anneal on V _T Instability | 78 |
| | 4.7 | One-way instability measurements | 80 |
| | 4.8 | Capacitor flatband-voltage instability | 83 |
| | 4.9 | Effect of elevated temperature on the V_T instability effect | 87 |
| | 4.10 | Summary | 95 |

| 5 | Mea | surement Time Dependence, Subthreshold Swing Analysis, and Charge | |
|---|----------|---|-------|
| S | eparatio | n | 98 |
| | 5.1 | Introduction | 98 |
| | 5.2 | Effect of measurement speed on the observed threshold-voltage instability | y due |
| | to gate | -bias stress | 99 |
| | 5.3 | Stretch-out of the subthreshold I_D - V_{GS} characteristics—a method for | |
| | determ | ining ΔN_{OT} | 104 |
| | 5.4 | Derivation of the subthreshold-swing analysis | 113 |
| | 5.5 | Experimental observation of variation in subthreshold swing versus | |
| | measur | rement speed | 123 |
| | 5.6 | Charge separation analysis | 126 |
| | 5.7 | Summary | 131 |
| 6 | Tun | neling Modeling | 133 |
| | 6.1 | Introduction | 133 |
| | 6.2 | Background | 134 |
| | 6.3 | Two-way tunneling model | 138 |
| | 6.4 | Steady-state conditions | 142 |
| | 6.5 | Tunneling time-step increments | 147 |
| | 6.6 | Tunneling model mechanics | 149 |
| | 6.7 | Comparison of model with experimental results | 154 |
| | 6.8 | Summary | 163 |
| 7 | Effe | ct of V _T Instability on SiC Power MOSFET Reliability | 165 |
| | 7.1 | Introduction | 165 |
| | 7.2 | Gate-bias stress-induced threshold-voltage instability in SiC power | |
| | DMOS | SFETs | 169 |
| | 7.3 | Effect of ON-state current stress on the threshold-voltage instability in Sid | С |
| | power | DMOSFETs | 172 |
| | 7.4 | High-temperature emulation of ON-state stress | 182 |
| | 7.5 | Degradation of ON-state characteristics | 188 |
| | 7.6 | High-temperature bias-stress mechanisms | 190 |
| | 7.7 | Reliability testing | 196 |
| | 7.8 | Summary | 201 |
| 8 | Sum | mary, Conclusions, and Future Work | 204 |
| | 8.1 | Summary and conclusions of present work | 204 |
| | 8.2 | Future work | 208 |
| 9 | Refe | erences | 211 |

List of Tables

| TABLE 3-1: TYPICAL VALUES USED FOR THE PARAMETERS OF (3-26) AND (3-27) | 32 |
|--|-------|
| TABLE 4-1: LATERAL TEST STRUCTURE SAMPLE SETS USED TO OBTAIN EXPERIMENTAL RESULTS PRESENTED IN CHAPTERS 4 AN | ID 5. |
| | 62 |

List of Figures

| FIGURE 2-1: A MODIFICATION OF THE DEAL [31] SI MOS FIGURE, INDICATING THE NAMES AND LOCATIONS OF THE VARIOUS |
|---|
| TYPES OF INTERFACIAL CHARGE FOR SIC MOS |
| FIGURE 2-2: HDL (ARL) HOLE TRAP MODEL [88] |
| FIGURE 3-1: SCHEMATIC OF THE ENERGY BANDS FOR, FROM LEFT TO RIGHT, THE <i>P</i> -TYPE POLY-SI GATE, THE GATE OXIDE, AND |
| THE <i>P</i> -TYPE SIC |
| FIGURE 3-2: SURFACE ELECTRIC FIELD FOR THE SIC AS A FUNCTION OF THE SIC SURFACE POTENTIAL, IN UNITS OF THE BULK |
| POTENTIAL, ϕ_B |
| FIGURE 3-3: INTERFACE TRAP DENSITY VERSUS BANDGAP ENERGY, CALCULATED USING (3-26), (3-27), AND TABLE 3-133 |
| FIGURE 3-4: VARIATION OF INTERFACE TRAP CHARGE WITH SURFACE POTENTIAL, PLOTTED IN UNITS OF THE BULK POTENTIAL33 |
| FIGURE 3-5: SCHEMATIC OF A GENERAL CHARGE DISTRIBUTION IN THE GATE OXIDE |
| FIGURE 3-6: SCHEMATIC OF A RECTANGULAR CHARGE DISTRIBUTION IN THE GATE OXIDE NEAR THE SIC INTERFACE |
| FIGURE 3-7: VOLTAGE DROP ACROSS THE OXIDE DUE TO Q_s |
| FIGURE 3-8: VOLTAGE DROP ACROSS THE OXIDE IS MODIFIED BY A RECTANGULAR OXIDE TRAP DISTRIBUTION AS IN FIGURE 3-6, |
| WHICH CHANGES THE BULK FIELD |
| FIGURE 3-9: THE VOLTAGE DROP ACROSS THE OXIDE IS MODIFIED BY THE HATCHED AREA DUE TO THE PRESENCE OF THE |
| RECTANGULAR DISTRIBUTION OF OXIDE TRAPS NEAR THE INTERFACE AS DEPICTED IN FIGURE 3-6 |
| FIGURE 3-10: VARIATION IN REQUIRED APPLIED GATE-TO-SOURCE VOLTAGE TO EFFECT A PARTICULAR SURFACE POTENTIAL IN A |
| SIC MOSFET ($N_A = 1 \times 10^{16}$ cm ⁻³ , $T_{ox} = 500$ Å) FOR BOTH AN IDEAL DEVICE WITH NO TRAPPED CHARGE AND ONE WITH |
| AN OXIDE TRAPPED CHARGE DENSITY OF +1×10 ¹² CM ⁻² 44 |
| Figure 3-11: Comparison of the ideal relationship between V_{GS} and gate oxide field, without interfacial charge |
| PRESENT, AND A SIMPLER RULE-OF-THUMB CALCULATION45 |
| FIGURE 3-12: VARIATION IN REQUIRED APPLIED GATE-TO-SOURCE VOLTAGE TO EFFECT A PARTICULAR OXIDE FIELD IN A SIC |
| MOSFET ($N_A = 1 \times 10^{10}$ cm ⁻³ , $T_{ox} = 500$ Å) FOR BOTH AN IDEAL DEVICE WITH NO TRAPPED CHARGE, WHERE THE SURFACE |
| AND BULK FIELDS ARE THE SAME; AND ONE WITH AN OXIDE TRAPPED CHARGE DENSITY OF $\pm 1\times 10^{12}$ cm $^{-2}$, where the |
| SURFACE AND BULK FIELDS ARE DIFFERENT |
| FIGURE 3-13: SPATIAL VARIATION OF ELECTRIC POTENTIAL IN BOTH GATE OXIDE AND SIC, AS A FUNCTION OF SURFACE |
| POTENTIAL FOR THE FOLLOWING CONDITIONS: ACCUMULATION (ψ_s = -0.1· ϕ_B), FLAT-BAND (ψ_s = 0), ZERO POTENTIAL |
| $(V_{GS} = 0; \psi_s = 0.57 \cdot \phi_B)$, MID-GAP $(\psi_s = \phi_B)$, INVERSION $(\psi_s = 2 \cdot \phi_B)$, AND STRONG INVERSION $(\psi_s = 2.1 \cdot \phi_B)$; FOR $N_A = 0.57 \cdot \phi_B$ |
| 1×10^{16} and τ_{ox} =50 nm |
| Figure 3-14: Variation of oxide trap distribution versus decay parameter λ , for equal total charge over first |
| 50 Å FROM THE INTERFACE54 |
| FIGURE 3-15: SCHEMATIC OF HOW THE ELECTRIC FIELD VALUES ARE DERIVED FROM THE ELECTRIC POTENTIAL, AND HOW THE |
| CHARGE DENSITY VALUES ARE CONFIRMED BY THE ELECTRIC FIELD VALUES |
| FIGURE 4-1: IMAGE OF THE LATERAL 4H SIC MOS TEST CHIP, THE MAIN TEST VEHICLE FOR THIS STUDY |
| FIGURE 4-2: LATERAL 4H SIC MOS TEST CHIP LAYOUT WITH MAIN STRUCTURES OF INTEREST HIGHLIGHTED |
| FIGURE 4-3: SCHEMATIC INDICATING THE APPLIED GATE VOLTAGE VERSUS TIME DURING THE INSTABILITY MEASUREMENTS65 |
| FIGURE 4-4: SCHEMATIC OF FAST I-V SYSTEM EMPLOYED AT NIST |
| FIGURE 4-5: GRAPH OF I_D - V_{GS} curves of a SIC MOSFET from Sample Set D with a 130-NM thick deposited gate |
| OXIDE, PLOTTED ON BOTH A LOG (LEFT) AND LINEAR (RIGHT) CURRENT SCALE, FOR A SERIES OF TWELVE FULL CYCLES OF |
| BIAS STRESS (±10 V FOR 1,000 SECONDS AT EACH POLARITY) AND MEASUREMENT AS INDICATED IN FIGURE 4-4. |
| POSITIVE-BIAS STRESS CAUSES A SHIFT OF THE I-V CHARACTERISTIC TO THE RIGHT AND A NEGATIVE-BIAS STRESS CAUSES A |
| SHIFT OF THE I-V CHARACTERISTIC BACK TO THE LEFT |
| FIGURE 4-6: CALCULATED V _T INSTABILITY OF A SIC MOSFET FROM SAMPLE SET D USING BOTH LINEAR EXTRAPOLATION AND |
| INVERSION CURRENT METHODS FOLLOWING EACH BIAS STRESS AND MEASUREMENT VERSUS CUMULATIVE STRESS TIME 69 |
| FIGURE 4-7: CALCULATED CHANGE IN V_T of a SIC MOSFET from Sample Set D using both linear extrapolation and |
| INVERSION CURRENT METHODS FOLLOWING EACH BIAS STRESS AND MEASUREMENT VERSUS CUMULATIVE STRESS TIME. 70 |
| Figure 4-8: Variation in the back-and-forth $V_{	au}$ instability of a SiC MOSFET from Sample Set A, depending on |
| WHETHER THE GATE BIAS IS SWEPT DOWN OR UP FOLLOWING A POSITIVE-BIAS STRESS. |
| |

| Figure 4-9: Change in $V_{	au}$ of a SiC MOSFET from Sample Set D with a 130-nm thick deposited gate oxide, |
|--|
| CALCULATED USING INVERSION CURRENT METHOD, VERSUS LOG STRESS TIME FOR VARIOUS APPLIED GATE BIASES FOR A SIC MOSFET WITH A THICK DEPOSITED GATE OXIDE |
| FIGURE 4-10: DATA FROM FIGURE 4-9 RE-PLOTTED SHOWING VARIATION IN V- OF A SIC MOSEFT FROM SAMPLE SET D USING |
| INVERSION CURRENT METHOD AS A FUNCTION OF BOTH BIAS MAGNITUDE AND POLARITY. THIS DEVICE IS OLDER WITH A |
| |
| FIGURE 4-11: VARIATION IN V-OF A SIC MOSEFT FROM SAMPLE SET I WITH A NEWER THERMAL OVIDE LISING INVERSION |
| CURRENT METHOD AS A FUNCTION OF DOTH DIAS MACHITUDE AND DOLARITY. THIS DEVICES IS MODE DECENT. WITH A |
| LESS THICK THERMAL OXIDE |
| FIGURE 4-12: GRAPH OF I_D - V_{GS} curves of a lateral 4H-SiC MOSFET from Sample Set F with a 60-NM thick thermal |
| OXIDE WITH AN NO POST-OXIDATION ANNEAL, PLOTTED ON BOTH A LOG (LEFT) AND LINEAR (RIGHT) CURRENT SCALE, FOR |
| THREE FULL CYCLES OF BIAS STRESS AND MEASUREMENT (±2 MV/cm, 34-KS STRESS, 1-S RAMP) |
| FIGURE 4-13: AVERAGE MEASURED V_T INSTABILITY OF A LATERAL 4H-SIC MOSFET FROM SAMPLE SET F WITH A 60-NM THICK |
| THERMAL OXIDE WITH AN NO POST-OXIDATION ANNEAL FOR VARIOUS BIAS STRESSES AND TIMES, AND 1-S |
| MEASUREMENT TIME |
| Figure 4-14: Average measured V_{τ} instability of variously processed lateral 4H-SiC MOSFET with 50-nm thick |
| THERMAL OXIDES FROM SAMPLE SET G. THE UNIMPLANTED DEVICE WITH AN NO POST-OXIDATION ANNEAL SHOWED THE |
| SMALLEST EFFECT WHILE THAT WITHOUT AN NO ANNEAL HAD THE LARGEST EFFECT. WHEN STRESSED WITH A GATE OXIDE |
| FIELD OF + 3 MV/cm. |
| FIGURE 4-15: In-Vision Characteristics depicting a ONE-way V-INSTABILITY OF A LATERAL 4H-SIC MOSEET with 50-NM |
| THICK THERMAL OXIDE AND POST-OXIDATION NO ANNEAL FROM SAMPLE SET G |
| FIGURE 4-16: PLOT OF LINEAR-WITH-LOG-TIME ONE-WAY V-INSTABILITY TAKEN FROM FIGURE 4-15 FOR AN N-CHANNEL |
| MOSEET FROM SAMPLE SET G |
| FIGURE 4-17: BACK-AND-FORTH INSTABILITY IN THE C-V CHARACTERISTICS OF A SIC MOS CAPACITOR FROM SAMPLE SET C |
| DUE TO BACK AND FORTH INSTABLETT IN THE C V CHARACTERISTICS OF A SIC MOS CALACTOR HOM SAME LE SET C |
| FIGURE A-18: COMPARISON OF THE CHANGE IN ELAT-BAND CARACITANCE VERSUS LOG STRESS TIME FOR SIC MOS CARACITORS |
| WITH A DEDOCITED AND THE PAND OVIDE WITH THE SAME PLAS STRESS CONDITIONS (-15V AND ±10 V) |
| FIGURE A-10. VARIATION IN ELAT-RAND VOLTAGE FOR A SIC MOS CARACITOR WITH A DEPOSITED OVIDE FROM SAMPLE SET C |
| FIGURE 4-15. VARIATION IN FLAT-BAND VOLTAGE FOR A SIC IVIOS CAPACITOR WITH A DEPOSITED OXIDE FROM SAMPLE SET C |
| DUE TO BACK-AND-FORTH GATE-BIAS STRESSING (-ISV AND +10 V) |
| FIGURE 4-20. VARIATION OF THRESHOLD VOLTAGE FOLLOWING GATE BIAS STRESSING AS A FUNCTION OF TEMPERATURE FOR AN |
| ENCLOSED GEOMETRY 4H-SIC LATERAL MOSTET FROM SAMPLE SET E |
| FIGURE 4-21. INSTABILITY OF I_D -V _{GS} CHARACTERISTICS AT BOTH 25 AND 200°C FOR SAMPLE SET E FOLLOWING BIAS STRESS |
| AND MEASUREMENT SEQUENCE SHOWN ON THE RIGHT |
| FIGURE 4-22: AVERAGE THRESHOLD-VOLTAGE INSTABILITY EFFECT AS A FUNCTION OF TEMPERATURE FOR SEVERAL DIFFERENT |
| SAMPLES FAIRLY REPRESENTATIVE OF THE STATE-OF-THE-ART. ALL DEVICES HAVE THERMALLY GROWN OXIDES THAT |
| RECEIVED A STANDARD POST-OXIDATION NO ANNEAL |
| FIGURE 4-23: INSTABILITY OF I _D -V _{GS} CHARACTERISTICS AT BOTH 25 AND 150 °C FOR SAMPLE SET H. THE RESPONSE AT |
| ELEVATED TEMPERATURE IS THE OPPOSITE OF THAT AT ROOM TEMPERATURE AND IS LIKELY DUE TO MOBILE ION DRIFT92 |
| FIGURE 4-24: AVG. INSTABILITY OF SAMPLE SET F AS A FUNCTION OF TEMPERATURE AND BIAS-STRESS TIME |
| FIGURE 4-25: SCHEMATIC DEMONSTRATING THE OPPOSING EFFECTS OF BIAS STRESS ON CHARGE TRAPS AND MOBILE IONS. FOR |
| EXAMPLE, A POSITIVE GATE-BIAS STRESS WILL CAUSE A POSITIVE SHIFT IN THE THRESHOLD VOLTAGE DUE TO INTERFACIAL |
| CHARGE TRAPPING, BUT WILL CAUSE A NEGATIVE SHIFT IN V_7 due to mobile ion drift |
| FIGURE 5-1: LINEAR-WITH-LOG-STRESS-TIME EXTRAPOLATION OF THE BIAS STRESS-INDUCED THRESHOLD-VOLTAGE INSTABILITY |
| DATA VERSUS INDIVIDUAL STRESS TIMES TO EARLY TIMES AND ZERO INSTABILITY OF A LATERAL 4H-SIC MOSFET FROM |
| Sample Set D with a $130\text{-}\mathrm{nm}$ thick deposited gate oxide with a nitrogen anneal. Selective data from |
| Figure 4-9 was used101 |
| FIGURE 5-2: LINEAR-WITH-LOG-TIME EXTRAPOLATION OF THE BIAS STRESS-INDUCED THRESHOLD-VOLTAGE INSTABILITY DATA TO |
| early stress times and zero instability of a lateral 4H-SiC MOSFET from Sample Set F with a 60-nm thick |
| THERMAL OXIDE WITH AN NO POST-OXIDATION ANNEAL104 |
| Figure 5-3: Close-up of the subthreshold region of the I_D - V_{GS} characteristic from Figure 4-5 (Sample Set D). |
| CLEARLY, THE CURVE IS MORE STRETCHED OUT WHEN SWEEPING UP IN GATE BIAS FOLLOWING A NEGATIVE-BIAS STRESS, |
| LEADING TO A GREATER VALUE OF S, THE SUBTHRESHOLD SWING105 |

| FIGURE 5-4: SCHEMATIC OF THE SUBTHRESHOLD SWING ANALYSIS TECHNIQUE, COMPARING THE MEASURED SUBTHRESHOLD |
|--|
| SWING WHEN THE GATE IS SWEPT UP IN BIAS FOLLOWING A NEGATIVE-BIAS STRESS WITH THE SUBTHRESHOLD SWING |
| WHEN THE GATE IS SWEPT DOWN IN BIAS FOLLOWING A POSITIVE-BIAS STRESS |
| FIGURE 5-5: COMPARISON OF THE EFFECT OF MEASUREMENT TIME ON THE CALCULATED INSTABILITY OF THE THRESHOLD |
| VOLTAGE FOR 4H SIC MOSFETS FROM SAMPLE SET F WITH AN NO ANNEAL (MODIFICATION OF FIGURE 5-2). THE 1-S |
| MEASUREMENTS (LOWER RIGHT-HAND CORNER) HAD ADEQUATE SUBTHRESHOLD DATA TO CALCULATE THE CHANGE IN |
| MID-GAP VOLTAGE, WHICH PROVIDES A BETTER (AND HIGHER) ESTIMATE OF A LOWER BOUND FOR THE NUMBER OF |
| SWITCHING OXIDE TRAPS, COMPARABLE TO THAT FROM THE FAST I-V MEASUREMENTS. |
| Figure 5-6: Comparison of mid-gap and inversion voltage methods for calculating V_T instability (modification |
| of Figure 4-7 using devices from Sample Set D). Note the noise in the mid-gap calculation due to the |
| LARGE EXTRAPOLATION |
| Figure 5-7: Calculated V_T using the inversion current method from Figure 4-6 (Sample Set D), along with the |
| calculated subthreshold swing value, S, plotted on the seconday y-axis. When V_T shifts more negatively |
| following a negative-bias stress, the subthreshold I - V characteristic is also more stretched out (larger S |
| VALUE) |
| FIGURE 5-8: CHANGES IN CALCULATED CHARGE TRAP DENSITY WHEN SUBJECT TO ALTERNATING POSITIVE AND NEGATIVE GATE- |
| bias stresses using the data from Figure 5-7 (Sample Set D). The N_{OT} calculated varies depending on |
| WHETHER THE CHANGE IN INVERSION VOLTAGE OR MID-GAP VOLTAGE IS USED, AND THE $N_{ m IT}$ CALCULATED IS DISTORTED BY |
| THE CHANGE IN THE SLOPE OF THE I - V characteristic due to the change in oxide trap occupation during the |
| MEASUREMENT |
| FIGURE 5-9: SCHEMATIC COMPARING THE IDEAL SUBTHRESHOLD SWING WITH AN ACTUAL MEASURED SUBTHRESHOLD SWING. |
| |
| FIGURE 5-10: VARIATION IN THE SUBTHRESHOLD <i>I-V</i> CHARACTERISTICS OF A 4H-SIC MOSFET FROM SAMPLE SET I DUE TO A |
| BACK-AND-FORTH GATE-BIAS STRESSING SEQUENCE AS A FUNCTION OF MEASUREMENT SPEED. THE WIDTH OF THE |
| ENVELOPE DECREASES WITH INCREASING MEASUREMENT SPEED, AS DOES THE STRETCH-OUT OF THE SUBTHRESHOLD $I\text{-}V$ |
| CHARACTERISTICS |
| FIGURE 5-11: VARIATION IN THE THRESHOLD VOLTAGE, CALCULATED USING THE INVERSION CURRENT METHOD AND THE DATA |
| SHOWN IN FIGURE 5-10, AS A FUNCTION OF MEASUREMENT TIME WHEN SWEEPING UP FOLLOWING A NEGATIVE-BIAS |
| STRESS AND SWEEPING DOWN FOLLOWING A POSITIVE-BIAS STRESS |
| FIGURE 5-12: VARIATION IN THE SUBTHRESHOLD SWING, S, CALCULATED USING THE DATA SHOWN IN FIGURE 5-10, AS A |
| FUNCTION OF MEASUREMENT TIME WHEN SWEEPING UP FOLLOWING A NEGATIVE-BIAS STRESS AND SWEEPING DOWN |
| FOLLOWING A POSITIVE-BIAS STRESS |
| FIGURE 6-1: THE SHARP EXPONENTIAL DECAY IN THE TUNNELING PROBABILITY WITH DISTANCE INTO THE OXIDE, COUPLED WITH |
| THE RELATIVELY SHARP DISTRIBUTION OF OXIDE TRAPS HERETOFORE UNAFFECTED BY THE TUNNELING AND THEREFORE YET |
| TO CHANGE CHARGE STATE, RESULTS IN A SHARP PEAK IN THE TUNNELING TRANSITION RATE, ILLUSTRATING THE CONCEPT |
| OF A "TUNNELING FRONT" WHICH MOVES DEEPER INTO THE OXIDE AT A LINEAR-WITH-LOG-TIME RATE |
| FIGURE 6-2: SCHEMATIC OF HOLES PREDOMINANTLY TUNNELING IN UNDER NEGATIVE BIAS TO UNCOVER POSITIVE CHARGE AND |
| ELECTRONS PREDOMINANTLY TUNNELING IN UNDER POSITIVE BIAS TO NEUTRALIZE THE POSITIVE CHARGE |
| FIGURE 6-3: SCHEMATIC OF POTENTIAL ENERGY LEVEL OF THE OXIDE TRAPS, AND THE EFFECTS OF BAND BENDING, BASED ON |
| EXPERIMENTAL RESULTS |
| FIGURE 6-4: CALCULATION OF THE STEADY-STATE FRACTION OF NEUTRALIZED (FILLED WITH ELECTRONS) OXIDE TRAPS VERSUS |
| OXIDE DEPTH AS A FUNCTION OF CONSTANT OXIDE-FIELD VALUES FOR THE FOLLOWING PARAMETER VALUES: $E_{T_{e}} = 2.7$ |
| EV, $E_{\tau_{-H}} = 6.3$ EV, and $M_{H}^{*} = 0.162$ ($M_{\epsilon}^{*} = 0.42$ and $\alpha_{\epsilon} = \alpha_{H}$) |
| FIGURE 6-5: CALCULATION OF THE STEADY-STATE FRACTION OF NEUTRALIZED (FILLED WITH ELECTRONS) OXIDE TRAPS VERSUS |
| OXIDE DEPTH AS A FUNCTION OF CONSTANT OXIDE-FIELD VALUES FOR THE FOLLOWING PARAMETER VALUES: $E_{T_{e}} = 6.0$ |
| EV, $E_{\tau_{-H}} = 3.0$ EV, and $M_{H}^{*} = 0.885$ ($M_{\epsilon}^{*} = 0.42$ and $\alpha_{\epsilon} = \alpha_{H}$) |
| FIGURE 6-6: CALCULATION OF THE STEADY-STATE FRACTION OF NEUTRALIZED (FILLED WITH ELECTRONS) OXIDE TRAPS VERSUS |
| Oxide depth as a function of constant oxide-field values for the following parameter values: $E_{	au_{-}arepsilon}$ = 6.0 |
| EV, $E_{\tau_{-}\mu} = 6.3$ EV, and $M_{\mu}^* = 0.377$ ($M_{\epsilon}^* = 0.42$ and $\alpha_{\epsilon} = \alpha_{\mu}$) |
| Figure 6-7: Calculation of the variation in net charge tunneling into oxide traps during a time increment of T_0 |
| SECONDS AS A FUNCTION OF TOTAL TUNNELING TIME |
| FIGURE 6-8: SCHEMATIC OF TUNNELING MODEL ALGORITHM |

| FIGURE 6-9: INITIAL VARIATIONS IN THE NORMALIZED POSITIVELY CHARGED OXIDE-TRAP DISTRIBUTION VS OXIDE DEPTH AS A |
|--|
| FUNCTION OF GATE BIAS DURING BOTH A 1-S MEASUREMENT SWEEPING UP IN GATE BIAS AND A SUBSEQUENT $+15 \text{ V}$ |
| FIGURE 6-10: SUBSCIENT VARIATIONS IN THE NORMALIZED DOSITIVELY CHARGED OVIDE-TRAD DISTRIBUTION VS OVIDE DEDTH |
| AS A FUNCTION OF GATE BIAS FOLLOWING (1) AN INITIAL +15 V BIAS STRESS, (2) A 1-S MEASUREMENTS SWEEPING |
| down in gate bias, and (3) a final -15 V 10° -s bias stress152 |
| FIGURE 6-11: FURTHER SUBSEQUENT VARIATIONS IN THE NORMALIZED POSITIVELY CHARGED OXIDE-TRAP DISTRIBUTION VS |
| OXIDE DEPTH AS A FUNCTION OF GATE BIAS FOLLOWING (1) AN INITIAL –15 V BIAS STRESS, (2) A 1-S MEASUREMENTS |
| SWEEPING UP IN GATE BIAS, AND (3) A FINAL +15 V 10° -s bias stress153 |
| FIGURE 6-12: VARIATIONS IN THE CHANGE IN OXIDE FIELD VS OXIDE DEPTH DUE TO THE VARIATIONS IN POSITIVELY CHARGE |
| OXIDE-TRAP DISTRIBUTION AS SHOWN IN FIGURE 6-11 |
| FIGURE 6-13: VARIATION IN V _{OT} , THE VOLTAGE SHIFT DUE TO THE PRESENCE OF OXIDE TRAPPED CHARGE, IS PLOTTED VERSUS |
| INDIVIDUAL BIAS STRESS SEQUENCES SIMULATING BOTH THE LONG-TERM BIAS STRESSES, AS WELL AS THE SHORTER BIAS |
| STRESSES THAT COMPRISE THE MEASUREMENT RAMP FOR SEVERAL DIFFERENT MEASUREMENT TIMES |
| FIGURE 6-14: SIMULATED PLOTS OF SURFACE POTENTIAL (IN TERMS OF THE BULK POTENTIAL) VERSUS GATE BIAS ARE SIMILAR IN |
| FORM TO PLOTS OF LOG(I_D) VS V_{GS} . MODEL CALCULATIONS PREDICT LARGER INSTABILITIES FOR FASTER MEASUREMENT |
| RAMP SPEEDS, MATCHING EXPERIMENTAL |
| FIGURE 6-15: PLOT OF THE NORMALIZED BACK-AND-FORTH THRESHOLD-VOLTAGE INSTABILITY VERSUS MEASUREMENT RAMP |
| SPEED, COMPARING MODEL CALCULATIONS AND EXPERIMENTAL RESULTS. |
| FIGURE 6-16: PLOT OF THE CHANGES IN V _{OT} , THE VOLTAGE SHIFT DUE TO THE PRESENCE OF OXIDE TRAPPED CHARGE, VERSUS |
| BOTH POSITIVE AND NEGATIVE-BIAS STRESS MAGNITUDE WITH A SIMULATED MEASUREMENT SPEED OF 1 S |
| FIGURE 6-17: SCHEMATIC INDICATING SEVERAL POSSIBLE RESPONSES OF THE DRAIN CURRENT VERSUS GATE VOLTAGE IN THE |
| SUBTHRESHOLD REGION, DEPENDING ON THE SPEED OF THE MEASUREMENT AND THE RESPONSE TIME OF THE OXIDE |
| TRAPS |
| FIGURE 6-18: SCHEMATIC SHOWING A COMPARISON OF THE BAND GAPS OF SI AND 4H-SIC AND HOW OXIDE TRAPS THAT FALL |
| BELOW THE VALENCE BAND EDGE OF SI MAY ACTUALLY BE WITHIN THE BANDGAP OF SIC |
| REGION [11] |
| Figure 7-2: Illustration of potential reliability issues due to V_7 instability effect. Subthreshold leakage current |
| MAY INCREASE DRAMATICALLY IN THE <i>OFF</i> STATE IF THE THRESHOLD VOLTAGE IS NOT SET PROPERLY |
| FIGURE 7-3: SCHEMATIC OF THE TEST-AND-MEASUREMENT SEQUENCE USED FOR THE ROOM-TEMPERATURE GATE-BIAS STRESS. |
| FIGURE 7-4: COMPARISON OF THE AVERAGE THRESHOLD-VOLTAGE INSTABILITY OF SEVERAL 5-A SIC DMOSFETS WITH THAT OF |
| LATERAL MOSFETS WITH AND WITHOUT AN IMPLANTED CHANNEL VS BIAS-STRESS TIME, CALCULATED BY FINDING THE |
| Shift in the linearly extrapolated $V_{	au}$ |
| FIGURE 7-5: AVERAGE THRESHOLD-VOLTAGE INSTABILITY FOR THE 20-A SIC DMOSFET VS STRESS TIME AS A FUNCTION OF |
| GATE-BIAS STRESS, CALCULATED BY FINDING THE VOLTAGE SHIFT FOR A CONSTANT CURRENT OF 90 $\mu A.$ 172 |
| FIGURE 7-6: STRESS-AND-MEASUREMENT SEQUENCE EMPLOYED TO INVESTIGATE THE EFFECT OF ON-STATE STRESSING ON THE |
| THRESHOLD-VOLTAGE INSTABILITY OF SIC POWER DMOSFETS (COMPARE WITH FIGURE 7-3 FOR THE GATE-BIAS STRESS- |
| AND-MEASUREMENT SEQUENCE) |
| FIGURE 7-7: COMPARISON OF V_7 INSTABILITY IN A DMOSFET DUE TO ON-STATE STRESSING, WITH 20 A FLOWING, AND THAT |
| DUE TO SIMPLE GATE-BIAS STRESSING, WITH V_{DS} = 0 V. IN EACH CASE V_{GS} = +15 V was applied during the first half |
| OF THE STRESS, AND A SIMPLE NEGATIVE GATE-BIAS STRESS WITH V_{GS} = -15 V and V_{DS} = 0 V was applied during the |
| SECOND HALF OF THE STRESS |
| FIGURE 7-8: COMPARISON OF ON-STATE STRESS INDUCED VOLTAGE INSTABILITY WHEN MEASURED USING THREE DIFFERENT |
| METHODS: FOR $I_D = 1 \times 10^{-5}$ A, FOR $I_D = 0.1$ A, and finding V_T by extrapolating the linear <i>I-V</i> curve to zero |
| CURRENT |
| FIGURE 7-9: GATE-BIAS STRESS SUBTHRESHOLD I_D - V_{GS} INSTABILITY CHARACTERISTICS OF A 50-A SIC POWER MOSFET AS THE |
| RESULT OF AN ON-STATE STRESS TEST. PRE- AND POST-ON-STATE-STRESS MEASUREMENTS ARE SHOWN AS WELL177 |
| FIGURE 7-10: GATE VOLTAGE REQUIRED FOR $I_D = 1 \times 10^{-5}$ A (with $V_{DS} = 0.1$ V) of a 20-A SIC DMOSFET during a |
| MEASUREMENT FOLLOWING BOTH POSITIVE AND NEGATIVE-BIAS STRESS: (A) BEFORE (PRE) AN ON -STATE STRESS. (B) |

| IMMEDIATELY AFTER (DURING) AN <i>ON-</i> STATE STRESS, AND (C) A DAY AFTER (POST) AN <i>ON-</i> STATE STRESS. THE |
|---|
| CUMULATIVE STRESS TIME IS PLOTTED ON A LINEAR SCALE |
| FIGURE 7-11: SUBTHRESHOLD INSTABILITY OF A 50-A SIC DMOSFET VERSUS VARIATION IN STRESS TIME FOR AN ON-STATE |
| STRESS, FOLLOWED IMMEDIATELY BY SEVERAL 10,000-S GATE-BIAS STRESS SEQUENCES PLOTTED VERSUS CUMULATIVE |
| STRESS TIME |
| FIGURE 7-12: SUBTHRESHOLD INSTABILITY OF VARIOUS SIC DMOSFETS DURING AN ON-STATE STRESS. TEMPERATURE |
| VARIATION IS DUE TO VARIATIONS IN SELF-HEATING, MEASURED USING AN IR THERMAL CAMERA |
| FIGURE 7-13: STRESS-AND-MEASUREMENT SEQUENCE EMPLOYED TO INVESTIGATE THE EFFECT OF AN EMULATED ON-STATE |
| STRESSING ON THE THRESHOLD-VOLTAGE INSTABILITY OF SIC POWER DMOSFETS (COMPARE WITH FIGURE 7-6)183 |
| FIGURE 7-14: EXAMPLE OF THE INSTABILITY IN THE I_D - V_{GS} CHARACTERISTICS OF A POWER MOSFET AS THE RESULT OF A HIGH- |
| TEMPERATURE-STRESS EMULATION OF AN ON-STATE STRESS TEST. PRE- AND POST-HIGH-TEMPERATURE-STRESS |
| MEASUREMENTS ARE SHOWN AS WELL |
| FIGURE 7-15: COMPARISON OF THE TOTAL BACK-AND-FORTH INSTABILITY OF THE I_D - V_{GS} CHARACTERISTICS VERSUS STRESS TIME |
| FOR A HIGH-TEMPERATURE-STRESS EMULATION OF THE ${\it ON}$ -STATE STRESS CALCULATED USING BOTH THE $V_{	au}$ and V_{GS} |
| ANALYSIS METHODS |
| FIGURE 7-16: COMPARISON OF THE SHIFT UNDER BOTH POSITIVE AND NEGATIVE-BIAS STRESS OF THE I_D - V_{GS} CHARACTERISTICS |
| VERSUS STRESS TIME FOR A HIGH-TEMPERATURE-STRESS EMULATION OF THE ON-STATE STRESS CALCULATED USING BOTH |
| The V_T and V_{GS} analysis methods |
| FIGURE 7-17: ON-STATE I_D - V_{DS} as a function of V_{GS} . The effect of the V_T instability would be to increase the ON- |
| STATE RESISTANCE ONLY BY A FEW PERCENT |
| FIGURE 7-18: A 1-HOUR <i>ON</i> -STATE STRESS LEADS TO AN INCREASE IN <i>ON</i> -STATE RESISTANCE. V_{DS} MUST INCREASE ABOUT 0.1 |
| V FOLLOWING THE STRESS TO MAINTAIN 20 A OF DRAIN CURRENT |
| FIGURE 7-19: SHORT STRESS-AND-MEASUREMENT INSTABILITY SEQUENCE TO DETERMINE EFFECT OF HIGH-TEMPERATURE |
| STRESS WITH OR WITHOUT APPLIED GATE-BIAS STRESS |
| FIGURE 7-20: SHORT STRESS-AND-MEASUREMENT INSTABILITY SEQUENCE TO DETERMINE EFFECT OF HIGH-TEMPERATURE |
| STRESS WITH OR WITHOUT APPLIED GATE-BIAS STRESS |
| FIGURE 7-21: SHORT STRESS-AND-MEASUREMENT INSTABILITY SEQUENCE TO DETERMINE EFFECT OF HIGH-TEMPERATURE |
| STRESS WITH OR WITHOUT APPLIED GATE-BIAS STRESS |
| FIGURE 7-22: SCHEMATIC SHOWING THE INCREASE IN THE NUMBER OF POSITIVELY CHARGED OXIDE TRAPS WITH TIME UNDER |
| NEGATIVE BIAS AT HIGH TEMPERATURE, DUE TO BOTH THE TUNNELING FRONT EXTENDING DEEPER INTO THE OXIDE AND |
| AN INCREASE IN THE NUMBER OF ACTIVE TRAPS196 |
| FIGURE 7-23: LARGE VARIATION IN THE DEVICE ROOM-TEMPERATURE I_D - V_{GS} CHARACTERISTIC CURVES FOLLOWING A POSITIVE |
| BIAS HTGB STRESS TEST AT 150 °C, DEPENDING ON MEASUREMENT DELAY TIME |
| Figure 7-24: Drain current plotted on a linear scale as a function of V_{GS} following a negative bias HTGB stress |
| TEST COMPARING (A) IMMEDIATE ROOM TEMPERATURE I_D - V_{GS} CHARACTERISTICS WITH A ONE-HOUR DELAY TIME AND (B) |
| IMMEDIATE HIGH TEMPERATURE I_D - V_{GS} CHARACTERISTICS VERSUS A ROOM-TEMPERATURE MEASUREMENT WITH A ONE- |
| HOUR DELAY TIME |

1 Introduction

1.1 Research goals and thesis outline

Silicon carbide (SiC) is a compound, wide bandgap semiconductor with many superior material qualities that make it very attractive as a replacement for silicon (Si) in high-temperature power electronics. Because a native oxide can be grown on SiC, it is especially attractive for fabricating metal-oxide-semiconductor field-effect transistors (MOSFETs), which are the devices of choice for power systems engineers because they are both voltage-controlled and normally off. However, several outstanding reliability issues still remain, including operational-stress induced instability of the threshold voltage.

The goals of this research have been to study the threshold-voltage (V_T) instability phenomenon in SiC MOSFETs as a function of time, bias, and temperature, and to understand both the cause of this effect, as well as how it affects the reliability of power SiC MOSFETs for high-power converters for both military and commercial applications.

After providing an outline and listing key contributions, Chapter 1 goes on to discuss the military and commercial interest in SiC, and in particular SiC MOSFETs. Chapter 2 reviews the different types of interfacial charge at the SiC conducting channel/gate-oxide insulator interfacial region and how they affect the device characteristics. Basic materials analysis results of the transition regions on both sides of the SiC-silicon dioxide (SiO₂) interface are discussed, followed by a review of the few specific mentions of oxide trap charge in the literature. Given the similarity of the V_T

instability in SiC MOSFETs with that observed previously in irradiated Si MOSFETs, these previously observed effects are summarized and a successful model explaining the Si MOS V_T instability involving direct tunneling of electrons to and from near-interfacial oxide traps is reviewed along with key materials analysis results of the physical and chemical nature of the oxide trap defect from electron spin resonance measurement results. Chapter 3 then discusses the effect of the different types of interfacial charge on the threshold voltage, the theoretical calculation of the oxide field due to the presence of this charge, in particular trapped oxide charge, and how these results are applied by a numerical simulator to more precisely calculate the electric potential and thereby the electric field. This is used as input for the tunneling model discussed in Chapter 6.

Chapter 4 presents the basic experimental results, showing the effect of bias stress, time, and temperature on V_T instability. Chapter 5 presents the effects of measurement speed, and discusses how all these effects are consistent with a tunneling model. Also discussed are the results of subthreshold swing and charge separation analyses. Chapter 6 presents the results of a two-way tunneling analysis, allowing for the simultaneous charging and discharging of near-interfacial oxide traps as a function of gate-bias stress and time, during both the stress and the measurement of that stress. Both the successes and limitations of this model in comparison with experimental results and analysis are discussed. Chapter 7 discusses the V_T instability in SiC power MOSFETs, with special attention to effects at elevated temperatures where these power devices would be expected to operate. Reliability issues are addressed, including potential failure modes and mechanisms. Chapter 8 presents a summary of the study along with the main

conclusions. Also discussed are topics for future work, based on the work presented here.

1.2 Key contributions

- Discovered and communicated the basic V_T instability phenomenon in SiC MOSFETs and its significance as a potential HTRB (high-temperature reverse bias) reliability failure mechanism by increasing *OFF*-state leakage current.
- Demonstrated that this V_T -instability effect is due to the charging and discharging of near-interfacial oxide traps, and discussed how this likely involves interface traps in a two-step process.
- Developed and applied a simultaneous two-way tunneling model to match and explain the experimental results.
- Demonstrated the critical importance of measurement time to accurately determine the number of active switching oxide traps and their effects, given that oxide traps will change charge state due to the bias applied during the measurement as well.
- Discovered a large increase in bias-stress induced V_T instability at temperatures above 100 °C and explained this in terms of a bias-dependent high-temperature increase in the number of active *E'*-center oxide defects.
- Based upon a large variation in high-temperature V_T -instability results, suggested that the flat response of some SiC MOSFETs versus temperature may be due to a balance of oxide charge trapping effects and mobile ion drift.

- Suggested that many traps that are considered to be interface traps may actually be near-interfacial oxide traps. This is important since different types of interfacial charge are likely associated with different physical defects and will therefore respond differently to processing variations.
- Showed the effect of processing on the magnitude of the V_T instability: devices with an implanted channel, such as the power DMOSFET, exhibit a small increase; whereas devices that did not receive a post-oxidation NO anneal of their gate oxide exhibit a significant increase.
- Showed how, given the complex time, temperature, and bias dependence of the V_T -instability effect, existing reliability standards based on Si device technology are likely inadequate for SiC MOSFETs.

1.3 Why SiC MOSFETs?

SiC has many potential material advantages over Si for the development of hightemperature, high-voltage, and high-frequency power switches, including a much wider bandgap (3.26 eV at room temperature vs 1.12 eV for Si), a far higher critical field (2.2 MV/cm vs 0.25 MV/cm for Si), a larger saturation velocity (2×10^7 cm/s vs 1×10^7 cm/s for Si), and a much greater thermal conductivity (3.0-3.8 W/cm·K vs 1.5 W/cm·K for Si) [1,2]. The SiC values given are for 4H SiC, which is the preferred SiC polytype for vertical power devices given its high bulk mobility parallel to the *c* –axis [3, 4]. Although the effective channel mobility of 4H SiC MOSFETs in the lateral direction, perpendicular to the *c*-axis, tends to be intrinsically poorer than for 6H devices (likely due to the slightly larger bandgap of 4H), this can be remedied with a standard postoxidation NO anneal [5].

The wide bandgap allows electronic devices to operate at much higher temperatures before the intrinsic carrier concentration overwhelms the extrinsic doping. The much larger critical field means that much higher blocking voltages can be achieved for the same semiconductor thickness, or alternatively, much lower drift-layer resistances can be achieved for the same blocking level [6, 7]. These factors lead to devices with lower conduction and switching losses and increase efficiency, and to devices with increased power density. The much larger thermal conductance is useful in high power applications where devices generate their own heat that must be removed. However, this heat conduction is ultimately controlled by the thermal conductance of the entire package—that includes the die attach, thermal interface, and heat sink—which generally constitutes more than 90 percent of the total resistance in even the best packaging [8]. Operating at higher frequencies and higher temperatures will enable systems with much smaller weight and volume, due to the decreased size of passive system components and decreased cooling requirements. Moving from Si to SiC and replacing mechanical functions with electronics should increase system efficiency, functionality, and reliability.

Military and commercial applications include the power conversion (DC to AC inversion and DC to DC conversion) for hybrid-electric vehicles (HEVs), power generation and distribution, motor control, and power factor correction for power supplies for computers and electronic appliances. It is estimated that the introduction of SiC Schottky diodes into just 1.5 percent of the market has to date already saved in efficiency gains the electricity output of three coal-fired power plants [9]. Looked at

another way, this increased efficiency can lead to an improved standard of living for people all around the world by reducing their energy costs.

SiC MOSFETs are very attractive power switches, being both voltage-controlled and normally off. Bipolar junction transistors (BJTs) are normally off, but are current controlled devices, which require more complicated—and power dissipating—control circuits. Junction field-effect transistors (JFETs) are voltage controlled devices, but are normally on. Power engineers are naturally cautious, and prefer direct, normally off, replacement devices which fail safe. However, SiC BJTs and JFETs are easier to fabricate, and do not have as many materials issues as the SiC MOSFET [7], some of which are described in Chapter 2. (An alternative point of view is put forward by Treu, et al. [10], who argue the case for the power JFET precisely because of the SiC MOSFET reliability issues.) The Doubly-implanted MOSFET (DMOSFET) is the more common type of power MOSFET, given the UMOSFET's (named for its U-shaped trench) etched sidewalls and high-field crowding issues, although the implantations required for the DMOSFET lead to device degradations as well [7].

SiC power DMOSFETs have been successfully demonstrated by various companies, including Cree, GE, and Rohm [11, 12, 13]. Cree in fact has very recently announced the first commercial release of a 20-A, 1200-V device [14], although with limited negative bias and temperature rating [15]. The development of several different power conversion 1200-V modules has also been recently demonstrated, at current ratings of up to 550 A and junction temperatures up to 200 °C [12, 16, 17, 18, 19].

As is the case for Si, a thermal oxide can be grown on SiC [20], enabling the fabrication of MOSFETs. However, interfacial charge trapped at and near the SiC-SiO₂

inversion channel-gate insulator interface is far greater than in the Si case, especially in the upper half of the bandgap [4, 21, 22], and can lead to significant degradation in device performance by substantially reducing the effective channel mobility [5], as well as shifting the threshold voltage [23]. Previous works on SiC MOSFET reliability include an initial oxide processing study on the effects of interfacial charge on oxide breakdown [24]. Although time-dependent dielectric breakdown (TDDB) measurements indicate that the intrinsic gate-oxide strength of SiC MOS structures is similar to that of Si MOS structures [25, 26, 27, 28], with estimated lifetimes of 100 years at 3 MV/cm or higher, concern remains regarding extrinsic failures [26, 27, 28]. The past few years have also led to a general appreciation of V_T instability as an important reliability issue [9, 26, 29, 30].

2 Background

2.1 Review of the different types of interfacial charge

According to the Deal Committee formulation for Si MOS [31], charge at the interface and in the gate oxide can be categorized as due either to fixed oxide charge (Q_F , N_F), mobile ionic charge (Q_M , N_M), interface trapped charge (Q_{IT} , N_{IT}), or oxide trapped charge (Q_{OT} , N_{OT}), with Q representing a net charge per unit area and N representing the net number of charges per unit area for each type of charge. (See Chapter 3 for a discussion of how the actual charge density—which may be, for example, distributed across the oxide— is related to an effective charge density at the interface.) It is likely that this same convention can also be successfully applied to SiC MOS. A modification of the famous figure from Deal [31], indicating the names and locations of the various types of interfacial charge, is presented in Figure 2-1 for SiC.

$$\begin{array}{|c|c|c|c|c|}\hline SiO_2 & \stackrel{(N_0^+)}{\longrightarrow} & \stackrel{+}{(K^+)} \\ & \stackrel{+}{\xrightarrow{}} & \stackrel{+}{\xrightarrow{}$$

Figure 2-1: A modification of the Deal [31] Si MOS figure, indicating the names and locations of the various types of interfacial charge for SiC MOS.

All the different types of charge, even those in the oxide if they are close enough to the SiC interface, can cause a decrease in the actual channel mobility due to Coulombic scattering [32, 33]. The net charge, due to the sum of the different types of positive and negative charge present, can also reduce the effective mobility for a given surface potential due to a reduction of free carriers in the channel [34]. The importance of using Hall mobility measurements to separate the actual channel mobility from the presumed effective mobility was popularized by Saks in a series of papers that are summarized in a recent review article [35]. This reduction in channel mobility results in a greater ON-state resistance than would otherwise be the case. The net charge will also cause a shift of the threshold voltage (V_T) , with a net positive charge causing a negative shift and a net negative charge causing a positive shift (see Chapter 3 for details). Although thinning the gate oxide will reduce the V_T shift caused by this trapped charge, it also invites potential long-term gate oxide reliability issues revealed by time-dependent dielectric breakdown (TDDB) analysis [25, 26, 27, 28]. For an *n*-channel MOSFET, a negative shift will cause an increase in OFF-state leakage current and a positive shift will also cause an increase in ON-state resistance (see Chapter 7 for a discussion of how this impacts the power MOSFET). Finally, oxide traps, especially those near the SiC interface, are probably the cause of the instabilities observed in the threshold voltage following gate-bias stressing [23, 36, 37, 38, 39, 40, 41, 42, 43, 44, 45, 46, 47, 48, 49, 50, 51, 52, 53], which is the focus of this study.

This interfacial charge may lie on both sides of the SiC-SiO₂ interface, where the conducting channel of the semiconductor meets the gate insulator of the MOSFET. Transition layers have been observed on both sides of this interface using transmission electron microscopy (TEM), electron energy loss spectroscopy (EELS), and X-ray photoelectron spectroscopy [54, 55, 56, 57, 58]. These layers appear to be due to

chemical/compositional changes and extend several nanometers on both sides of the interface. A carbon (C) rich (or Si poor) region on the SiC side of the interface has been reported by several different groups [56, 59]. Recent work employing electron spin resonance techniques have led to an initial understanding of the physical and chemical nature of the interface traps. The dominant defect for thermally grown oxides on SiC appears to be a Si vacancy within the SiC near the interface [60, 61], and the dominant defect for deposited oxides on SiC appears to be a dangling bond right at the interface [62].

Many papers have also reported evidence of an oxide transition layer [57]. This oxide layer may be due to the presence of C [59], or it may simply be that the topology and the geometry of the SiC surface are not suitable for an abrupt oxide interface [63]. It has also been suggested that although the oxide on SiC is stoichiometric SiO₂, there exists a greater strain in the oxide grown on SiC [64]. In SiC these various effects can result in C clusters or C-related dangling bonds, suboxide bonds, and/or other defects which can trap charge—such as an oxygen (O) vacancy related to a broken Si-Si bond, referred to as an E' center [65]. This particular defect, which may very well be the source of the dominant oxide trap [61], will be discussed in much greater detail in a subsequent section of this chapter. It has been reported that the nitrogen (N) from the post-oxidation NO anneal used to reduce the interface trap density may cause additional charge trapping [66].

Numerous papers have addressed the effect of interface traps, particularly on low channel mobility. Modeling papers have found that Coulombic scattering due to interfacial charge is the dominant mechanism limiting channel mobility, but they

generally assumed that this charge was mainly due to interface traps and did not specifically discuss the effect of oxide traps [32, 67]. A recent review article discusses interface traps and SiC in great detail [68]. Oxide traps are mentioned in passing, with speculation that an E' center may be involved. Finding processing steps to reduce N_{IT} , particularly the high density of interface traps (D_{IT}) near the conduction band edge, has been the main focus for the past decade and a great deal of effort has been exerted in finding processing steps that reduce these densities, including the development of the nitrogen post-oxidation anneal [69, 70, 71].

However, the success of this effort in reducing negatively charged interface traps has led, in some cases, to negative threshold voltages which may result in normally-*ON* devices or, at the very least, MOSFETs with high *OFF*-state leakage currents. This is due to the as-yet unreduced large numbers of interfacial and near-interfacial positive charge, which charge separation analysis indicates exceeds 1×10^{12} cm⁻² in typical SiC MOSFETs [72]. Clearly, more focus is needed on finding improved processing steps to reduce N_F and N_{OT} as well so that the present balance between large numbers of positive and negative trapped charge in *n*-channel SiC MOSFETs, which in general leads to reasonable threshold voltages, is made less precarious. The threshold-voltage instability phenomenon clearly demonstrates that not all the non- N_{IT} trapped charge is fixed and that at least some of this charge must be due to N_{OT} that is close enough to the interface to exchange charge over the time scales observed [23]. In this capacity, V_T instability measurements provide a method for determing the lower bound for the calculation of N_{OT} (see Chapter 5).

In Si MOS, N_F was believed to occur in the first 25 Å of the oxide from the Si interface due to structural defects such as ionized Si, but was not in electrical communication with the Si [31]. Presumably similar fixed charge exists for SiC MOS, but the physical nature of this charge has not yet been identified.

Mobile ions have generally been thought to not be an issue, although the variation in the high-temperature V_T -instability response, discussed in Chapters 4 and 7, may be the result of a variation in the mobile ion content. Since charge-trapping, likely caused by the filling or emptying of near-interfacial oxide traps, and mobile ion drift have opposite effects on V_T for a given bias stress, a small effective V_T shift at high temperature may be due to a balance of these two effects.

Oxidation in the presence of alumina has been recently reported to increase channel mobility [73, 74]. However, many contaminants are introduced into the oxide by this process, including alkali metals such as Na, K, Ca, and Mg, and transition metals such as Fe, Ni, Ti, and Cr. Significant threshold voltage shifts due to mobile ions have been reported as a result, due mostly to the sodium contamination [73].

2.2 Other work related to oxide charge trapping in SiC MOS

In addition to my work on the threshold-voltage instability of SiC MOSFETs due to the charging and discharging of near-interfacial oxide traps described in this thesis (and presented at various international conferences and published in various proceedings and journals [23, 37, 38, 39, 42, 43, 45, 47, 48, 50, 51]), a few other works in the literature have also discussed oxide charge trapping in SiC MOSFETs, and recent work has confirmed the threshold-voltage instability phenomenon described here.

Most of the work on SiC MOSFETs regarding charge trapping at or near the SiC/SiO₂ interface has been related to the low effective mobilities that are typically measured, and processing variations that reduce the interface trap density. Oxide traps or fixed charge are rarely mentioned. Although oxide traps are only rarely discussed in SiC, it is a big issue for Si MISFET devices that have gate insulators made with high-dielectric constant (high-k) compounds. With the continued scaling of Si CMOS devices and the accompanying thinning of their gate oxides, the use of alternative dielectrics with high-k values has become conventional. Since devices with these alternative gate dielectrics have a lot of oxide traps, it is not surprising that they also have significant thresholdvoltage instabilities [75]. One measurement method developed for analyzing the trapping effect in high-k Si MISFETs is a fast I-V (current-voltage) system [76]. This system was replicated by Dr. John Suehle's team at NIST (National Institute for Science and Technology), with whom I began a collaboration after visiting them and realizing that their fast *I-V* system would be very helpful in the study of oxide traps in SiC. (The details of the experimental procedure are given in Chapter 4 and some of the experimental results of our collaboration are presented in Chapter 5.) Moshe Gurfinkel, the graduate student in his lab that set-up and used the fast *I-V* system became very interested in the study of SiC and made numerous additional measurements providing additional confirmation of the time-sensitive nature of the I_D - V_{GS} characteristics [40, 41].

Marko Tadjer, another graduate student at the University of Maryland, also became interested in the study of the V_T instability of SiC MOSFETs after hearing about my results, and subsequently also presented a study of the bias, time, and temperature effects [46].

Following my presentation at the 2006 Spring MRS meeting [38], I advised researchers from GE regarding their own, subsequent V_T -instability measurements [44]. Even more recently, researchers from Germany [49] and Japan [52] have presented initial studies of shifts in V_T as a function of bias, time, and temperature.

One early paper on SiC MOS charge trapping presented results of measurements on MOS capacitors [24]. It actually assessed the potential of various alternative dielectrics in addition to deposited and thermally grown SiO₂. The researchers from Cree presented results not only on the interface trap density, but the net oxide charge densities as well by comparing their measured and calculated ideal flat-band voltages. They observed hysteresis in their *C-V* (capacitance-voltage) measurements, but attributed that to interface-trap charging.

In their study of the effects of nitridation on properties of the SiC-SiO₂ interface, Afanas'ev, et al. concluded that two different types of traps are present, fast interface traps and very slow states "likely related to defects in the near-interface oxide layer" [77]. Rozen, et al. found that nitridation reduced the interface trap density and susceptibility to electron trapping, but increased hole trapping [66]. Recent work by Kong, Dimitrijev,and Han have concluded that time-dependent variations in SiC MOS capacitor *C-V* characteristics are due to the presence of near-interface oxide traps that are either neutral or positively charged [78]. They adopted a direct tunneling analysis presented previously by Tewksbury, et al. [79] for the study of V_T instability in Si MOSFETs. Even more recent work by Basile, et al. also looking at the time-dependent responses of SiC MOS capacitor C-V characteristics and using constant-current deep-level transient spectroscopy (DLTS) found three types of charge traps: oxide traps, interface traps, and

semiconductor bulk traps [80]. The presence of a SiC bulk trap has also been suggested by Agarwal, et al. [81]. These conclusions are consistent with the materials analysis work mentioned in the previous section.

Several papers over the past decade have addressed the radiation response of SiC MOSFETs [82, 83, 84]. *n*-channel devices have generally exhibited a positive shift whereas *p*-channel devices have exhibited a significant negative shift. This implies that interface trap formation dominates over the creation of oxide traps, although very recent and as yet unpublished results show a negative shift of V_T following the irradiation of an *n*-channel SiC MOSFET under positive-bias stress—implying that the filling of positively-charged oxide traps is dominating the response [85]. Lee, et al. [83] used subthreshold swing analysis to attempt to separate interface traps from oxide traps. The importance of oxide traps is discussed in detail in the next section.

2.3 Review of threshold-voltage instability in irradiated Si MOS

The instability of V_T in SiC MOSFETs is reminiscent of a similar effect observed in irradiated Si MOSFETs described in earlier work [86, 87, 88, 89, 90]. Exposing Si MOSFETs to ionizing radiation was found to lead to the creation of electron-hole pairs in the gate oxide, with some fraction escaping immediate recombination. With a positive bias applied to the gate, the electrons are quickly swept out to the gate electrode and the holes make their way toward the Si-SiO₂ interface via a stochastic-hopping transport mechanism, with many holes becoming trapped in deep trap sites in the oxide near the interface [91, 92]. These trapped holes lead to a negative shift of the threshold voltage. However, this V_T shift is not permanent, and is observed to "anneal" over time, even at

room temperature. That is, especially under the presence of a positive gate bias, the postirradiation V_T shift becomes less negative over time. This was initially explained by electrons tunneling in from the Si substrate to recombine with the trapped holes. However, it was further observed that if subsequently a negative gate bias is applied, that some of the previously annealed charge would return and V_T would shift back to the left [86, 87].

My colleagues and I at the former Harry Diamond Laboratories (HDL—a precursor of the present U.S. Army Research Laboratory) performed a series of studies to better understand this "reverse annealing," or reversibility, phenomenon [88, 89, 90], and proposed a molecular model to explain this instability, which we attributed to the filling and emptying of near-interfacial oxide traps via a direct tunneling mechanism. This V_T instability in irradiated Si MOSFETs is in many ways very similar to the bias instability observed in as-processed SiC MOSFETs described in this work.

We found that there was a difference in the magnitude of the reversibility depending on the processing. Devices with hardened gate oxides, whose initial postirradiation negative V_T shift is much smaller than that of unhardened devices, displayed significant "reverse annealing." Devices with unhardened or "soft" gate oxides showed little or no "reverse annealing" [88]. However, at elevated temperature (up to 150 °C), the devices with "soft" gate oxides also began to display significant reversibility of the V_T annealing. The devices with "hard" gate oxides also showed some increase in reversibility, but the percentage change was much smaller [89].

We also found that the amount of reversibility as a percentage of the initial V_T shift was measurement speed dependent. Fast 100-µs ramp *I-V* (and *C-V* for similarly

processed capacitors) measurements revealed that only a relatively small fraction (about 15 percent for one sample set) of the total initial V_T shift that was annealed was reversible, even though slower 1-s ramp *I-V* measurements gave the impression that such samples experienced a 100 percent reversibility of all the previously annealed charge. We also observed using our fast *I-V* system that if the bias was switched from positive gate bias to negative very early during the original anneal, e.g., after only 1 ms, the device would experience an initial V_T shift back to the left and then resume annealing back to the right, although at a slower rate than under positive gate bias [88]. This time dependence of the measurements is also observed in as-processed SiC MOSFETs (see Chapter 5).

2.4 Review of magnetic resonance work and the HDL oxide hole trap model

Many years ago, Lenahan and Dressendorfer [93] confirmed using electron spin resonance (ESR) techniques, coupled with electrical *C-V* measurements, that the *E'* center was in fact the dominant radiation-induced oxide hole trap in Si MOSFET gate oxides, confirming previous speculation by Marquardt and Sigel [94]. They found that the V_{MG} shift, due entirely to oxide trapped holes since the interface traps are uncharged at the mid-gap surface potential, correlated very well with both the increase in the number of *E'* centers immediately following irradiation, as well as during a subsequent anneal. They also confirmed, by etching back the oxide, that most of the *E'* centers were located near the Si-SiO₂ interface.

The E' center consists of two Si atoms, each back-bonded to three different O atoms, and is activated by the breaking of a weak Si-Si bond, which occurs due to an O

vacancy (see Figure 2-2 (a) and (b)). This is due to the Si rich nature of the oxide in the interfacial region [95]. The weak Si-Si bond can be broken either by radiation-induced holes or during processing. As discussed below, it appears that they can also be broken at elevated temperature in the presence of an applied bias [96].



Figure 2-2: HDL (ARL) hole trap model [88].

Figure 2-2 is a depiction of the hole trap model put forward by my colleagues and me at HDL to explain the hole trap "reverse annealing" phenomena, based on the results of Lenahan and Dressendorfer [93]. When the radiation-induced hole breaks the weak Si-Si bond (see Figure 2-2(a)), the hole is captured by one of the Si atoms and the other is left with a dangling bond—see Figure 2-2(b). The ESR measurement detects the unpaired spin of this dangling bond, with different defects identified by their own unique signatures. During the subsequent anneal, the ESR signal decreased along with the reduction in the V_{MG} shift. This implies that either the bond was reformed, eliminating both the trapped hole and the unpaired spin of the dangling bond, or that a compensating electron tunneling into the oxide was captured by the previously uncharged Si atom, leading to a charge-neutral dipole and the elimination of the unpaired spin by filling both orbitals—see Figure 2-2(c). The reverse-anneal effect was explained simply by one of the electrons tunneling back out, leaving behind the original active E' complex, as depicted in Figure 2-2(b). This model was subsequently confirmed with ESR work performed by Conley and Lenahan, et al. [97, 98], who showed that the E' signal grew larger again under negative bias during the reverse anneal, following a previous decrease under positive bias, and that this process was repeatable—just like the electrical V_T instability. Recent work has incorporated this same model as the first step of a new two-step model to explain negative bias temperature instability (NBTI) effects in Si MOS [99]. This defect has also been suggested to act as a neutral electron trap [100].

It is likely that when the weak Si-Si bond is first broken, the O atoms back-bonded to the positively charged Si atom without the dangling bond relax back to a more planar configuration and that the two Si atoms move further apart, depending upon the existing bond strain [101]. The further apart they move, the less likely that the bond will reform. Only reformation of the Si-Si bond would result in true "annealing" of the trapped positive charge and deactivation of the defect site. Devices with hardened gate oxides are generally under more bond strain and therefore more likely to move apart and not reform [88]. However, as mentioned previously, a significant increase in the V_T instability occurs in devices with unhardened oxides at temperatures above 100 °C, suggesting that

elevated temperature will also cause the two Si atoms to move apart [89]. It may also cause some Si-Si bonds to break, creating active defect sites.

This is essentially the effect that was recently reported by Ryan, Lenahan, et al. who found, using on-the-fly ESR techniques in the study of NBTI in Si MOS, that the number of E' centers increased dramatically with bias at temperature, but not with temperature stress alone [96]. In addition, they found that the number of oxide defects markedly decreased when the sample was returned to room temperature. This is very similar to the V_T instability behavior observed in SiC MOSFETs (see Chapter 7). Since the V_T instability is likely a measure of the number of active near-interfacial oxide traps, these results suggest that the application of a gate bias at elevated temperature will break existing Si-Si bonds and activate additional E' center defects that can act as nearinterfacial oxide traps. This argument is made more compelling given that Cochrane, Lenahan, et al. have recently reported that E' type oxide defects have been observed in SiC MOSFETs using electrically detected magnetic resonance (EDMR) techniques under a strong negative gate bias—conditions most conducive to detecting unpaired spins in the E' complex [61]. (This appears to confirm previous speculation regarding the nature of at least some of the near-interfacial oxide traps [21, 23].) They further suggested that in this case the EDMR likely involved spin dependent trap-assisted tunneling between defects on both sides of the interface. If all this is so, then it is important to develop improved processing methods to decrease the number of precursor oxide defect sites-or to decrease the bond strain near the interface.

2.5 Summary

Although most works on SiC MOSFETs focus on interface traps and fixed charge, it is likely that oxide traps, especially those very close to the SiC interface, are responsible for the threshold-voltage instability via a direct tunneling mechanism. It is important to distinguish between interface traps and near-interfacial oxide traps, even though it may be difficult to distinguish between the two during very fast measurements, since it is likely that they are associated with different types of physical defects—and therefore processing variations will affect them differently.

The V_T instability observed in as-processed SiC MOSFETs is reminiscent of a similar instability previously reported in irradiated Si MOSFETs. In that case the dominant radiation-generated defect was found to be an E' center, which consists of a broken Si-Si bond associated with an O vacancy. Even with all the reports of excess C on both sides of the interface, the only oxide defect so far observed using ESR on SiC MOS test structures is an E' type defect. This suggests that the tunneling analysis previously applied to irradiated Si MOSFETs may be successfully applied to as-processed SiC MOSFETs to explain the V_T instability. Furthermore, processing variations that improved radiation hardness in Si MOSFETs may provide insight in how to—and how not to—reduce V_T instability in SiC MOSFETs.

3 Calculation of Oxide Field and Threshold-Voltage Shift Caused by Interfacial Charge

3.1 Introduction: Calculation of electric potential in the gate oxide and the SiC channel region

This chapter describes the calculations employed to determine important input parameters for the tunneling model to be described in Chapter 6. Specifically, it is necessary, given a distribution of charged near-interfacial oxide traps, to determine the electric field in that region of the gate oxide for any applied gate-to-source voltage, V_{GS} . It is also important to determine the corresponding inversion carrier concentration, as that likely plays a role in determining the interface-trap occupation response time, as well as being a possible source of electrons which can tunnel into the oxide to neutralize the likely positively charged near-interfacial oxide traps.

The electric potential is calculated in both the oxide and SiC regions, first analytically, and then more precisely by way of numerical simulation using standard techniques. The analytical solution provides a starting point for the more exact numerical solution. Each will be described in turn, from which the gate oxide field and inversion carrier concentration can be readily derived.

In order to calculate the electric potential in both the oxide and the SiC for a given V_{GS} , it is first necessary to determine the relationship between the applied gate voltage and the potential drops across both the oxide and the SiC. The standard formulation, including the effects of the work-function difference between the poly-Si gate and the
SiC semiconductor, and the effects of trapped charge at the SiC-SiO₂ interface, is as follows:

$$V_{GS} = V_{FB} + V_{OX} + V_{int} + V_{SiC}$$

$$= V_{FB} + V_{OX_net} + V_{SiC}$$
(3-1)

where V_{FB} is the gate voltage needed to achieve flat-band conditions due to the workfunction difference noted above, V_{OX} is the potential drop across the gate oxide due only to inversion channel charge, V_{int} is the voltage shift caused by the presence of the net sum of the interfacial trapped charge present—and which modifies the potential drop across the gate oxide, V_{OX_net} , and V_{SiC} is the potential drop in the SiC.

For the ideal case, where there is no interfacial trapped charge, this reduces to

$$V_{GS_ideal} = V_{FB} + V_{OX} + V_{SiC}$$
(3-2)

Before proceeding to describe each component in detail, it is useful to define the electric potential, ϕ , the bulk potential in the SiC, ϕ_B , and the band bending in the SiC, ψ . The bulk potential in the SiC is calculated from the doping density, and is related to the difference between the intrinsic, mid-gap energy level, E_i , and the Fermi level, E_F :

$$\phi_B = \frac{kT}{q} \cdot \ln\left(\frac{N_a}{n_i}\right) = \frac{E_{i_bulk} - E_F}{q}$$
(3-3)

where N_a is the *p*-type doping in the SiC, n_i is the intrinsic carrier concentration, *q* is the electronic charge, *k* is Boltzmann's constant, and *T* is absolute temperature.

The band-bending can be defined as the amount of deviation from the SiC bulk flat-band condition, which can also be related to the bulk potential and Fermi level:

$$\psi(x) = \frac{E_{i_{bulk}} - E_i(x)}{q} = \phi_B + \frac{E_F - E_i(x)}{q}$$
(3-4)

The surface potential is the degree of band-bending at the SiC-SiO₂ interface, and can be defined by the deviation from flat-band conditions in terms of the bulk potential, $\psi_s (\phi_B)$. In other words, at flat-band, there is no band bending and so $\psi_s = 0$. At mid-gap, ψ_s has been bent by one ϕ_B , such that E_i is now equal to E_F and the electron carrier concentration at the surface equals that of the holes. When ψ_s has been bent by two ϕ_B , the surface is at inversion such that E_i is now one ϕ_B below E_F and the electron carrier concentration at the surface is now equal to the hole carrier concentration in the bulk for an *n*-channel MOSFET. Any further increase in the band bending at this point will drive the surface into strong inversion, with a dramatic increase in the electron carrier concentration in the first few nanometers of the SiC. On the other hand, when the surface potential is negative, the surface is driven into accumulation, where the hole carrier concentration increases dramatically.

$$\psi_{s} = \psi(x=0) = \frac{E_{i_bulk} - E_{i}(x=0)}{q}$$
(3-5)

The potential, $\phi(x)$, is defined so that when $\phi = 0$ the equilibrium electron and hole carrier concentrations are both equal to the intrinsic carrier concentration, n_i . This implies that the potential at the interface, ϕ_{int} , will equal zero at mid-gap, when $\psi_s = \phi_B$. Thus, the potential in the SiC can be written in terms of the band bending:

$$\phi(x) = \psi(x) - \phi_B = \frac{E_F - E_i(x)}{q}$$
(3-6)

And the electron and hole carrier concentrations can be defined in terms of the electric potential:

$$n(x) = n_i \cdot \exp\left(\frac{q \cdot \phi(x)}{kT}\right) = n_i \cdot \exp\left(\frac{E_F - E_i(x)}{kT}\right)$$
(3-7)

$$p(x) = n_i \cdot \exp\left(\frac{-q \cdot \phi(x)}{kT}\right) = n_i \cdot \exp\left(\frac{-[E_F - E_i(x)]}{kT}\right)$$
(3-8)

3.2 Calculation of the Flat-band voltage, V_{FB}

The individual components of the voltage drop from gate to source can be calculated as follows. The flat-band voltage is due to the difference in the work functions of the poly-Si gate and SiC substrate.

$$V_{FB} = \phi_{ms} = (\chi_{Si} + \frac{E_{g_Si}}{2 \cdot q} + \phi_{poly}) - (\chi_{SiC} + \frac{E_{g_SiC}}{2 \cdot q} + \phi_B)$$
(3-9)

where ϕ_{ms} is the work-function difference between the "metal" (*p*-type poly-Si) and the semiconductor (*p*-type SiC); χ_{Si} and χ_{SiC} are the electron affinities of the poly-Si and SiC, respectively; E_{g_Si} and E_{g_SiC} are the bandgap energies of Si and SiC, respectively; ϕ_{poly} is the bulk potential of the highly doped *p*-type poly-Si gate, and and ϕ_B is the bulk potential for the *p*-type SiC—which is the normalized difference between the Fermi level, E_F , and the mid-gap level, E_i .

For example, if the SiC *p*-type bulk is doped 1×10^{16} cm⁻³ and the poly-Si is boron doped to 3×10^{19} cm⁻³, then

$$V_{FB} = \phi_{ms} = (4.05 + \frac{1.11}{2} + 0.555) - (3.60 + \frac{3.26}{2} + 1.424) = -1.494 \text{ V}$$

$$\chi_{\rm Si} = 4.05 \text{ eV}$$

$$Eg_{\rm Si} = 1.11 \text{ eV}$$

$$V_{\rm FB} = -|\phi_{\rm ms}|$$

$$Eg_{\rm SiO2} = 9.0 \text{ eV}$$

$$Eg_{\rm SiO2} = -1.4 \text{ eV}$$

Figure 3-1: Schematic of the energy bands for, from left to right, the *p*-type poly-Si gate, the gate oxide, and the *p*-type SiC.

3.3 Calculation of V_{SiC} and V_{OX} in the absence of interfacial charge

The potential drop in the SiC is the difference between the electric potential at the surface, $\phi(x=0)$, and in the bulk, $\phi(x=bulk)$, which is the surface potential, ψ_s :

$$V_{SiC} = \phi(x=0) - \phi(x=bulk) = [\psi(x=0) - \phi_B] - [-\phi_B] = \psi_s$$
(3-10)

The electron and hole carrier concentrations, (3-7) and (3-8), respectively, can be re-written in terms of the band-bending. First, note that the equilibrium hole and electron carrier concentrations can be found, for a conventionally doped *p*-type semiconductor, as

$$p_0 = N_a = n_i \cdot \exp\left(\frac{q \cdot \phi_B}{kT}\right) \tag{3-11}$$

$$n_0 = \frac{n_i^2}{p_0} = n_i \cdot \exp\left(\frac{-q \cdot \phi_B}{kT}\right)$$
(3-12)

Therefore, the electron and hole carrier concentrations can be re-written as follows:

$$n(x) = n_i \cdot \exp\left(\frac{q \cdot \phi(x)}{kT}\right)$$

$$= n_i \cdot \exp\left(\frac{q \cdot [\psi(x) - \phi_B]}{kT}\right) = n_0 \cdot \exp\left(\frac{q \cdot \psi(x)}{kT}\right)$$
(3-13)

$$p(x) = n_i \cdot \exp\left(\frac{-q \cdot \phi(x)}{kT}\right)$$

$$= n_i \cdot \exp\left(\frac{-q \cdot [\psi(x) - \phi_B]}{kT}\right) = p_0 \cdot \exp\left(\frac{-q \cdot \psi(x)}{kT}\right)$$
(3-14)

The electric field at the semiconductor surface can be related to the surface potential by solving Poisson's Equation, as is described in Sze [102], Streetman [103], and surely other standard device physics texts.

$$\frac{\partial^2 \psi(x)}{\partial x^2} = -\frac{\rho(x)}{\varepsilon_s} = -\frac{q}{\varepsilon_s} \cdot \left[p(x) - n(x) - N_a^- + N_d^+ \right]$$
(3-15)

where ρ is the charge density, ε_s is the permittivity of the semiconductor, and N_d is the *n*-type doping in the SiC (if present).

Poisson's Equation can be solved to obtain the following equation for the electric field:

$$\mathcal{E}(x) = \sqrt{\frac{2 \cdot kT \cdot N_a}{\varepsilon_s}} \cdot \sqrt{\left(\exp\left(\frac{-q \cdot \psi(x)}{kT}\right) + \frac{q \cdot \psi(x)}{kT} - 1\right) + \frac{n_0}{p_0} \cdot \left(\exp\left(\frac{q \cdot \psi(x)}{kT}\right) - \frac{q \cdot \psi(x)}{kT} - 1\right)}$$
(3-16)

The pre-factor is commonly written in terms of the Debye length,

$$L_D = \sqrt{\frac{\varepsilon_s \cdot kT}{q^2 \cdot N_a}} \tag{3-17}$$

so that

$$\mathcal{E}(x) = \frac{\sqrt{2} \cdot kT}{q \cdot L_{D}} \cdot \sqrt{\left(\exp\left(\frac{-q \cdot \psi(x)}{kT}\right) + \frac{q \cdot \psi(x)}{kT} - 1\right) + \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(\exp\left(\frac{q \cdot \psi(x)}{kT}\right) - \frac{q \cdot \psi(x)}{kT} - 1\right)}$$
(3-18)

When x=0, this relationship provides a way of relating the electric field at the SiC surface to the surface potential (see Figure 3-2):

$$\mathcal{E}_{s} = \frac{\sqrt{2 \cdot kT}}{q \cdot L_{D}} \cdot \frac{\sqrt{2 \cdot kT}}{q \cdot L_{D}} \cdot \frac{\sqrt{\left(\exp\left(\frac{-q \cdot \psi_{s}}{kT}\right) + \frac{q \cdot \psi_{s}}{kT} - 1\right) + \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(\exp\left(\frac{q \cdot \psi_{s}}{kT}\right) - \frac{q \cdot \psi_{s}}{kT} - 1\right)}}$$
(3-19)



Figure 3-2: Surface electric field for the SiC as a function of the SiC surface potential, in units of the bulk potential, ϕ_B .

This in turn can be used to find the electric field at the gate oxide surface, using Gauss' Law, which can then be used to calculate the voltage drop across the oxide due to the inversion charge:

$$V_{OX} = \boldsymbol{\mathcal{E}}_{OX} \cdot \boldsymbol{t}_{OX} = \left[\frac{\boldsymbol{\mathcal{E}}_{SiC}}{\boldsymbol{\mathcal{E}}_{OX}} \cdot \boldsymbol{\mathcal{E}}_{SiC}\right] \cdot \boldsymbol{t}_{OX} = \frac{\left(\boldsymbol{\mathcal{E}}_{0} \cdot \boldsymbol{\mathcal{E}}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC}\right)}{\left[\frac{\boldsymbol{\mathcal{E}}_{0} \cdot \boldsymbol{\mathcal{E}}_{OX}}{\boldsymbol{t}_{OX}}\right]} = \frac{-\boldsymbol{\mathcal{Q}}_{s}}{\boldsymbol{\mathcal{C}}_{OX}}$$
(3-20)

where ε_0 , ε_{OX} , ε_{SiC} are the free-space, relative oxide, and relative SiC permittivities, respectively; \mathcal{E}_{OX} is the electric field throughout the oxide and \mathcal{E}_{SiC} is the electric field at the SiC surface; t_{OX} is the oxide thickness; C_{OX} is the oxide capacitance; and Q_s is the surface areal charge density for the semiconductor.

Therefore, the ideal case of (3-2), where there is no interfacial trapped charge, can be re-written using (3-9), (3-10), and (3-20):

$$V_{GS_ideal} = V_{FB} + V_{OX} + V_{SiC} = \phi_{ms} + \frac{\left(\varepsilon_0 \cdot \varepsilon_{SiC} \cdot \varepsilon_s\right)}{C_{OX}} + \psi_s$$
(3-21)

Thus, knowing the surface potential, both the voltage drop across the

semiconductor, V_{SiC} , and that across the gate oxide, V_{OX} , can be calculated. However, the modifications to the oxide field (and voltage drop across the oxide) due to interfacial trapped charge, and the spatial distribution of inversion layer charge in the SiC, still need to be found.

3.4 Voltage shift due to interface traps

In the absence of any interfacial trapped charge, the field in the oxide is easily calculated from the surface field in the SiC using Gauss's Law, (3-20), as discussed in Section 3.3 above, once the surface potential is determined. However, in present state-of-

the-art SiC MOSFET devices, there exist significant numbers of interface traps and nearinterfacial oxide traps which must be considered when calculating the field and voltage drop across the oxide.

As discussed above in Chapter 2, experimental evidence exists to suggest that not all interface traps exist right at the SiC-SiO₂ interface, but instead also extend into the SiC channel region [60, 61]. A more sophisticated model should take this into account, although that will not be considered here. Traps that extend into the gate oxide are not interface traps, but rather near-interfacial oxide traps. The calculations presented here assume that all interface traps exist exactly at the SiC-SiO₂ interface. In this case, the field in the oxide is modified by this interface trap charge, found by applying Gauss' Law

$$\boldsymbol{\mathcal{E}}_{OX_B} = \frac{\left(\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX} \cdot \boldsymbol{\mathcal{E}}_{OX_S} - \boldsymbol{Q}_{IT}\right)}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}} = \frac{\left(\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_S} - \boldsymbol{Q}_{IT}\right)}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}}$$
(3-22)

where \mathcal{E}_{OX_B} is the electric field in the bulk of the oxide, \mathcal{E}_{OX_S} is the electric field at the surface of the oxide, and \mathcal{E}_{SiC_S} is explicitly the electric field at the surface of the SiC.

The voltage drop across the gate oxide is now modified by the presence of the interface-trap charge:

$$V_{OX_net} = \boldsymbol{\mathcal{E}}_{OX_B} \cdot \boldsymbol{t}_{OX} = \frac{\left(\boldsymbol{\varepsilon}_0 \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_S} - \boldsymbol{\mathcal{Q}}_{IT}\right)}{\boldsymbol{\varepsilon}_0 \cdot \boldsymbol{\varepsilon}_{OX}} \cdot \boldsymbol{t}_{OX} = \frac{-\boldsymbol{\mathcal{Q}}_s - \boldsymbol{\mathcal{Q}}_{IT}}{\boldsymbol{\mathcal{C}}_{OX}}$$
(3-23)

where Q_{IT} is surface charge density due to the interface trap charge, and is simply related to the net number of actively charged interface traps

$$N_{IT} = \frac{Q_{IT}}{q} \tag{3-24}$$

 Q_{IT} and N_{IT} are calculated from the interface trap density, D_{IT} , as follows. First, mid-gap neutrality of interface traps is assumed, wherein traps below mid-gap are

assumed to be donor-like: positively charged when unoccupied and neutral when filled, and traps above mid-gap are assumed to be acceptor-like: neutral when unoccupied and negatively charged when filled.

For a given surface potential, which determines the energy level at the interface, the fraction of traps that are filled at any energy level, *E*, follow Fermi-Dirac statistics:

$$f_e(E) = \frac{1}{1 + \exp\left\{\left(\frac{1}{kT}\right) \cdot \left[E - \left(\frac{E_{g_SiC}}{2} + q \cdot (\psi_S - \phi_B)\right)\right]\right\}}$$
(3-25)

where E_{g_SiC} is the energy band gap of SiC. This relationship is true once equilibrium has been established between the interface traps and SiC substrate [20, 102, 104].

The interface trap density versus bandgap energy is assumed according to the following formulations [32]. On the low, or valence side of mid-gap

$$D_{IT_{v}}(E) = D_{IT_{mid_{v}}} + D_{IT_{edge_{v}}} \cdot \exp\left\{\frac{-(E - E_{v_{s}})}{\sigma_{v}}\right\}$$
(3-26)

where $D_{IT_mid_v}$ is the assumed interface trap density around mid-gap, $D_{IT_edge_v}$ is the assumed interface trap density at the valence band edge, σ_v is the assumed exponential decay factor of the band-edge traps, and E_{v_SiC} is the energy of the valence band edge.

Similarly, on the high, or conductance side of mid-gap

$$D_{IT_{c}}(E) = D_{IT_{mid_{c}}} + D_{IT_{edge_{c}}} \cdot \exp\left\{\frac{-(E_{c_{sic}} - E)}{\sigma_{c}}\right\}$$
(3-27)

where $D_{IT_mid_c}$ is the assumed interface trap density around mid-gap, $D_{IT_edge_c}$ is the assumed interface trap density at the conductance band edge, σ_c is the assumed exponential decay factor of the band-edge traps, and E_{c_SiC} is the energy of the conductance band edge. Typical values are given in Table 3-1. A graph of the interface trap density as defined in (3-26) and (3-27), covering the entire bandgap, is shown in Figure 3-3.

| D _{IT_mid_v} | 6.0×10 ¹³ | (cm ⁻² ⋅eV ⁻¹) | D _{IT_mic} |
|------------------------|----------------------|---------------------------------------|---------------------|
| D _{IT_edge_v} | 2.3×10 ¹³ | (cm ⁻² ·eV ⁻¹) | D _{IT_edg} |
| σ_v | 0.061 | | σ_{c} |

| D _{IT_mid_c} | 6.0×10 ¹³ | (cm ⁻² ⋅eV ⁻¹) |
|------------------------|----------------------|---------------------------------------|
| D _{IT_edge_c} | 2.3×10 ¹³ | (cm ⁻² ⋅eV ⁻¹) |
| σ_c | 0.061 | |

Table 3-1: Typical values used for the parameters of (3-26) and (3-27).

Therefore, the total interface trap charge for a given surface potential can be calculated by integrating over the entire bandgap, keeping in mind that for traps below mid-gap, it is the number of unoccupied states that is important.

$$Q_{IT} = q \cdot \int_{E_{v_{-}SiC}}^{E_{mg_{-}SiC}} D_{IT}(E) \cdot \left[1 - f_{e}(E)\right] dE - q \cdot \int_{E_{mg_{-}SiC}}^{E_{c_{-}SiC}} D_{IT}(E) \cdot f_{e}(E) dE$$
(3-28)

A graph of this result, again using the values in Table 3-1, is shown in Figure 3-4. If this charge were the only contribution to the interfacial charge, it could be written as

$$V_{IT} = V_{\text{int}} = -\frac{Q_{IT}}{C_{OX}}$$
(3-29)



Figure 3-3: Interface trap density versus bandgap energy, calculated using (3-26), (3-27), and Table 3-1.



Figure 3-4: Variation of interface trap charge with surface potential, plotted in units of the bulk potential.

The effect of interface traps is to simply change the uniform field across the gate oxide, although that charge is a function of the surface potential and there is a response time associated with filling and emptying the interface traps, which is a function of the SiC carrier concentration at the surface as well as the distance in energy from the bandgap edges [20, 102].

Although fixed charge will not be considered, since experimental evidence suggests it may not be significant (see Chapter 5), the effect of any such charge would be to simply change the uniform field across the gate oxide by an additional factor since fixed charge is generally believed to be located at the oxide interface [31].

$$V_{OX_net} = \boldsymbol{\mathcal{E}}_{OX_B} \cdot \boldsymbol{t}_{OX} = \frac{\boldsymbol{\varepsilon}_0 \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_S} - (\boldsymbol{Q}_{IT} + \boldsymbol{Q}_F)}{\boldsymbol{\varepsilon}_0 \cdot \boldsymbol{\varepsilon}_{OX}} \cdot \boldsymbol{t}_{OX} = \frac{-\boldsymbol{Q}_s - (\boldsymbol{Q}_{IT} + \boldsymbol{Q}_F)}{\boldsymbol{C}_{OX}} \quad (3-30)$$

In terms of (3-1), this modified voltage drop across the oxide is the sum of that due only to charge in the semiconductor, V_{OX} , and that due to the sum of the net interfacial charge, V_{int} :

$$V_{OX_net} = -\frac{Q_s}{C_{OX}} - \frac{Q_{IT} + Q_F}{C_{OX}} = V_{OX} + V_{int} = V_{OX} + V_{IT} + V_F$$
(3-31)

where V_F is the voltage shift due to fixed charge.

3.5 Voltage shift due to oxide traps, calculation of the oxide trap charge centroid, and its effect on the oxide field

The oxide traps in the gate oxide are the only other type of interfacial trap to be considered. If all the charge in the oxide was right at the semiconductor interface as well, then the same simple expression could be used to describe the voltage shift due to this type of charge:

$$V_{OT} = -\frac{Q_{OT}}{C_{OX}} = -\frac{Q_{OT}}{\varepsilon_{OX}} \cdot t_{OX}$$
(3-32)

where Q_{OT} is the total oxide trap charge per unit area.

This expression is reasonably accurate for thick oxides, or where the charge does not penetrate very deeply into the oxide. But a more accurate analysis requires calculating the charge centroid, which in the case of switching oxide traps will be dependent on the time and bias history of the device. (Since mobile ions are likely not an issue in these devices—although devices with Al gates may be and those exposed to the MEO process definitely are [73, 74]—this type of charge will not be considered here. However, if it was, a similar type of calculation would have to be done since mobile ions are also spread across the oxide.) The expression for the voltage shift is modified as follows

$$V_{OT}(t) = \frac{-Q_{OT}(t)}{\varepsilon_{OX}} \cdot \overline{d_{OX}}(t) = \frac{-Q_{OT}(t)}{C_{OX}} \cdot \left[\frac{\overline{d_{OX}}(t)}{t_{OX}}\right] = \frac{-Q_{OT}(t)}{C_{OX}} \cdot f_{cent.}$$
(3-33)

where $\overline{d_{0X}}$ is the centroid of this charge, similar in concept to the center of mass when calculating moments, and $f_{cent.}$ is the charge centroid factor.

If the oxide charge is described by a volumetric oxide trap density, n_{OT} , as follows

$$Q_{OT}(t) = \int_{x_1}^{x_2} q \cdot n_{OT}(x, t) dx$$
 (3-34)

where $0 \le x_1 < x_2 \le t_{OX}$, then the charge centroid can be calculated using the following expression, which is the integral summation of each bit of charge at each position in the oxide divided by the total charge, Q_{OT} , in the oxide:

$$\overline{d_{OX}}(t) = \frac{\int_{x_1}^{x_2} x \cdot q \cdot n_{OT}(x,t) dx}{\int_{x_1}^{x_2} q \cdot n_{OT}(x,t) dx} = \frac{\int_{x_1}^{x_2} x \cdot q \cdot n_{OT}(x,t) dx}{Q_{OT}(t)}$$
(3-35)

Therefore, the shift in voltage can be written as

$$V_{OT}(t) = \frac{-Q_{OT}(t)}{\varepsilon_{OX}} \cdot \overline{d_{OX}}(t) = -\frac{q}{\varepsilon_{OX}} \cdot \int_{x_1}^{x_2} x \cdot n_{OT}(x, t) dx$$
(3-36)

For a linearly varying charge distribution versus position in the oxide,

$$n_{OT} = \frac{(n_{\max} - n_{\min})}{(x_2 - x_1)} \cdot (x - x_1) + n_{\min}$$
(3-37)

where n_{max} and n_{min} are both charge per volume (see Figure 3-5), the charge centroid due to this charge would be

$$\overline{d_{OX}} = \frac{\int_{x_1}^{x_2} x \cdot q \cdot \left[\frac{(n_{\max} - n_{\min})}{(x_2 - x_1)} \cdot (x - x_1) + n_{\min}\right] dx}{\int_{x_1}^{x_2} q \cdot \left[\frac{(n_{\max} - n_{\min})}{(x_2 - x_1)} \cdot (x - x_1) + n_{\min}\right] dx}$$
(3-38)



Figure 3-5: Schematic of a general charge distribution in the gate oxide.

A useful, simplified result is when a uniform distribution extends only a certain

depth into the oxide (see Figure 3-6). The depth to which the distribution extends can be defined as

$$t_{OT} = t_{OX} - f_{OT} \cdot t_{OX} = (1 - f_{OT}) \cdot t_{OX}$$
(3-39)

where $0 \le f_{OT} < 1$, so that $t_{OT} = 0$ when $f_{OT} = 1$. The charge centroid can be readily calculated, using (3-35):

$$\overline{d_{OX}} = \frac{\int_{f_{OT} \cdot t_{OX}}^{t_{OX}} x \cdot (q \cdot n_{const}) dx}{\int_{f_{OT} \cdot t_{OX}}^{t_{OX}} (q \cdot n_{const}) dx} = \frac{(q \cdot n_{const}) \cdot (\frac{t_{OX}^2}{2} - f_{OT}^2 \cdot \frac{t_{OX}^2}{2})}{(q \cdot n_{const}) \cdot (t_{OX} - f_{OT} \cdot t_{OX})} = \frac{(1 + f_{OT})}{2} \cdot t_{OX} \quad (3-40)$$

which would result in the following voltage shift

$$V_{OT} = -\frac{Q_{OT}}{\varepsilon_{OX}} \cdot \overline{d_{OX}} = -\frac{Q_{OT}}{C_{OX}} \cdot \frac{(1+f_{OT})}{2} = -\frac{Q_{OT} \cdot f_{cent.}}{C_{OX}}$$
(3-41)

where the charge centroid factor for this case is

$$f_{cent.} = \frac{\overline{d_{OX}}}{t_{OX}} = \frac{\left(1 + f_{OT}\right)}{2}$$
(3-42)



Figure 3-6: Schematic of a rectangular charge distribution in the gate oxide near the SiC interface.

For example, if the charge distribution extends to a depth that is eight percent of the total oxide thickness such that t_{OT} is $0.08 \cdot t_{OX}$ and $f_{OT} = 0.92$, then

$$\overline{d_{OX}} = f_{cent.} \cdot t_{OX} = \frac{(1+f_{OT})}{2} \cdot t_{OX} = \frac{(1+0.92)}{2} \cdot t_{OX} = 0.96 \cdot t_{OX}$$
(3-43)

and

$$V_{OT} = -\frac{Q_{OT}}{C_{OX}} \cdot \frac{\overline{d_{OX}}}{t_{OX}} = -\frac{Q_{OT} \cdot f_{cent.}}{C_{OX}} = -\frac{Q_{OT} \cdot (0.96)}{C_{OX}}$$
(3-44)

This would be the case for a uniform charge distribution that extended 40 Å into a 500-Å thick gate oxide, which results in a four percent correction in the magnitude of the voltage shift. For example, if the oxide trap density was 1×10^{12} cm⁻², then the resulting voltage shift would actually be

$$V_{oT} = -\frac{Q_{oT} \cdot (0.96)}{C_{oX}} = -\frac{Q_{oX}}{\varepsilon_{oX}} \cdot (0.96 \cdot t_{oX})$$

$$= -\frac{1.6 \times 10^{-19} (\text{C}) \cdot 1 \times 10^{12} (\text{cm}^{-2})}{3.9 \cdot 8.85 \times 10^{-14} (\text{F/cm})} \cdot (0.96 \cdot 500 \times 10^{-8} \text{ cm}) = -2.224 \text{ V}$$
(3-45)

instead of the –2.317 V that would be calculated if the effects of the spatial distribution of the charge were ignored. This difference is small, and not very important in itself, especially when considering the measurement uncertainty from experimental results. However, the electric field does change significantly within the oxide trap region, and that is important in accurately modeling the tunneling effect, which is discussed in Chapter 6.

Therefore, modifying (3-31) to include the term for oxide trap charge yields:

$$V_{OX_net} = V_{OX} + V_{int} = -\frac{Q_s}{C_{OX}} - \frac{Q_{IT} + Q_F + Q_{OT} \cdot f_{cent.}}{C_{OX}}$$
(3-46)

Although mobile ions are not considered here, a full description of all the possible charge components, discussed in Chapter 2, affecting the voltage drop across the oxide would be

$$V_{OX_net} = V_{OX} + V_{int} = \frac{-Q_s}{C_{OX}} + \frac{-(Q_{IT} + Q_F + Q_{OT} \cdot f_{cent.} + Q_M \cdot f_M)}{C_{OX}}$$
(3-47)

where Q_M is the mobile ion charge, and f_M is its charge centroid.

The modification of the voltage drop across the gate oxide due to a rectangular distribution of oxide trap charge near the interface, given by (3-41), and its resultant effect on the electric field within the gate oxide, can be better understood by way of a graphical example.

Without any interfacial charge, including oxide trap charge, the field in the bulk of the oxide is equivalent to the field at the interface, found from the electric field in the SiC at the interface using Gauss' Law (3-20):

$$\boldsymbol{\mathcal{E}}_{OX_B} = \boldsymbol{\mathcal{E}}_{OX_S} = \left(\frac{\boldsymbol{\mathcal{E}}_{SiC}}{\boldsymbol{\mathcal{E}}_{OX}}\right) \cdot \boldsymbol{\mathcal{E}}_{SiC_S}$$
(3-48)

and the voltage drop is simply the hatched area shown in Figure 3-7.



Figure 3-7: Voltage drop across the oxide due to Q_s .

With the presence of a rectangular distribution of interfacial oxide charge, Q_{OT} , as in Figure 3-6, Gauss' Law is modified ala (3-22), giving a modified electric field in the bulk of the oxide:

$$\boldsymbol{\mathcal{E}}_{OX_{B}} = \boldsymbol{\mathcal{E}}_{OX_{S}} - \frac{\boldsymbol{\mathcal{Q}}_{OT}}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}} = \frac{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_{S}} - \boldsymbol{\mathcal{Q}}_{OT}}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}}$$
(3-49)

which can be rewritten as

$$\boldsymbol{\mathcal{E}}_{OX_S} - \boldsymbol{\mathcal{E}}_{OX_B} = \frac{\boldsymbol{\mathcal{Q}}_{OT}}{\boldsymbol{\mathcal{E}}_{0} \cdot \boldsymbol{\mathcal{E}}_{OX}}$$
(3-50)

The modified voltage drop across the oxide can be readily found graphically, as depicted in Figure 3-8:

$$V_{OX_net} = V_{OX} + V_{int}$$

$$= V_{OX_b} + V_{OX_s} = \mathcal{E}_{OX_B} \cdot t_{OX} + \left(\mathcal{E}_{OX_S} - \mathcal{E}_{OX_B}\right) \cdot \frac{(1 - f_{OT})}{2} \cdot t_{OX}$$
(3-51)



Figure 3-8: Voltage drop across the oxide is modified by a rectangular oxide trap distribution as in Figure 3-6, which changes the bulk field.

This result can also be readily found by calculating the changing oxide field in the interfacial region due to the rectangular charge distribution.

$$\boldsymbol{\mathcal{E}}(x) = \boldsymbol{\mathcal{E}}_{OX_{B}} + \int_{f_{OT} \cdot f_{OX}}^{f_{OX}} \frac{q}{\varepsilon} \cdot \boldsymbol{n}_{OT}(x) dx = \boldsymbol{\mathcal{E}}_{OX_{B}} + \frac{q}{\varepsilon} \cdot \boldsymbol{n}_{const} \cdot (x - f_{OT} \cdot t_{OX})$$
(3-52)

The change in the voltage drop across the oxide due to the presence of this positive trapped charge is the total oxide trap charge times the charge centroid factor. This can be found using (3-47), (3-50), and (3-43) derived above.

$$V_{OT} = V_{int} = V_{OX_net} - V_{OX}$$

$$= (V_{OX_b} + V_{OX_s}) - V_{OX}$$

$$= \left\{ \boldsymbol{\mathcal{E}}_{OX_B} \cdot \boldsymbol{t}_{OX} + \left(\boldsymbol{\mathcal{E}}_{OX_S} - \boldsymbol{\mathcal{E}}_{OX_B} \right) \cdot \frac{(1 - f_{OT})}{2} \cdot \boldsymbol{t}_{OX} \right\} - \boldsymbol{\mathcal{E}}_{OX_S} \cdot \boldsymbol{t}_{OX}$$

$$= \left(\boldsymbol{\mathcal{E}}_{OX_S} - \boldsymbol{\mathcal{E}}_{OX_B} \right) \cdot \left(\frac{(1 - f_{OT})}{2} - 1 \right) \cdot \boldsymbol{t}_{OX}$$

$$= -\left(\boldsymbol{\mathcal{E}}_{OX_S} - \boldsymbol{\mathcal{E}}_{OX_B} \right) \cdot \frac{(1 + f_{OT})}{2} \cdot \boldsymbol{t}_{OX} = -\left(\frac{\boldsymbol{\mathcal{Q}}_{OT}}{\boldsymbol{\mathcal{E}}_{0} \cdot \boldsymbol{\mathcal{E}}_{OX}} \right) \cdot \boldsymbol{f}_{cent.} \cdot \boldsymbol{t}_{OX}$$

$$= -\frac{\boldsymbol{\mathcal{Q}}_{OT} \cdot \boldsymbol{f}_{cent.}}{C_{OX}}$$

$$(3-53)$$

This result can also be determined graphically, as shown below in Figure 3-9.



Figure 3-9: The voltage drop across the oxide is modified by the hatched area due to the presence of the rectangular distribution of oxide traps near the interface as depicted in Figure 3-6.

3.6 Effect of interfacial charge on threshold voltage and the gate oxide field

The net interfacial charge described by (3-47) can be normalized as if all the charge was at the interface:

$$V_{\rm int} = \frac{-Q_{net} \cdot f_{net}}{C_{OX}} = \frac{-(Q_{IT} + Q_F + Q_{OT} \cdot f_{cent.} + Q_M \cdot f_M)}{C_{OX}}$$
(3-54)

Then (3-1) can be re-written, using (3-21), as:

$$V_{GS} = V_{FB} + V_{OX} + V_{int} + V_{SiC} = \phi_{ms} + \frac{\left(\varepsilon_0 \cdot \varepsilon_{SiC} \cdot \varepsilon_{SiC_s}\right)}{C_{OX}} - \frac{Q_{net} \cdot f_{net}}{C_{OX}} + \psi_s \qquad (3-55)$$

Thus the threshold voltage can be found from (3-55), by setting the surface potential to two times the bulk potential, keeping in mind the proper occupancy of the interface traps:

$$V_T = V_{GS}(\psi_s = 2 \cdot \phi_B) = \phi_{ms} + \frac{\left(\varepsilon_0 \cdot \varepsilon_{SiC} \cdot \varepsilon_{SiC}\right)}{C_{OX}} - \frac{Q_{net} \cdot f_{net}}{C_{OX}} + 2 \cdot \phi_B$$
(3-56)

The difference between the measured and ideal threshold voltage is due to the shifts caused by the interfacial charge:

$$V_T - V_{T_{ideal}} = -\frac{Q_{net} \cdot f_{net}}{C_{OX}} = V_{int} = V_{IT} + V_F + V_{OT} + V_M$$
(3-57)

where V_M is the voltage shift due to mobile ions in the gate oxide.

Figure 3-10 shows the effect of trapped charge on the relationship between the applied gate-to-source voltage and the SiC surface potential. For the ideal case without trapped charge, circles indicate the applied gate voltage required to place the device in accumulation, flat-band, mid-gap, inversion, and strong inversion (where the surface potential is -0.1, 0.0, 1.0, 2.0, and 2.1 times the SiC bulk potential, ϕ_B , respectively). The addition of 1×10^{12} cm⁻² trapped positive oxide charge, as discussed in the previous section, results in a negative shift versus V_{GS} of 2.224 V.



Figure 3-10: Variation in required applied gate-to-source voltage to effect a particular surface potential in a SiC MOSFET ($N_a=1\times10^{16}$ cm⁻³, $t_{ox} = 500$ Å) for both an ideal device with no trapped charge and one with an oxide trapped charge density of $+1\times10^{12}$ cm⁻².

The relationship between the oxide field and the gate-to-source voltage, without any trapped charge present, is as follows:

$$\boldsymbol{\mathcal{E}}_{OX} = \frac{V_{OX}}{t_{OX}} = \frac{\left(V_{GS_ideal} - V_{FB} - V_{SiC}\right)}{t_{OX}} = \frac{\left(V_{GS_ideal} - \phi_{ms} - \psi_{s}\right)}{t_{OX}}.$$
(3-58)

This result is plotted in Figure 3-11, along with a simpler rule-of-thumb calculation:

$$\boldsymbol{\mathcal{E}}_{OX} = \frac{V_{OX}}{t_{OX}} \approx \frac{V_{GS}}{t_{OX}}$$
(3-59)

For a negative value of V_{GS} , where the device is in accumulation, the magnitude of the oxide field is overestimated by V_{FB} / t_{OX} . For the case of positive V_{GS} , where the device is in strong inversion, the magnitude of the oxide field is overestimated by $(V_{FB} + 2 \times \phi_B) / t_{OX}$. Since $\phi_{ms} \sim -\phi_B$, the approximate values err almost equally for both positive and negative oxide fields.



Figure 3-11: Comparison of the ideal relationship between V_{GS} and gate oxide field, without interfacial charge present, and a simpler rule-of-thumb calculation.

For the example given above, where a device has an oxide trap density of 1×10^{12} cm⁻² with a uniform charge distribution that extends 40 Å into a 500-Å thick gate oxide, the oxide field at the surface and in the bulk, calculated from (3-20) and (3-49), are compared to the ideal value, as seen in Figure 3-12.

A similar negative shift of V_{GS} is seen when plotting the oxide surface field versus applied gate voltage. The curve when trapped charge is present (top, red line with diamonds) is translated 2.224 V to the left of the ideal curve (middle, black line with circles). This figure also shows a comparison of the required applied gate voltage, when charge in the oxide is present, between the surface oxide field (top, red line with diamonds) and the bulk oxide field (bottom, blue line with squares)—the two are the same when no charge is present in the oxide. For the same applied gate voltage, the two fields always vary by the same amount. Using our example from above, $\mathcal{E}_{OX_S} - \mathcal{E}_{OX_B} = 0.463 \text{ MV/cm}.$

Whereas the surface fields at the same applied gate voltage are never the same for different amounts of oxide trap charge, the bulk oxide fields in either accumulation or strong inversion are very similar. This in turn has important implications for various testing scenarios, where the oxide is charging during the measurement. It implies that for an oxide reliability test where the bulk oxide field is most important, such as for a TDDB test, charging may not be as important, but for threshold-voltage instability, which appears to be driven by charge tunneling into and out of near-interfacial oxide traps, it is the field in the trapped oxide-charge region which is most important—and that varies greatly with trapped charge.



Figure 3-12: Variation in required applied gate-to-source voltage to effect a particular oxide field in a SiC MOSFET ($N_a=1\times10^{16}$ cm⁻³, $t_{ox} = 500$ Å) for both an ideal device with no trapped charge, where the surface and bulk fields are the same; and one with an oxide trapped charge density of $+1\times10^{12}$ cm⁻², where the surface and bulk fields are different.

3.7 Numerical simulation of electric potential in the SiC channel region

Now that the effect of interfacial charge has been considered, the gate oxide can be calculated—by first calculating the electric potential. The calculation of the potential can be broken up into two parts: (1) calculation in the SiC—described in this section; and (2) calculation in the gate oxide—described in the next section.

This can be done since the electric potential is known at the boundary points: deep in the bulk of the SiC, at the SiO₂ interface, and at the gate:

$$\phi_{subst} = \psi(x = bulk) - \phi_B = -\phi_B \tag{3-60}$$

$$\phi_{\rm int} = \psi(x=0) - \phi_B = \psi_s - \phi_B \tag{3-61}$$

$$\phi_{gate} = \psi(x = gate) - \phi_B = (V_{OX} + V_{int}) + \psi_s - \phi_B$$
 (3-62)

These act as boundary conditions for the numerical simulations.

The potential, electric field, charge carrier concentrations, and other variables can be modeled either analytically using the depletion approximation, or numerically using a finite difference code that solves Poisson's Equation. Before employing numerical simulation techniques to determine the exact electric potential and carrier concentrations within the semiconductor, a standard analytical approach was used to obtain an approximate solution to provide a proper starting point for the exact solution.

In order to readily calculate the band-bending—and therefore the electric potential and carrier concentrations—as a function of position within the semiconductor, the depletion approximation can be used wherein the ionized acceptor ions are assumed to be the only source of charge. Poisson's Equation (3-15) is now written as

$$\frac{\partial^2 \psi(x)}{\partial x^2} = -\frac{\rho(x)}{\varepsilon_s} = -\frac{q}{\varepsilon_s} \cdot \left[-N_a^- \right] = \frac{q \cdot N_a^-}{\varepsilon_s}$$
(3-63)

Integrating (3-63) twice yields an expression for the spatial variation in the band-bending potential, in terms of the surface potential and surface electric field, which can also be written in terms of the maximum depletion width, W_D —beyond which the field is assumed to be zero.

$$\psi(x) = \frac{q \cdot N_a^-}{\varepsilon_s} \cdot \frac{x^2}{2} - \boldsymbol{\mathcal{E}}_s \cdot x + \psi_s = \frac{q \cdot N_a^-}{\varepsilon_s} \cdot \frac{(x - W_D)^2}{2}$$
(3-64)

The depletion approximation also yields a simpler relationship between the surface electric field and the band-bending at the surface over the range of flatband to inversion, which can also be re-written in terms of the maximum depletion width:

$$\boldsymbol{\mathcal{E}}_{s} = \sqrt{\frac{2 \cdot q \cdot N_{a} \cdot \boldsymbol{\psi}_{s}}{\varepsilon_{s}}} = \frac{q \cdot N_{a}}{\varepsilon_{s}} \cdot W_{D}$$
(3-65)

This then gives the electric potential and bending of the energy bands as a function of position in the semiconductor using (3-6), which can be used as a good approximation and starting point for an exact solution using numerical methods, to be discussed next.

The spatial variation in electric potential for various surface potentials is shown in Figure 3-13.



Figure 3-13: Spatial variation of electric potential in both gate oxide and SiC, as a function of surface potential for the following conditions: accumulation ($\psi_s = -0.1 \cdot \phi_B$), flat-band ($\psi_s = 0$), zero potential ($V_{GS} = 0$; $\psi_s = 0.57 \cdot \phi_B$), mid-gap ($\psi_s = \phi_B$), inversion ($\psi_s = 2 \cdot \phi_B$), and strong inversion ($\psi_s = 2.1 \cdot \phi_B$); for $N_a = 1 \times 10^{16}$ and $t_{ox} = 50$ nm.

In a similar manner, the spatial variation in electron carrier concentration can be found using either (3-7) or (3-13)

The numerical simulation of the electric potential in the SiC is performed using standard techniques, as follows. First, a Taylor Series expansion is performed for an incremental change in the electric potential:

$$\phi(x + \Delta x) = \phi(x) + \frac{\partial \phi(x)}{\partial x} \cdot \Delta x + \frac{1}{2} \cdot \frac{\partial^2 \phi(x)}{\partial x^2} \cdot (\Delta x)^2 + O \cdot (\Delta x)^3$$
(3-66)

where Δx is a small spatial increment and O() represents higher order terms. A similar series expansion can be calculated for an incremental decrease in the electric potential.

$$\phi(x - \Delta x) = \phi(x) - \frac{\partial \phi(x)}{\partial x} \cdot \Delta x + \frac{1}{2} \cdot \frac{\partial^2 \phi(x)}{\partial x^2} \cdot (\Delta x)^2 - O \cdot (\Delta x)^3$$
(3-67)

Adding (3-66) and (3-67) gives

$$\phi(x + \Delta x) + \phi(x - \Delta x) = 2 \cdot \phi(x) + \frac{\partial^2 \phi(x)}{\partial x^2} \cdot (\Delta x)^2$$
(3-68)

which can be re-written as

$$\frac{\partial^2 \phi(x)}{\partial x^2} = \frac{\phi(x + \Delta x) - 2 \cdot \phi(x) + \phi(x - \Delta x)}{(\Delta x)^2}$$
(3-69)

Thus a discretized version of Poisson's Equation (3-15) is

$$\frac{\partial^2 \phi}{\partial x^2} \Big|_i = \frac{\phi_{i-1} - 2 \cdot \phi_i + \phi_{i+1}}{(\Delta x)^2}$$

$$= -\frac{q}{\varepsilon_s} \cdot \left[n_i \cdot \exp\left(\frac{-q \cdot \phi_i}{kT}\right) - n_i \cdot \exp\left(\frac{q \cdot \phi_i}{kT}\right) + D_i \right]$$
(3-70)

where *D* is the doping in the channel. For the *p*-type substrate,

$$D = -N_a \tag{3-71}$$

By defining a function, f, as the difference of the two sides of (3-70),

$$f_{i}(\phi_{i-1},\phi_{i},\phi_{i+1}) = \frac{\phi_{i-1} - 2 \cdot \phi_{i} + \phi_{i+1}}{(\Delta x)^{2}} + \frac{q}{\varepsilon_{s}} \cdot \left[n_{i} \cdot \exp\left(\frac{-q \cdot \phi_{i}}{kT}\right) - n_{i} \cdot \exp\left(\frac{q \cdot \phi_{i}}{kT}\right) + D_{i} \right]$$
(3-72)

and calculating the derivative in terms of ϕ ,

$$\frac{\partial f_i(\phi_{i-1},\phi_i,\phi_{i+1})}{\partial \phi_{i-1}} = \frac{\partial f_i(\phi_{i-1},\phi_i,\phi_{i+1})}{\partial \phi_{i+1}} = \frac{1}{(\Delta x)^2}$$
(3-73)

$$\frac{\partial f_i(\phi_{i-1},\phi_i,\phi_{i+1})}{\partial \phi_i} = \frac{-2}{\left(\Delta x\right)^2} - \frac{q \cdot n_i}{\varepsilon_s \cdot \left(\frac{kT}{q}\right)} \cdot \left[\exp\left(\frac{-q \cdot \phi_i}{kT}\right) + \exp\left(\frac{q \cdot \phi_i}{kT}\right)\right]$$
(3-74)

Newton's Method can then be used to find an iterative, self-consistent solution for the electric potential throughout the SiC channel region. Since

$$f'(x) = \frac{f(x + \Delta x) - f(x)}{\Delta x}$$
(3-75)

Rearranging terms and setting $f(x+\Delta x)=0$ gives

$$f(x + \Delta x) = f(x) + f'(x) \cdot \Delta x = 0$$
(3-76)

Similarly,

$$f_{i}(\phi_{i-1} + \Delta\phi_{i-1}, \phi_{i} + \Delta\phi_{i}, \phi_{i+1} + \Delta\phi_{i+1})$$

$$= f_{i}(\phi_{i-1}, \phi_{i}, \phi_{i+1}) + \frac{\partial f_{i}}{\partial \phi_{i-1}} \cdot \Delta\phi_{i-1} + \frac{\partial f_{i}}{\partial \phi_{i}} \cdot \Delta\phi_{i} + \frac{\partial f_{i}}{\partial \phi_{i+1}} \cdot \Delta\phi_{i+1} = 0$$
(3-77)

Rearranging terms gives the following expression

$$\frac{\partial f_i}{\partial \phi_{i-1}} \cdot \Delta \phi_{i-1} + \frac{\partial f_i}{\partial \phi_i} \cdot \Delta \phi_i + \frac{\partial f_i}{\partial \phi_{i+1}} \cdot \Delta \phi_{i+1} = -f_i(\phi_{i-1}, \phi_i, \phi_{i+1})$$
(3-78)

which can be written in matrix format as:

$$\begin{bmatrix} \frac{\partial f_i}{\partial \phi_{i-1}}, \frac{\partial f_i}{\partial \phi_i}, \frac{\partial f_i}{\partial \phi_{i+1}} \end{bmatrix} \times \begin{bmatrix} \Delta \phi_{i-1} \\ \Delta \phi_i \\ \Delta \phi_{i+1} \end{bmatrix} = -f_i(\phi_{i-1}, \phi_i, \phi_{i+1})$$
(3-79)

A similar matrix can be written for the next point:

$$\begin{bmatrix} \frac{\partial f_{i+1}}{\partial \phi_i}, \frac{\partial f_{i+1}}{\partial \phi_{i+1}}, \frac{\partial f_{i+1}}{\partial \phi_{i+2}} \end{bmatrix} \times \begin{bmatrix} \Delta \phi_i \\ \Delta \phi_i \\ \Delta \phi_{i+1} \\ \Delta \phi_{i+2} \end{bmatrix} = -f_i(\phi_i, \phi_{i+1}, \phi_{i+2})$$
(3-80)

Combining all *N* equations produces a matrix expression

$$\left[\frac{\partial \overline{f}}{\partial \overline{\phi}}\right] \times \left[\Delta \overline{\phi}\right] = -\left[\overline{f}\right]$$
(3-81)

which must be solved for $[\Delta \phi]$:

Such matrix arithmetic can be readily done using MATLAB software from Mathworks. With each iteration, the new value of the electric potential, $[\phi]$, throughout the semiconductor is updated by the incremental value, $[\Delta \phi]$, calculated as above, for each mesh point within the SiC. For example, for the k^{th} iteration:

$$\left[\overline{\phi}\right]^{k+1} = \left[\overline{\phi}\right]^{k} + \left[\Delta\overline{\phi}\right]^{k}$$
(3-83)

When $[\Delta \phi]$ falls below a chosen limit, the system of simultaneous equations is considered to have converged to a solution, yielding a self-consistent value for ϕ throughout the SiC channel region. As was the case for the analytic solution, once the electric potential is found, the carrier concentrations are readily found as well.

3.8 Numerical simulation of electric potential and field in the gate oxide region for a given oxide trap distribution

The tunneling model to be described in Chapter 6 has as one of its inputs the electric field in the gate oxide, which varies spatially due to the charged oxide traps in the interfacial region. This section will describe how the field is determined, given an oxide trap distribution. The first step is to assume a certain volumetric density distribution of oxide traps as a function of oxide depth.

$$n_{OT_total}(x) = n_{OT_surf_} \cdot \exp\{-\lambda \cdot x\}$$
(3-84)

where λ is the decay factor and $n_{OT_surf.}$ is the density at the interface, which is related to the assumed total areal density, N_{OT_total} , and depth over which oxide traps exist, t_{OT} , as defined in (3-39):

$$n_{OT_surf.} = N_{OT_total} \cdot \frac{\lambda}{1 - \exp\{-\lambda \cdot t_{OT}\}}$$
(3-85)

which reduces to

$$n_{OT_surf_} = \frac{N_{OT_total}}{t_{OT}}$$
(3-86)

when $\lambda = 0$, which is the case for a rectangular distribution. Figure 3-14 illustrates a range of distributions, depending on the value of λ , assuming that all the oxide traps lie within 50 Å of the interface.

When discretizing this oxide trap distribution, an areal density sheet of charge can be calculated at each mesh point, where n_{OT} is the distribution of positively charged oxide traps:

$$dQ_{OT}(n) = q \cdot n_{OT}(n) \cdot dx \tag{3-87}$$

Summing all the increments gives the total oxide trap charge:

$$Q_{OT} = \sum_{n=1}^{N} dQ_{OT}(n)$$
(3-88)



Figure 3-14: Variation of oxide trap distribution versus decay parameter λ , for equal total charge over first 50 Å from the interface.

The charge centroid was calculated in (3-35); the discrete version is

$$\overline{d_{OX}} = \frac{\sum_{n=1}^{N} dQ_{OT}(n) \cdot (t_{OX} - n_{OT} \cdot dx)}{\sum_{n=1}^{N} dQ_{OT}(n)}$$
(3-89)

and the charge centroid factor is found as in (3-42) and V_{OT} as in (3-44).

Although it is possible to solve for ϕ simultaneously throughout both the SiC semiconductor and the gate oxide, by defining the mesh point at the interface using Gauss' Law

$$\varepsilon_{OX} \cdot \frac{(\phi_i - \phi_{i-1})}{\Delta x} = \varepsilon_{SiC} \cdot \frac{(\phi_{i+1} - \phi_i)}{\Delta x}$$
(3-90)

it is more convenient to solve for the electric potential separately, in the SiC as described previously, and in the gate oxide as described below. This is possible since (3-19) describes the electric field at the surface exactly due to the surface potential, and the electric potential at the interface is defined by (3-61). However, solving for the electric potential piecemeal does require an iterative method for determining the proper surface potential for a given applied gate-to-source voltage, since the tunneling model seeks to simulate the experimental method of necessarily applying an external bias instead of choosing an internal field or potential.

Newton's Method can once again be employed, but this time solving for ψ_s instead of ϕ , by defining a function, *f*, as the difference of the two sides of (3-55)

$$f(\psi_s) = V_{GS} - \left(\phi_{ms} + \frac{\varepsilon_0 \cdot \varepsilon_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC} (\psi_s)}{C_{OX}} - \frac{Q_{OT} \cdot f_{cent.}}{C_{OX}} - \frac{Q_{IT} (\psi_s)}{C_{OX}} + \psi_s\right)$$
(3-91)

and calculating the derivative in terms of ψ_s .

Once the surface potential is found for a given V_{GS} , then the electric potential at the interface and gate can be found using (3-61) and (3-62), respectively, and the electric field in the bulk of the gate oxide, where no oxide traps exist, can be found by again applying Gauss' Law

$$\boldsymbol{\mathcal{E}}_{OX_B} = \frac{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_S} - (\boldsymbol{\mathcal{Q}}_{OT} + \boldsymbol{\mathcal{Q}}_{IT})}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}}$$
(3-92)

Next, the electric potential at the edge of the oxide trap region of the gate oxide can be found:

$$\phi_{bulk} = \phi(x = t_{OT}) = \phi_{gate} - \mathcal{E}_{OX_B} \cdot (t_{OX} - t_{OT})$$
(3-93)

The electric potential as a function of position in the bulk of the gate oxide, which ranges from $x = t_{OT}$ (the furthest distance that oxide traps extend from the SiC interface) to $x = t_{OX}$ (the full thickness of the gate oxide, all the way to the poly-Si gate), is then

$$\phi(x:t_{OT}\dots t_{OX}) = \phi_{bulk} + \mathcal{E}_{OX_B} \cdot (x - t_{OT})$$

$$= \phi_{bulk} + \left[\frac{\phi_{gate} - \phi_{bulk}}{t_{OX} - t_{OT}}\right] \cdot (x - t_{OT})$$
(3-94)

Before solving exactly for the electric potential in the oxide trap region using numerical simulation techniques, an approximate solution is first calculated to provide a proper starting point for the exact solution, assuming a rectangular distribution of filled oxide traps.

$$\phi(x:0...t_{OT}) = \phi_{bulk} + \mathcal{E}_{OX_B} \cdot (x - t_{OT})$$

$$-\frac{1}{2} \cdot \left[\frac{\mathcal{E}_{OX_S_TT} - \mathcal{E}_{OX_B}}{t_{OT}}\right] \cdot (x - t_{OT})^2$$
(3-95)

where

$$\boldsymbol{\mathcal{E}}_{OX_S_IT} = \frac{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{SiC} \cdot \boldsymbol{\mathcal{E}}_{SiC_S} - \boldsymbol{Q}_{IT}}{\boldsymbol{\varepsilon}_{0} \cdot \boldsymbol{\varepsilon}_{OX}}$$
(3-96)

A numerical solution can again be found by solving simultaneously—iteratively using Newton's Method—a set of discretized Poisson's Equations, as in (3-70), in the oxide trap region of the gate oxide

$$\frac{\partial^2 \phi}{\partial x^2} \Big|_i = \frac{\phi_{i-1} - 2 \cdot \phi_i + \phi_{i+1}}{\left(\Delta x\right)^2} = -\frac{q}{\varepsilon_{OX}} \cdot \left[n_{OT_i}\right]$$
(3-97)

so that the function, f, the difference of the two sides of (3-97), is

$$f_{i}(\phi_{i-1},\phi_{i},\phi_{i+1}) = \frac{\phi_{i-1} - 2 \cdot \phi_{i} + \phi_{i+1}}{(\Delta x)^{2}} + \frac{q \cdot n_{OT_{i}}}{\varepsilon_{OX}}$$
(3-98)

and the derivative in terms of ϕ is

$$\frac{\partial f_i(\phi_{i-1}, \phi_i, \phi_{i+1})}{\partial \phi_{i-1}} = \frac{\partial f_i(\phi_{i-1}, \phi_i, \phi_{i+1})}{\partial \phi_{i+1}} = \frac{1}{(\Delta x)^2}$$
(3-99)

$$\frac{\partial f_i(\phi_{i-1},\phi_i,\phi_{i+1})}{\partial \phi_i} = \frac{-2}{\left(\Delta x\right)^2}$$
(3-100)

Once the electric potential is found, the electric field can be readily calculated:

$$\boldsymbol{\mathcal{E}}_{OX}\left(n+0.5\right) = -\left(\frac{\phi(n+1) - \phi(n)}{\Delta x}\right) \tag{3-101}$$

As a check, the areal charge density at each mesh point can be found from the change in the electric field:

$$dQ(n) = \varepsilon_{OX} \cdot \left[\mathcal{E}_{OX}(n+0.5) - \mathcal{E}_{OX}(n-0.5) \right]$$
(3-102)

which should be the same value as that found from (3-87). Figure 3-15 graphically shows the relationship between the different variables and their mesh point counters.



Figure 3-15: Schematic of how the electric field values are derived from the electric potential, and how the charge density values are confirmed by the electric field values.

3.9 Summary of calculations

This chapter has described how the ideal threshold voltage can be found by calculating the voltage drop across the oxide and semiconductor when the surface potential is at inversion, along with the adjustment due to the work function difference between the poly-Si gate and SiC substrate. It further described how the various types of interfacial charge can cause a shift in V_T .

Of particular interest is the determination of the field in the oxide near the interface for a given applied gate-bias and assumed trap distribution and occupancy, since this information is required by the tunneling model to calculate a time-dependent change in the distribution of occupied oxide traps. The process is as follows:
- 1. Choose desired gate-to-source bias.
- 2. Determine surface potential consistent with chosen V_{GS} value and assumed charge distribution for both oxide traps and interface traps.
- 3. Calculate boundary conditions for electric potential at gate, interface, and SiC substrate bulk based on chosen V_{GS} and calculated surface potential, and verify by calculating electric field at interface from the surface potential.
- 4. Self-consistently determine electric potential throughout the device based on assumed interfacial charge distributions and doping density using numerical simulation techniques. Use approximate solution determined from analytical calculations as starting point.
- 5. Calculate oxide field from electric potential.

The inversion carrier concentration is also readily calculated once the potential in the SiC is determined.

In addition, it was observed that the bulk oxide field does not vary much at all with trapped charge, and that the actual positive and negative high-field values are only slightly lower than the generally assumed rule-of-thumb value of V_{GS} / t_{OX} —which may be of importance when deciphering TDDB results. However, the oxide field at the surface can change significantly with interfacial charge.

4 Basic Experimental Results of Threshold-Voltage Instability Effect

4.1 Introduction

The experimental results of this study are presented in three parts. The basic threshold-voltage instability effect is presented in this chapter. The time-dependence of the measurements is presented in the following chapter. Finally, the effect of this threshold-voltage instability effect on the reliability of actual SiC power MOSFET devices will be presented in Chapter 7.

This chapter will present basic instability test results on both lateral MOSFET and capacitor test structures. The basic physical mechanisms that are responsible for this phenomenon are discussed.

4.2 The device sample sets

The devices studied in this work came primarily from one leading U.S. manufacturer of SiC devices, which prefers to remain unnamed, although I also studied devices from two other sources and observed similar threshold-voltage instabilities. This instability, which is typically on the order of between 0.25 and 0.5 volts for devices with gate oxide thicknesses of 500 to 1000 Å, following a gate-bias stress of about +/- 1 MV/cm applied for 5 minutes under each polarity, has been observed in both deposited and thermal oxides, in devices with both as-grown and implanted epitaxial SiC at the interface, and in devices on both 4H and 6H SiC. The results presented in this chapter are all on lateral 4H-SiC MOSFETs, along with some 4H-SiC capacitor data. Other than the obvious improvement provided by the NO post-oxidation anneal, this is not a study of processing variations. The same basic effect was always observed. Although numerous different sample sets were used in this study, they all behaved in the same general way, other than more recent devices showing a general improvement in their characteristics [105].

Figure 4-1shows an image of the primary lateral test chips used in this study (Sample Sets C, D, and E; as described below), with side-by-side as-grown epi ($N_a \sim$ 1×10^{16} cm⁻³) and implanted epi ($N_a \sim 1 \times 10^{17}$ cm⁻³) chips available. The implanted epi devices mimic the implanted region of the implanted DMOSFET channel (see Figure 7-1), and generally have lower effective channel mobilities compared to the as-grown epi devices. Figure 4-2 highlights various devices available for testing. There are lateral *n*channel MOSFETs, with varying gate areas and geometries: wide, short devices—100×3 and 100×5 (channel width by length in microns), 100×100 up to 400×400 so-called "FATFETs" because of their large gate area, and circular geometry enclosed MOSFETs with the source in the center and the drain on the outside that range from 424×1.0 up to 424×20 microns. These test chips also include *n*-type and *p*-type MOS capacitors with areas of 4×10^{-4} cm², 9×10^{-4} cm², and 16×10^{-4} cm². MOS Hall-bar test structures are also included, which could allow for the acquisition of Hall mobility and free carrier concentration measurements, the latter potentially useful in verifying the number of free carriers available to tunnel into the oxide traps under positive bias.

The lateral *n*-channel 4H-SiC MOSFET devices whose instability results are shown in this work came from nine different samples sets (shown in Table 4-1), wherein two of the sample sets had differently processed devices.

61

| Sample Set | t _{ox} (Å) | oxide | growth | N anneal |
|------------|---------------------|---------|------------------------|------------------------|
| А | 500 | LPCVD | 410°C LTO + 1200°C Dry | 1300°C NH ₃ |
| В | 600 | thermal | 1200°C Dry + 950°C Wet | 1300°C NH ₃ |
| С | 1,000 | LPCVD | 410°C LTO + 1200°C Dry | 1300°C NH ₃ |
| D | 1,300 | LPCVD | 410°C LTO + 1200°C Dry | 1300°C NH ₃ |
| E | 600 | thermal | 1200°C Dry + 950°C Wet | 1300°C NO |
| F | 600 | thermal | 1200°C Dry + 950°C Wet | 1175°C NO |
| G | 500 | thermal | 1175°C Dry + 950°C Wet | 1175°C NO |
| G | 500 | thermal | 1175°C Dry + 950°C Wet | none |
| Н | 500 | thermal | not available | yes |
| I | 500 | thermal | 1175°C Dry + 950°C Wet | 1175°C NO |

 Table 4-1: Lateral test structure sample sets used to obtain experimental results presented

 in Chapters 4 and 5.



Figure 4-1: Image of the lateral 4H SiC MOS test chip, the main test vehicle for this study.



Figure 4-2: Lateral 4H SiC MOS test chip layout with main structures of interest highlighted.

Thus, several different process splits were available for comparison: thermally grown oxides vs deposited (LPCVD) oxides, NO post-oxidation anneal vs no nitridation, and as-grown SiC epi vs implanted SiC.

The gate-bias stressing experiments were performed primarily on lateral 4H SiC MOSFETs with enclosed circular geometries. Because these devices have an enclosed geometry, they did not suffer from parasitic edge leakage and therefore allowed for lower subthreshold current measurements that were important when calculating the subthreshold swing, which will be discussed in the following chapter. Subthreshold swing analysis is useful in determining an upper bound for the interface trap density, and the change in this slope is a measure of the number of oxide trap states that have filled or emptied during the course of the measurement itself, providing a lower bound for the

total oxide trap density. A clean subthreshold current measurement is also useful in extracting the threshold voltage using the inversion current method, as mentioned below. Low edge leakage is also important when making charge-pumping measurements, which is another measurement technique that can provide insight regarding the number of oxide traps and interface traps present.

4.3 Experimental procedure

The standard threshold-voltage instability measurement was performed as follows (see Figure 4-3). Initially, the gate voltage was ramped from negative to positive bias, with 50 mV applied to the drain, using an Agilent 4155 parameter analyzer. The fastest sweep possible with reasonable resolution is around 1 s. It should be noted that once the device has been bias stressed, the initial sweep may be dependent on its previous bias history. Next, a positive bias was applied to the gate for the desired bias-stress time while the other terminals were grounded, again using the Agilent 4155. Following the termination of the positive-bias stress, the gate voltage was ramped from positive to negative bias to determine the shift of the drain current-gate voltage (I_D - V_{GS}) characteristic. A negative bias equal in magnitude but opposite in polarity was then applied to the gate for the same bias-stress time. Finally, the gate voltage was once again ramped from negative to positive bias to measure the return shift.



Figure 4-3: Schematic indicating the applied gate voltage versus time during the instability measurements.

Unless mobile ions are involved, a positive-bias stress will result in a positive shift in V_T and a negative-bias stress will cause a negative shift in V_T . This effect can be explained by electrons filling or emptying near-interfacial oxide traps via direct tunneling in response to the electric field. Typically three or four full cycles are applied to the device under test. V_T can be readily calculated by linear extrapolation of the maximum slope to zero current, or by choosing a current level such as the calculated inversion current which occurs in the subthreshold region and then finding the corresponding voltage. The average shift in V_T is then calculated. It was found that either method for calculating V_T results in a similar V_T instability value, and that this measurement is very repeatable, as is discussed below [38].

Because of the linear-with-log-stress-time response described below, I suspected that a tunneling mechanism that is very sensitive to the measurement time was involved. Therefore, I encouraged colleagues at NIST to make measurements with their fast *I-V*

system (a schematic of which is depicted in Figure 4-4) that they developed, which can be used to ramp the gate voltage as quickly as 20 μ s. The details of this measurement system, which was developed to look at charge traps in high-*k* dielectric materials and that I realized would be useful in the study of oxide traps in SiC MOSFETs, are described elsewhere [76]. Here I will simply mention that a pulse generator is connected to the gate terminal and a fast op-amp is employed to measure the drain current without changing the drain bias. Both the drain current and gate voltage signals are captured by a digital oscilloscope. The main limitation is that the *I-V* sweep is limited to the voltage range during the stress and that the drain bias is also applied during the stress. However, because it is limited to 100 mV, it should not lead to hot carrier injection.



Figure 4-4: Schematic of fast *I-V* system employed at NIST.

Similarly, *p*-type MOS capacitors were measured and bias-stressed using an HP 4194 Impedance/Gain-Phase Analyzer. In this case, the gate bias was initially swept up in voltage from accumulation to inversion. The flat-band voltage, V_{FB} , was estimated from the ideal flat-band capacitance, C_{FB} . The *C*-*V* characteristics took a little longer to measure, with each measurement sweep typically taking 2 to 10 seconds to complete.

FATFET *n*-channel MOSFETs were also measured as capacitors in some cases, to confirm the *p*-type MOS capacitor results, by slightly biasing the source and drain positively with respect to the substrate.

Unless otherwise indicated, all the measurements presented in this work were performed at room temperature.

4.4 Basic experimental effect of gate-bias stress induced threshold-voltage instability

Figure 4-5 shows a series of *I-V* characteristics, plotted on both a log (on the left) and linear (on the right) current scale, measured using the Agilent 4155 with a 1-s sweep time after both positive and negative-bias stressing, as described above for an enclosed geometry device from Sample Set D. Three sweeps are apparent for each scale—the initial sweep, all the sweeps following positive-bias stressing, and all the sweeps following negative-bias stressing. In fact, there are twelve curves for each post bias-stressing sweep. In this case, each gate-voltage bias stress lasted for 1,000 seconds, with $\pm 10V$ applied to the gate of a 130 nm deposited oxide MOSFET.



Figure 4-5: Graph of I_D - V_{GS} curves of a SiC MOSFET from Sample Set D with a 130-nm thick deposited gate oxide, plotted on both a log (left) and linear (right) current scale, for a series of twelve full cycles of bias stress (±10 V for 1,000 seconds at each polarity) and measurement as indicated in Figure 4-4. Positive-bias stress causes a shift of the *I*-*V* characteristic to the right and a negative-bias stress causes a shift of the *I*-*V* characteristic back to the left.

Figure 4-6 plots the extracted threshold voltage (for both the linear and inversion current methods) versus cumulative stress time. As can readily be seen, the effect of the alternate positive and negative-bias stress is to repeatedly shift the threshold voltage back and forth, with increasing V_T following a positive-bias stress and decreasing V_T following a negative-bias stress. The linear V_T extraction consistently yields a more positive value than does the inversion current method.



Figure 4-6: Calculated V_T instability of a SiC MOSFET from Sample Set D using both linear extrapolation and inversion current methods following each bias stress and measurement versus cumulative stress time.

Figure 4-7 plots the magnitude of the threshold voltage back-and-forth instability for each bias stress as a function of cumulative stress time using the data of Figure 4-6. In this case it is clear that although the two methods extract different values for V_T , the back-and-forth instability calculated is almost identical, varying by less than ten percent. These instabilities are much larger than those usually observed in SiC MOSFETs since the gate oxide is much thicker for these samples, and the voltage shift is proportional to the oxide thickness (see (3-32)). Thus, the data from these samples very clearly demonstrates this instability phenomenon, which has been observed in SiC MOSFETs with both thermal as well as deposited gate oxides. By comparison, it should be noted that instabilities in as-processed Si MOSFETs with similar oxide thicknesses are typically only a few mV at room temperature.



Figure 4-7: Calculated change in V_T of a SiC MOSFET from Sample Set D using both linear extrapolation and inversion current methods following each bias stress and measurement versus cumulative stress time.

It has also been observed that the result is affected by the measurement. Therefore, it is important how the measurement is made. For example, the direction and extent that the gate is swept during the measurement can make a big difference. The importance of sweeping down in gate bias following a positive-bias stress is illustrated in Figure 4-8, which shows the variation in the back-and-forth V_T instability of a SiC MOSFET from Sample Set A, depending on whether the gate bias is swept down or up following a positive-bias stress. This difference occurs due to the strong effect of the gate bias during the measurement, during which time an additional effective bias stress is being applied. When the bias is swept down following a positive gate bias stress, the bias during the sweep is not that much different than that of the stress and therefore oxide traps do not readily change charge state. On the other hand, when the bias is swept up following a positive gate bias stress, the initial bias during the sweep is very much different than that of the stress and more oxide traps will change charge state. How much they change charge state is also a function of the sweep time, as will be discussed in detail in Chapter 5. Also note that the curves following a negative gate-bias stress are basically identical. This reiterates the repeatability of the basic effect.



Figure 4-8: Variation in the back-and-forth V_T instability of a SiC MOSFET from Sample Set A, depending on whether the gate bias is swept down or up following a positive-bias stress.

4.5 Effect of bias-stress magnitude

Figure 4-9 shows a plot of the V_T instability (calculated using the inversion current method) for a device from Sample Set D for a variety of different gate-bias stresses, ranging from ±2.5 V up to ±20V versus the bias-stress time, plotted on a log-time scale. The magnitude of the instability clearly increases with the magnitude of the

gate bias stress applied. Although there is noise in this data, which is the result of averaging three bias stresses at each polarity, the overall trend appears to be linear with log time. The ± 5 V data looks particularly clean, and is taken out to 100,000 seconds (in other words, to obtain that last data point the bias stress was held for over a day at each polarity before switching). No obvious saturation is observed. The early time data (bias stresses lasting for less than 100 seconds at a time) is strongly influenced by the measurement itself, leading to sub-linear-with-log-time behavior—especially for this data set, for which the Agilent 4155 ramp time had not been optimized and thus took closer to 10 s to complete. This linear-with-log-time behavior is consistent with charge tunneling into and out of oxide traps that are distributed uniformly spatially into the oxide from the SiC/SiO₂ interface.



Figure 4-9: Change in V_T of a SiC MOSFET from Sample Set D with a 130-nm thick deposited gate oxide, calculated using inversion current method, versus log stress time for various applied gate biases for a SiC MOSFET with a thick deposited gate oxide.

Figure 4-10 re-plots the data from Figure 4-9, but displays the actual calculated back-and-forth threshold voltage values, showing that for this sample with a deposited oxide, there is a much greater sensitivity to the bias magnitude under negative bias than under positive bias. This information is useful in trying to determine the energy level of the near-interfacial oxide traps, as will be discussed in detail in Chapter 6.



Figure 4-10: Data from Figure 4-9 re-plotted showing variation in V_T of a SiC MOSFET from Sample Set D using inversion current method as a function of both bias magnitude and polarity. This device is older, with a thick deposited oxide.

Figure 4-11 shows the results of a similar experiment, but in this case it is for an enclosed geometry MOSFET from Sample Set I, whose gate oxide was thermally grown. In this case, the much greater sensitivity to the magnitude of the bias occurs under positive bias. Devices from Sample Set H (both with and without an NO post-oxidation

anneal) which also had a thermal oxide were tested in a similar fashion and exhibited similar results. It is interesting to note that although a thermal oxide is more sensitive to positive bias and a deposited oxide is more sensitive to negative bias, both types of oxides exhibit a similar overall gate-bias stress-induced V_T instability. This may possibly be due to different energy levels for their respective oxide traps.



Figure 4-11: Variation in V_T of a SiC MOSFET from Sample Set I with a newer, thermal oxide using inversion current method as a function of both bias magnitude and polarity. This devices is more recent, with a less thick thermal oxide.

Figure 4-12 shows the I_D - V_{GS} instability characteristics of an enclosed geometry 4H-SiC lateral *n*-channel MOSFET from Sample Set F, which has a 60-nm thick thermal gate oxide and received a standard post-oxidation nitrogen anneal using NO, measured using the Agilent 4155 and its 1-s ramp speed.



Figure 4-12: Graph of I_D - V_{GS} curves of a lateral 4H-SiC MOSFET from Sample Set F with a 60-nm thick thermal oxide with an NO post-oxidation anneal, plotted on both a log (left) and linear (right) current scale, for three full cycles of bias stress and measurement (±2 MV/cm, 34-ks stress, 1-s ramp).

Figure 4-13 shows the average threshold-voltage instability for various bias-stress times and gate oxide fields (the results from Figure 4-12 are indicated in the upper right data point of Figure 4-13: individual bias stress time of 34,000 s with a gate oxide field stress of ± 2 MV/cm, due to an effective gate-bias stress of ± 13 V). Once again, a linearwith-log-stress-time response of the V_T instability with applied bias is observed, consistent with electrons tunneling directly from the SiC into oxide traps that are distributed rather uniformly with distance into the insulator. The longer the bias stress time, the deeper into the oxide the tunneling front can reach (see Chapter 6 for a full discussion of the tunneling mechanism). The greater the bias, the more effectively the tunneling process occurs. Under reverse bias, the charge is presumably tunneling out in a similar manner. The number of traps that are switching charge state to create the instability observed for a 1,000 s bias stress with a ± 1.5 MV/cm gate oxide field applied was found for this sample and measurement to be about 1.0×10^{11} cm⁻² oxide traps:

$$N_{OT} \cong \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 600 \times 10^{-8} \text{ (cm)}} \cdot 0.27 \text{ (V)} = 0.97 \times 10^{11} \text{ cm}^{-2}$$
(4-1)

The number of traps that are switching charge state to create the instability observed for a 1,000 s bias stress with a ± 1.5 MV/cm gate oxide field applied for the device from Sample Set D shown in Figure 4-9 is calculated to be about 2.0×10^{11} cm⁻² oxide traps:

$$N_{OT} \cong \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 1300 \times 10^{-8} \text{ (cm)}} \cdot 1.23 \text{ (V)} = 2.04 \times 10^{11} \text{ cm}^{-2}$$
(4-2)

The greater number of oxide traps in the older sample may be due to the less sophisticated post-oxidation nitride anneal—using NH₃ instead of NO, the fact that it is a deposited oxide, or simply that the oxidation process was not as optimized.



Figure 4-13: Average measured V_T instability of a lateral 4H-SiC MOSFET from Sample Set F with a 60-nm thick thermal oxide with an NO post-oxidation anneal for various bias stresses and times, and 1-s measurement time.

A similar V_T instability of a lateral 4H-SiC MOSFET from Sample Set G with a 50-nm thick thermal oxide with an NO post-oxidation anneal was observed for various bias stresses and times. The ±1.5 MV/cm response should lie somewhere between the ±5 V and ±10 V responses, so that the number of traps that are switching charge state to create the instability observed for a 1,000 s bias stress would be about 1.6×10^{11} cm⁻² oxide traps:

$$N_{OT} \cong \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 500 \times 10^{-8} \text{ (cm)}} \cdot 0.36 \text{ (V)} = 1.55 \times 10^{11} \text{ cm}^{-2}$$
(4-3)

4.6 Effect of the Post-oxidation NO Anneal on *V_T* Instability

Figure 4-14 shows the average instability versus gate-bias stress time for several different relative state-of-the-art 4H-SiC MOSFETs from Sample Set G with enclosed geometries and 50-nm thick thermal gate oxides that were thermally grown. The applied bias in each case was ± 15 V, and measurement times were about 1 s.

The bottom curve is the response of a control sample that received the standard post-oxidation NO anneal and had an as-grown epi channel. The top curves are the response of a sample that did not receive the post-oxidation anneal. Clearly, the lack of an NO anneal has a strong effect on the magnitude of the V_T instability, resulting in an instability about three times as great. One measurement set was taken out to bias-stress times of 320,000 s, which is over three and-a-half days for each individual bias stress time. Even at these very long times, the magnitude of the instability continues to increase! The total number of oxide traps switching charge state in this case is over 6×10^{11} cm⁻² oxide traps:

$$N_{OT} \cong \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 500 \times 10^{-8} \text{ (cm)}} \cdot 1.43 \text{ (V)} = 6.17 \times 10^{11} \text{ cm}^{-2}$$
(4-4)

The middle curve is the response of a sample that also received the NO postoxidation anneal, and whose channel region received an implant to mimic the implanted channel of the DMOSFET (DMOSFET results will be discussed in full in Chapter 7). The result is an instability that is about 33% greater than for the unimplanted sample.



Figure 4-14: Average measured V_T instability of variously processed lateral 4H-SiC MOSFET with 50-nm thick thermal oxides from Sample Set G. The unimplanted device with an NO post-oxidation anneal showed the smallest effect while that without an NO anneal had the largest effect, when stressed with a gate oxide field of ± 3 MV/cm.

The device without the NO anneal had a far larger V_T instability—about three times larger. So the NO anneal decreases the number of switching oxide traps that we observe. This result is consistent with the work of Afanas'ev, et al. [77], who found that the NO anneal reduced the number of slow electron traps that they attributed to defects located in the near-interfacial oxide layer, as well as recent work by Gurfinkel, et al. [106]]. It is interesting to note that previous work on nitrided oxides on Si found that nitridation also appeared to reduce the number of E'-type defects [107].

But it is well known that the NO anneal reduces measured interface traps as well [69, 70, 71]. Recent frequency-dependent charge pumping results also indicate that oxide

traps as well as interface traps are reduced by the NO anneal [108]. Interface traps should have a time response in SiC MOSFETs that is faster than that of the measurement times discussed in this work. So the fact that the instability effect increases with increasing stress time indicates that oxide traps are exchanging charge with the SiC. But the tunneling model suggests that those oxide traps which are close to the interface will respond on the same time-scale as the interface traps. It is therefore quite likely that traps near the interface are affecting the interface trap measurements as well. Although it may not make a difference operationally, and it may be impossible to de-convolve the two effects: oxide traps which vary with distance into the oxide and interface traps that vary with energy, there may still be two or more physically distinct defects. For example, recent magnetic resonance studies suggest that the dominant interface trap in 4H-SiC MOSFETs with thermal oxides is a Si vacancy in the SiC near the interface [60, 61]. This is a trap in the SiC, not in the oxide, and is consistent with EELS and TEM results [54, 56].

4.7 One-way instability measurements

In addition to the "back-and-forth" instability measurements described in Section 4.3 and discussed throughout this work, "one-way" instability measurements have also been performed. In this case, the polarity of the gate-bias stress is maintained. For a positive-bias stress as depicted in Figure 4-15, the stress time between measurements increases at a linear-with-log-time rate, and the gate bias is always swept down. (For a one-way negative-bias stress sequence, all the measurements would be made with the gate bias always swept up.)

80

Figure 4-15 shows how following a long-term negative gate-bias stress of 10^5 s, a 10-s positive gate-bias stress results in a significant shift to the right. Subsequent positive gate-bias stresses result in a further linear-with-log-time shift to the right. This linear-with-log-time shift is explicitly shown in Figure 4-16.



Figure 4-15: I_D - V_{GS} characteristics depicting a one-way V_T instability of a lateral 4H-SiC MOSFET with 50-nm thick thermal oxide and post-oxidation NO anneal from Sample Set G.



Figure 4-16: Plot of linear-with-log-time one-way V_T instability taken from Figure 4-15 for an *n*-channel MOSFET from Sample Set G.

These results are once again consistent with electrons tunneling to and from nearinterfacial oxide traps. The large initial shift is due to many decades of tunneling compressed into the first measurement (see Section 5.2), with the subsequent additional shifts due to the change in oxide trap charge states over a specific oxide depth (see Section 6.2). This linear-with-log-time shift is also observed under negative-bias stress [109] and explains the linear-with-log-time response of the back-and-forth instability measurements.

4.8 Capacitor flatband-voltage instability

The instability in *C-V* characteristics of a *p*-type SiC MOS capacitor with a deposited oxide from Sample Set C, shown in Figure 4-17, illustrates that this instability effect is not limited to SiC MOSFETs, implying that inversion carriers are not necessarily required. As expected, a positive-bias stress causes a shift to the right and a negative-bias stress causes a shift to the left. Once again, the direction of the gate sweep during the measurement is determined by the polarity of the bias: sweep down following a positive-bias stress and sweep up following a negative-bias stress.



Figure 4-17: Back-and-forth instability in the C-V characteristics of a SiC MOS capacitor from Sample Set C due to back-and-forth gate-bias stressing (-15V and +10 V).

Figure 4-18 shows a comparison of the V_{FB} instability of two different *p*-type SiC MOS capacitors, one from Sample Set B with a thermal oxide and one from Sample Set C with a deposited oxide, versus various gate-bias stress times, again plotted on a log time scale. The capacitors were alternately stressed with +10 V and -15 V. The thermally grown oxides were approximately 60-nm thick and were previously determined to have an oxide trapped charge density of 1×10^{12} cm⁻² [72]. The 0.6 V instability at 10,000 seconds represents a change in charge of about 2×10^{11} cm⁻², or 20 percent of the total "fixed" or oxide trapped charge present—even at room temperature. Interestingly, although the deposited oxide is thicker (around 100 nm), because it experiences a slightly larger shift and the total oxide trapped charge density was found to be only about 0.8×10^{11} cm⁻², the fraction of charge responsible for the shift is also about 20 percent in this case.



Figure 4-18: Comparison of the change in flat-band capacitance versus log stress time for SiC MOS capacitors with a deposited and thermal oxide with the same bias stress conditions (–15V and +10 V).

Figure 4-19 shows the actual variations in the flat-band voltage due to both positive and negative gate-bias stressing for the *p*-type SiC MOS capacitor with the deposited oxide from Sample Set C whose total back-and-forth instability was shown in Figure 4-18. Clearly, this MOS capacitor is more sensitive to the negative-bias stress. This result is consistent with that of the MOSFET with a deposited oxide whose response is shown in Figure 4-10. Just as Figure 4-11 showed a different response for a thermal oxide, so too the response of a thermal oxide from Sample Set B whose back-and-forth instability is also depicted in Figure 4-18 is different. In this case, there is a rough balance, with each bias polarity contributing approximately half of the shift in flat-band voltage.



Figure 4-19: Variation in flat-band voltage for a SiC MOS capacitor with a deposited oxide from Sample Set C due to back-and-forth gate-bias stressing (-15V and +10 V).

More recent results by Habersat and Lelis on both *p*-type capacitors and FATFET *n*-channel MOSFETs with thermal gate oxides from Sample Set G exhibit similar instabilities when the source and drain of the MOSFETs are biased slightly positive to the substrate, providing minority carriers to the inversion channel [109]. Although the flatband instabilities of the FATFETs were about 30 percent greater than that of the MOS capacitors, both of these instabilities are in line with the fast *I-V* and extrapolated V_T instability values presented in Figure 5-5, to be discussed in Chapter 5. They also showed that one-way V_{FB} instability measurements also have a linear-with-log-time response [109].

When comparing one-way V_T and V_{FB} instabilities under both positive and negative gate bias with similar measurement speeds, it was observed that a positive gatebias stress causes a larger shift in V_T and that a negative gate-bias stress causes a larger shift in V_{FB} (consistent with the results shown in Figure 4-19). This result is also consistent with the notion that the bias applied during the measurement affects that which is measured. V_T is measured with the gate bias slightly positive so that the bias applied during the measurement is of the same polarity as a positive gate-bias stress. Similarly, V_{FB} is measured with the gate bias negative so that the bias applied during this measurement is also of the same polarity as a negative gate-bias stress. To measure V_T following a negative-bias stress or V_{FB} following a positive-bias stress necessarily requires not just a change in bias but a change in polarity as well.

It was further observed by Habersat and Lelis [109] that the NO post-oxidation anneal only had a major effect on V_T under positive-bias stress. The NO anneal did not appear to have any effect on V_{FB} under either bias polarity stress. These results taken together may suggest that the NO anneal affects oxide traps in the upper half of the bandgap and that the flat-band measurements may be measuring traps lower in the bandgap. See Chapter 6 for further discussion of the possible energy levels of these oxide traps.

4.9 Effect of elevated temperature on the V_T instability effect

Figure 4-20 below shows a clean set of V_T instability data for an *n*-channel SiC MOSFET from Sample Set E as a function of temperature for a gate-bias stress time of 180 s and gate bias stress of \pm 10 V. This chart shows the threshold voltage determined

using the inversion current method, with the green bar indicating the pre-stress V_T at each temperature, the red bar indicating V_T following a positive-bias stress and the blue bar indicating V_T following a negative-bias stress. (It should be noted that "pre-stress" in this case simply means before bias stresses at that particular temperature. Clearly, the device had been bias stressed previously at lower temperatures. This simply provides context for the back-and-forth instability.) The overall trend in V_T is towards more negative values with increasing temperature, as would be expected due to both the increase in thermal carriers with temperature as well as the filling of fewer negatively charged interface traps at higher temperature. The magnitude of the threshold-voltage instability increases with increasing temperature.



Figure 4-20: Variation of threshold voltage following gate bias stressing as a function of temperature for an enclosed geometry 4H-SiC lateral MOSFET from Sample Set E.

The instability of the I_D - V_{GS} characteristic of a 4H-SiC n-channel MOSFET as a function of temperature is shown in Figure 4-21, for both T = 25 and 200 °C, following the bias and measurement sequence shown in Figure 4-3. For this typical device from Sample Set E, a positive-bias stress caused a positive shift in the threshold voltage and a negative-bias stress caused a negative shift, at both T = 25 and 200 °C. This is the response that we see at room temperature for all the samples, and the effect is very repeatable. Again, it is likely due to charges filling and emptying oxide traps near the SiC interface in response to the bias stress. This result is for one of the four different sample sets studied as a function of temperature (see Figure 4-22) that are fairly representative of the state-of-the-art, processed either from different sources or with variations in the processing (Sample Sets E, F, G, and H). Although the exact details are proprietary, all samples tested had thermal gate oxides with an NO post-oxidation anneal. The main difference appears to be the control of mobile ions in the oxide.



Figure 4-21: Instability of I_D - V_{GS} characteristics at both 25 and 200 °C for Sample Set E following bias stress and measurement sequence shown on the right.

Figure 4-22 shows the average threshold-voltage instability effect as a function of temperature for all four sample sets studied as a function of temperature. In this case, the gate bias stress time was three minutes (180 s), with a gate bias stress that produced an electric field across the gate oxide of about ± 2 MV/cm, with the full cycle of stress and measurements (see Figure 4-3) repeated three times and averaged. As can be seen in Figure 4-22, although the room temperature instabilities are fairly consistent ranging from +0.25 to +0.40 V, a wide range of responses is observed at temperatures above 100 °C. The instability increases dramatically above 175 °C for the device from Sample Set E, exceeding 1 V at 225 °C. This response is consistent with that shown in Figure 4-20.

The number of traps that are switching charge state to create the instability observed for Sample Set E vary considerably versus temperature. At room temperature, the number of oxide traps is found to be about 1.5×10^{11} cm⁻² oxide traps. The number of traps that are switching charge state to create the instability at 225 °C is found to be about 4.2×10^{11} cm⁻² oxide traps:

$$N_{OT} \cong \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 600 \times 10^{-8} \text{ (cm)}} \cdot 1.15 \text{ (V)} = 4.14 \times 10^{11} \text{ cm}^{-2}$$
(4-5)



Figure 4-22: Average threshold-voltage instability effect as a function of temperature for several different samples fairly representative of the state-of-the-art. All devices have thermally grown oxides that received a standard post-oxidation NO anneal.

On the other hand, the response becomes negative for the device from Sample Set F above 200 °C. This response is even more dramatic in the case of the device from Sample Set H, where the V_T instability effect becomes negative above 100 °C. The number of traps that are switching charge state to create the instability observed for Sample Set H at room temperature is found to be about 1.2×10^{11} cm⁻² oxide traps. This is on the low end of the range typically calculated from room-temperature gate-bias stressing. The negative V_T shift at 150 °C will not, by itself, give any sensible numbers for the number of traps that are switching charge state since clearly a second mechanism is at play.

Figure 4-23 shows the instability of the I_D - V_{GS} characteristic for a typical device from Sample Set H at both 25 and 150 °C. At 150 °C, the *I*-V curves shift in the opposite direction, moving more negative following a positive-bias stress and moving more positive following a negative-bias stress.



Figure 4-23: Instability of I_D - V_{GS} characteristics at both 25 and 150 °C for Sample Set H. The response at elevated temperature is the opposite of that at room temperature and is likely due to mobile ion drift.

Figure 4-24 shows the response of the device from Sample Set F in more detail, showing the average instability not only as a function of temperature, but of different bias-stress times as well. Not surprisingly, the longer the bias-stress time, the more exaggerated the V_T instability effect. It is important to note that the results shown in Figure 4-22 are for a stress time of 180 s. It is not until longer stress times that the full effect at 200 °C is observed. This effect is explored further in Chapter 7 on the DMOSFETs.



Figure 4-24: Avg. instability of Sample Set F as a function of temperature and bias-stress time.

Since a positive average shift is consistent with charge tunneling into and out of near-interfacial oxide traps, the observation of a negative average shift indicates the presence of a different mechanism. The effect of this mechanism is consistent with mobile ions moving in the oxide in response to the applied gate bias stresses. The two mechanisms are shown in Figure 4-25. The left side of the figure shows the case of a positive gate bias. Electrons are likely to fill traps and/or holes are likely to be emitted, in each case leading to a positive shift in V_T . On the other hand, positively charged mobile ions will drift towards the SiC interface and negatively charged mobile ions will drift towards the gate electrode. In each case the result is a negative shift in V_T , according to (3-47). Since interfacial charge trapping will lead to an increase in net negative charge at the SiC/SiO₂ interface and mobile ion drift will lead to an increase in net positive charge near the interface, it is clear that the two mechanisms will have opposite effects on

the threshold-voltage instability. It is useful to note that the presence of mobile ions in some oxides has also been reported in the literature [44].



Figure 4-25: Schematic demonstrating the opposing effects of bias stress on charge traps and mobile ions. For example, a positive gate-bias stress will cause a positive shift in the threshold voltage due to interfacial charge trapping, but will cause a negative shift in V_T due to mobile ion drift.

The large non-linear with-log-time V_T instability response as a function of temperature for devices from Sample Set E shown in Figure 4-20 and Figure 4-22, which show an exponential increase in instability with increasing temperature, suggest the presence of a third mechanism, since a direct tunneling process is not expected to have a strong temperature dependence. Although this may involve thermal de-trapping [110, 111], this mechanism cannot by itself explain the back-and-forth instability observed. It is likely that this increase in instability is also due, and maybe primarily due, to an increase in the number of active oxide traps, as discussed both in Chapter 2 in terms of the observed increase in E' centers (the only oxide defect observed in SiC MOS
structures) in Si MOS structures with increasing temperature [96], and in Chapter 7 in terms of explaining similar observed increases in V_T instability in similarly processed SiC power MOSFETs [51].

The slight decrease in V_T instability for the device from Sample Set E in Figure 4-22 between 25 and 100 °C may be due to the decrease in the number of filled interface traps at higher temperatures, if a two step tunneling mechanism involving interface traps exists—as discussed in Chapter 6. As mentioned above, the results above 100 °C wherein the instability once again increases can be explained by an increase in the number of active oxide traps.

Although the device from Sample Set G in Figure 4-22 appears to have an ideal response, it quite possible that this is achieved by a delicate balancing of the movement of mobile ions and the filling and emptying of near-interfacial oxide traps. If this is so, then it is of great importance to find charge separation methods such that it can be determined whether both types of defects exist so that both types of defects can be reduced. This is equivalent to stating that further improved testing methods are required to ensure reliable SiC MOSFET device operation (see Section 7.7 for a discussion of reliability testing of SiC power MOSFETs).

4.10 Summary

The basic experimental results of the threshold-voltage instability effect in SiC MOSFETs have been presented. It has been observed that a positive gate-bias stress causes a V_T shift to the right, a negative gate-bias stress causes a V_T shift back to the left, and that this effect is repeatable.

The longer the bias stress time, the larger the instability observed, and the increase proceeds at a linear-with-log-time rate. This result is consistent with electrons tunneling in and out of near-interfacial oxide traps that are distributed relatively uniformly with distance into the oxide (over the range that this can be measured). Linear-with-log-time responses are observed not only in the "back-and-forth" instability, but also "one-way" instability measurements under both positive and negative bias. Standard-room temperature gate-bias stressed V_T -instability results yield switching oxide-trap densities of 1×10^{11} to 2.5×10^{11} cm⁻².

The direction of the gate sweep during the measurement is important. Sweeping up in gate voltage (from negative or zero volts up to a positive bias) results in a smaller observed V_T shift than when sweeping down following a positive-bias stress. This is explained by the notion that the bias applied during the measurement also affects the charge occupation of these near-interfacial oxide traps.

Larger gate biases lead to larger shifts, although this effect is sub-linear. Modern devices with thermal gate oxides tend to be more sensitive to the magnitude of the positive gate bias. Older devices with deposited oxides showed greater sensitivity to the magnitude of the negative bias.

MOS capacitors and large-area MOSFETs measured as capacitors show similar flat-band voltage instabilities. The V_{FB} instability is significantly larger under one-way negative gate-bias stress sequences, whereas the V_T instability is significantly larger under one-way positive gate-bias stress sequences. However, the overall V_{FB} instability is larger than the V_T instability, at least when measured with a ramp speed of around 1-10 s. Chapter 5 will show how faster measurements reveal a much larger V_T instability.

96

A post-oxidation NO anneal significantly reduces the amount of V_T instability, although only due to positive-bias stress. SiC MOSFETs that did not receive the standard post-oxidation NO anneal have a much larger V_T instability, with switching oxide-trap densities as large as 6×10^{11} calculated with standard 1-s measurement speeds. The NO anneal does not affect V_{FB} instability. These results may indicate the presence of two distinct energy levels for the oxide traps, with the NO anneal only affecting the traps near the conduction band edge. This result is very similar to that observed for interface traps, and may be another indication of the difficulty of separating out the effects of oxide traps and interface traps for very fast measurements.

Various responses in V_T instability were observed with increasing temperature. Some devices exhibited an exponential increase in the V_T instability with increasing temperature, especially at long bias-stress times. One possible explanation for this is the activation of additional oxide traps above 100 °C. Switching oxide-trap densities as large as 4×10^{11} were calculated for measurement speeds of around 1 s.

A few, older devices exhibited a large shift in the opposite direction with increasing temperature, shifting to the left following a positive-bias stress and to the right following a negative-bias stress. This type of instability is almost surely due to the presence of mobile ions in the gate oxide.

Some devices show a relatively flat response with temperature. This may either indicate improved gate-oxide processing techniques, or simply an uneasy balance of charge trapping and mobile ion effects. It is important to perform charge separation techniques to either verify the improved oxide processing or to monitor the densities of both types of defects.

97

5 Measurement Time Dependence, Subthreshold Swing Analysis, and Charge Separation

5.1 Introduction

This chapter discusses the critical issue of the speed of the I_D - V_{GS} measurement, which is necessarily performed to determine the effect of the gate-bias stress. The measurement time matters since a bias is necessarily applied during the measurement. The slower the measurement, the longer the time during which a bias, different than that of the stress itself, is able to affect the charge states of the near-interfacial oxide traps.

It turns out that the speed of the measurement also determines whether and to what extent the subthreshold swing differs between a measurement wherein the gate is swept up following a negative-bias stress or swept down following a positive-bias stress. Again, this is because during slower measurements, the oxide traps can noticeably change charge state, contributing to the stretch-out of the subthreshold I_D - V_{GS} characteristic.

A calculation of a more accurate lower bound for the number of active oxide traps can then applied to the key question of determining the composition of the various types of interfacial charge that cause a shift in V_T both from its theoretical ideal value, as well as during a bias stress and measurement. The implications of a time-varying oxide trap component are discussed.

5.2 Effect of measurement speed on the observed threshold-voltage instability due to gate-bias stress

Figure 5-1 shows a linear-with-log-stress-time extrapolation of bias-stress induced threshold-voltage instability data versus individual stress times to early times and zero instability for a 4H-SiC n-channel MOSFET with a 130 nm-thick deposited gate oxide. (The relatively thick gate oxide for this sample leads to a significant amplification of the measured instability, as (3-33) implies. Thinner gate oxides have correspondingly smaller V_T instabilities.) The full measurement sequence depicted in Figure 4-3 was repeated three times for each stress time and bias plotted versus the average V_T instability. The measurement sweeps took about 1 s and the individual stress times varied from 10 to 100,000 s. A subset of the data from Figure 4-9 is shown for gate bias stresses of ± 5 V, ± 10 V, and ± 20 V, which correspond to electric fields with a magnitude of about 0.4, 0.8, and 1.5 MV/cm, respectively. When fitting a straight line to the instability data versus log time, the results for stresses that lasted less than 100 seconds were omitted, since the measurement times in this case were not much faster than the stress times, clearly affecting the result. One outlier data point for both the ± 10 V and ± 20 V curves was omitted as well.

Extrapolating the data in Figure 5-1 to early stress times predicts that we would have to reduce the stress time to less than 10 ps in order not to see any instability effect. These results are consistent with electrons directly tunneling into and out of oxide traps spatially distributed into the oxide from the SiC/SiO₂ interface. It is interesting to note that in irradiated Si MOS devices with a similar oxide thickness, a similar instability or "reversibility" was observed, as discussed in Chapter 2 [88]. In that case a "tunneling

front," wherein those traps in its wake have been either filled or emptied depending on the field and those beyond it yet unaffected, was determined to move at the rate of about two angstroms per decade of time into the oxide, with the first transitions occurring at about 0.1 ps, giving rise to the linear-with-log-time behavior for a uniform trap distribution [112, 113]. Electron tunneling to so-called E' centers (a broken Si-Si bond due to an oxygen vacancy) in the oxide was determined to be the dominant mechanism in that case [89, 93, 98]. The oxide trap in SiC MOS may well be due to more than one type of defect. Some traps could be associated with C in the transition region that may extend 5 nm or more into the oxide. Some of the oxide traps may be associated with the nitrogen introduced during the post-oxidation NO anneal. However, the only trap definitely identified in SiC is an E'-center type defect [61].

In order to extrapolate to zero instability, the data in Figure 5-1 were extrapolated over twelve decades of time. It is quite remarkable that the three different bias-stress extrapolations all meet within one decade of time—between 1 and 10 ps. Furthermore, it is very suggestive that the extrapolated zero instability times are very close to the initial tunneling transition times of 0.1 ps mentioned above for Si MOS. If this tunneling model to spatially distributed oxide traps is correct for SiC MOS, then Figure 5-1 implies that oxide traps extend at least 36 Å into the oxide from the Si interface, assuming that the tunneling front of 2 Å per decade of time for Si MOS applies to SiC MOS as well, which is reasonable. This distance is compatible with the oxide transition layers or regions for SiC MOS, which are believed to be at least 3-5 nm thick [56, 57, 59]. Figure 5-1 and the tunneling model also suggest that if bias-stress times on the order of microseconds could be applied along with equally fast measurements, then half the V_T instability effect

100

observed for the very long bias stress times of Figure 5-1 would occur in these extremely short times.



Figure 5-1: Linear-with-log-stress-time extrapolation of the bias stress-induced thresholdvoltage instability data versus individual stress times to early times and zero instability of a lateral 4H-SiC MOSFET from Sample Set D with a 130-nm thick deposited gate oxide with a nitrogen anneal. Selective data from Figure 4-9 was used.

This raises the question of the importance of the measurement ramp speed. It was already mentioned above that data for stress times that were close in magnitude to the measurement times were discarded since the measured instability fell dramatically away. The reason for this fall-off is likely that the gate bias applied during the measurement is in effect applying a new stress bias. The slower the gate bias ramp is, the longer that this effective additional bias stress is applied. This would not be terribly significant if the measurement bias was such that it reinforced the stress bias. However, for some portion of the gate sweep the measurement bias will oppose the stress bias. Given that significant shifts in V_T can occur in very short times, we would therefore expect that varying the measurement time will significantly affect the results. Consider the following thought experiment. Suppose that following the stress bias, an infinitely fast gate sweep could be applied—or at least one that was faster than the first tunneling transition times of 0.1 to 10 ps. In this case, the full effect of the bias stress on V_T would be seen. On the other extreme, suppose that following the stress bias, an infinitely slow gate sweep was applied instead. In this case, the effect of the stress bias would be negligible and only the steady state value for each bias stress along the gate-bias sweep would be measured. In this case, no instability would ever be measured. These results lie in between these two extreme cases, and the faster that the measurement can be made, the more of the effect of the bias stress will be observed.

To verify this, I, along with my colleagues at ARL and NIST at my behest, performed the following set of experiments on the same relatively state-of-the-art 4H-SiC lateral n-channel MOSFETs, mentioned with regard to the *I-V* curves of Figure 4-13, which had a thermal gate oxide and a standard post-oxidation nitrogen anneal using NO.

First, bias stresses were applied for long times and the V_T instability was measured using the Agilent 4155. The average threshold-voltage instability for various bias-stress times and gate oxide fields is shown in Figure 4-13. Once again a linear-withlog-stress-time response of the V_T instability with applied bias is observed, consistent with electrons tunneling from the SiC into oxide traps that are distributed rather uniformly with distance into the insulator. The longer the bias stress time, the deeper into the oxide the tunneling front can reach. The greater the bias, the more effectively the tunneling process occurs. Under reverse bias, the charge is presumably tunneling out in a

102

similar manner. The number of traps that are switching charge state to create the instability observed for a 3,400 s bias stress with a ± 2 MV/cm gate oxide field applied was found for this sample and measurement to be about 1.2×10^{11} cm⁻² oxide traps.

Next, to investigate the effect of measurement time, a similar sample was measured at NIST using their fast *I-V* system [76], with measurement times ranging from 1 ms to as quick as 20 μ s—much faster than the 1-s measurement time for the data in Figure 4-13. Figure 5-2 shows that the fast *I-V* system measured much greater instabilities, even though the bias stress times were 10 s or less and the applied fields were just over 1 MV/cm (around \pm 7 V). The 1.5 and 2.0 MV/cm curves from Figure 4-13 are re-plotted for comparison and appear in the lower right of Figure 5-2. Clearly, the gate bias applied during the measurement had an important effect on the result.

Also shown in Figure 5-2, extrapolating the linear-with-log-stress-time V_T instability response (ignoring stress times that once again are close to the measurement time) back to early stress times suggests that we would have to reduce the stress time to 1 ps or less to not see any instability effect. These extrapolations are consistent with those in Figure 5-1, though the zero instability stress time now varies over about four decades of time instead of just one. On the other hand, extrapolating the 20-µs measurement time data out to longer bias-stress times suggests that the V_T instability would be as great as 1 V for a stress time of 34 ks, which would imply an actual lower bound for N_{OT} of 3.6×10^{11} cm⁻² oxide traps:

$$N_{OT} = \frac{C_{OX}}{q} \cdot \Delta V_T = \frac{\varepsilon}{q \cdot t_{OX}} \cdot \Delta V_T$$

$$= \frac{3.9 \cdot 8.85 \times 10^{-14} \text{ (F/cm)}}{1.6 \times 10^{-19} \text{ (C)} \cdot 600 \times 10^{-8} \text{ (cm)}} \cdot 1.0 \text{ (V)} = 3.60 \times 10^{11} \text{ cm}^{-2}$$
(5-1)

This is three times the number calculated from the data for the much slower 1-s measurement!



Figure 5-2: Linear-with-log-time extrapolation of the bias stress-induced threshold-voltage instability data to early stress times and zero instability of a lateral 4H-SiC MOSFET from Sample Set F with a 60-nm thick thermal oxide with an NO post-oxidation anneal.

5.3 Stretch-out of the subthreshold I_D - V_{GS} characteristics—a method for determining ΔN_{OT}

Figure 5-3 shows the instability in the subthreshold region of an I_D - V_{GS} characteristic for a 4H-SiC lateral MOSFET, and is in fact a blow-up of the subthreshold

characteristics taken from Figure 4-5. It shows the result of twelve full cycles of biasing, where each cycle consists of first stressing under positive gate bias for 1,000 s at +10 V followed by a measurement sweeping down in gate voltage, and then stressing under negative gate bias at -10 V for an additional 1,000 s followed by a measurement sweeping back up in gate voltage. Although the particular result shown in Figure 5-3 is from a relatively thick 130-nm gate oxide from several years back and does not represent state-of-the-art gate oxides, the large instability in this case clearly demonstrates the basic effect.



Figure 5-3: Close-up of the subthreshold region of the I_D - V_{GS} characteristic from Figure 4-5 (Sample Set D). Clearly, the curve is more stretched out when sweeping up in gate bias following a negative-bias stress, leading to a greater value of *S*, the subthreshold swing.

It has been found in the course of this analysis of the threshold-voltage instability in as-processed SiC MOSFETs that not only does the I_D - V_{GS} characteristic shift to the right following a positive-bias stress and back to the left following a negative-bias stress consistent with a change in the number of filled oxide traps [23], but that the slope of the subthreshold *I-V* characteristic is more upright when measured by ramping the gate bias down in voltage following a positive-bias stress than the slope of the subthreshold *I-V* characteristic when ramping the gate bias up in voltage following a negative-bias stress [43, 47, 48, 105] (see Figure 5-3). This is likely due to near-interfacial oxide traps switching charge state during the measurement when a negative gate-bias stress is followed by a positive applied gate bias during the measurement when the surface potential varies from mid-gap to inversion. This implies that part of the effect of the bias-stress instability is mitigated by the measurement itself when comparing the change in voltage required to put the surface potential at inversion:

$$V_{inv_meas_down} - V_{inv_meas_up} = \frac{q \cdot [\Delta N_{OT_stress} - \Delta N_{OT_meas}]}{C_{OX}}$$
(5-2)

If instead the subthreshold current is extrapolated down to much lower currents, such as that when the surface potential is at mid-gap, then the change in voltage at that current level should provide a voltage instability value that is closer to that due only to the bias stress without the mitigating effect of the measurement:

$$V_{mg_meas_down} - V_{mg_meas_up} = \frac{q \cdot \Delta N_{OT_stress}}{C_{OX}}$$
(5-3)

A schematic of this analysis method is shown in Figure 5-4.



Figure 5-4: Schematic of the subthreshold swing analysis technique, comparing the measured subthreshold swing when the gate is swept up in bias following a negative-bias stress with the subthreshold swing when the gate is swept down in bias following a positive-bias stress.

Thus the analysis of the change in stretch-out of the subthreshold $I_D - V_{GS}$ characteristic can be used to characterize the number of switching oxide traps more directly, providing a larger lower bound for N_{OT} , the number of oxide traps present. This change in slope of the subthreshold *I-V* characteristic following gate-bias stressing is unlikely to be due to a change in the number of active interface traps since this is a low field stress (less than 1 MV/cm) in the case of Figure 5-3. In addition, the effect is very repeatable, and it is highly unlikely that interface traps are being repeatedly activated and passivated at room temperature under these low oxide fields.

Comparing (5-2) and (5-3) indicates that we would expect to see less than the actual number of oxide traps that change charge state due to the bias stress by the number of oxide traps that change charge state during the measurement when considering the change in inversion voltages. The change in the inversion voltage is what has been routinely reported in the literature when discussing the threshold-voltage instability [23]. The fast *I-V* measurements do not allow nearly as much time for changes in the number of occupied oxide traps during the measurement and thus it is reasonable to expect that they would yield values similar to those calculated from the change in mid-gap voltage, with the very important caveat that extrapolations of the slope of the subthreshold *I-V* characteristic are made over twenty orders of magnitude to get down to the theoretical mid-gap current levels for SiC, with few decades of actual data. (It should be noted that the fast *I-V* contains no subthreshold data because of the speed of that measurement.)

Figure 5-5 shows the application of this analysis to the data in Figure 5-2. Variation in the average threshold-voltage instability versus bias-stress time is plotted as a function of measurement time. Clearly, the average instability calculated using the shift in mid-gap voltage with a 1-s measurement is much larger than that calculated using the shift in inversion voltage, and of the same order as that found when measured in tens of microseconds using a fast *I-V* system. It is also clear that this method is not as exact, given the large extrapolations required, and the results using the change in mid-gap voltage are rather noisy. These extrapolated results provide a band or range rather than exact values. But it is striking that at least two-thirds of the actual V_T instability of more than 0.8 V caused by the gate-bias stress is obscured by the counter-shift in voltage that occurs during the slower measurement, just as the direct tunneling model and its linear-

108

with-log-time response would suggest [112, 113]. The number of switching oxide traps present calculated using either the fast *I-V* measurements or the mid-gap voltage extrapolation method give a much greater lower bound for N_{OT} of 3.6×10^{11} cm⁻² oxide traps—three times the number calculated using the inversion voltage method.



Figure 5-5: Comparison of the effect of measurement time on the calculated instability of the threshold voltage for 4H SiC MOSFETs from Sample Set F with an NO anneal (modification of Figure 5-2). The 1-s measurements (lower right-hand corner) had adequate subthreshold data to calculate the change in mid-gap voltage, which provides a better (and higher) estimate of a lower bound for the number of switching oxide traps, comparable to that from the fast *I-V* measurements.

Figure 5-6 re-plots the data from Figure 4-7 where a 4H SiC MOSFET with a deposited oxide ~1300 Å thick was subjected to twelve full cycles of bias stressing. The

linear and inversion current extraction methods give similar values for V_T , whereas the change in mid-gap voltage indicates a much larger actual instability present. The calculated shifts (using both the mid-gap and inversion voltage methods) are much larger for this sample compared to that in Figure 5-5 because of the much thicker gate oxide, as well as being older and less state-of-the-art. But once again, the change in mid-gap voltage is at least three times as large as the change in inversion voltage commonly used to calculate the V_T instability. However, the uncertainty in the extrapolation based on only a few decades of subthreshold current is also very evident in Figure 5-6 as well.



Figure 5-6: Comparison of mid-gap and inversion voltage methods for calculating V_T instability (modification of Figure 4-7 using devices from Sample Set D). Note the noise in the mid-gap calculation due to the large extrapolation.

Figure 5-7 re-plots the change in V_T calculated using the inversion current method shown in Figure 4-6, but with the change in the subthreshold swing, *S*, added. This figure indicates that following a negative-bias stress and a sweep up in gate voltage, V_T is observed to shift more negatively and the subthreshold swing value, *S*, increases, which means that the I_D - V_{GS} characteristic is more stretched out. Likewise, following a positive-bias stress and a sweep down in gate voltage, V_T is observed to shift more positively and the subthreshold swing value, *S*, decreases, which means that the I_D - V_{GS} characteristic is more upright. This is consistent with the subthreshold curves of Figure 5-3. Again, this effect is very repeatable.



Cumulative Stress Time (seconds)

Figure 5-7: Calculated V_T using the inversion current method from Figure 4-6 (Sample Set D), along with the calculated subthreshold swing value, *S*, plotted on the seconday *y*-axis. When V_T shifts more negatively following a negative-bias stress, the subthreshold *I-V* characteristic is also more stretched out (larger *S* value).

Figure 5-8 is a plot of the related changes in charge trap density during the twelve full cycles of bias stress and measurement, calculated from the data of Figure 5-7. The change in oxide trap charge calculated from the change in inversion voltage yields a

value of about 1.5×10^{11} traps per cm². The value calculated from the change in mid-gap voltage yields a value in the range of $5.5 \pm 0.3 \times 10^{11}$ traps per cm². This is because about 4×10^{11} traps per cm² are changing charge state during the measurement. Also plotted on this graph is the number of interface traps, using the relationship of (5-35) discussed below in Section 5.4, assuming for the sake of argument that the entire deviation from the ideal subthreshold swing is due to interface traps. The lower set of values for N_{IT} is obtained following a positive-bias stress, when the slope of the subthreshold I-Vcharacteristic is more upright. The higher set of values is following a negative-bias stress, when the slope of the subthreshold *I-V* characteristic is more stretched out due to the likely added factor of oxide traps changing charge state. This difference in the set of values for N_{IT} is also seen to be about 4×10^{11} traps per cm², as would be expected. This method does not necessarily provide exact values, but may give approximate values that provide some useful information regarding an upper bound for the number of interface traps present since other factors can also contribute to a deviation from the ideal subthreshold swing, including lateral non-uniformities [114]—as well as some oxide traps possibly changing charge state while sweeping down. Regardless, given that the magnitude of the threshold-voltage instability keeps increasing with stress times as great as 3×10^5 s (see Figure 4-14), and that the subthreshold swing changes back and forth depending on the stress and measurement conditions, these are strong arguments that these effects are not due to changes in the number of interface traps but instead are due to oxide traps filling and emptying during both the bias stress and the measurement that follows. Subtracting out the ΔN_{OT} component indicates that the number of interface traps

112

(the smaller values calculated when sweeping down following a positive-bias stress) are approximately equal in magnitude to the number of switching oxide traps in this case.



Figure 5-8: Changes in calculated charge trap density when subject to alternating positive and negative gate-bias stresses using the data from Figure 5-7 (Sample Set D). The N_{OT} calculated varies depending on whether the change in inversion voltage or mid-gap voltage is used, and the N_{IT} calculated is distorted by the change in the slope of the *I-V* characteristic due to the change in oxide trap occupation during the measurement.

5.4 Derivation of the subthreshold-swing analysis

This section provides a theoretical justification why the analysis of the subthreshold swing from the subthreshold region of the I_D - V_{GS} characteristic can provide useful insight into the charge trapping effects of SiC MOSFETs. Subthreshold-swing analysis was successfully applied to the analysis of radiation effects in Si MOS a

generation ago [115, 116] and the general concept of the subthreshold swing is briefly discussed in Sze [117].

The subthreshold swing, *S*, can be defined as follows:

$$\frac{1}{S} = \frac{d\log(I_D)}{dV_{GS}}$$
(5-4)

which is the inverse of the slope. The subthreshold current is derived in Sze [117]:

$$I_{D} = \mu_{n} \cdot \left(\frac{Z}{L}\right) \cdot \frac{\varepsilon_{s}}{\sqrt{2} \cdot L_{D} \cdot \beta^{2}} \cdot \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(1 - e^{-\beta \cdot V_{DS}}\right) \cdot e^{\beta \cdot \psi_{s}} \cdot \frac{1}{\sqrt{\beta \cdot \psi_{s}}}$$
(5-5)

where

$$\beta = \frac{q}{kT} \tag{5-6}$$

 μ_n is the electron mobility, Z is the channel width, L is the channel length, and the other terms are as previously defined. The surface potential, ψ_S , is equal to ϕ_B for midgap and $2 \cdot \phi_B$ for inversion. The gate voltage required to put the surface potential at inversion, V_{inv} , is one way to define the threshold voltage (as briefly mentioned in Chapter 4), and the corresponding current can be referred to as the inversion current, I_{inv} . Similarly, the gate voltage required for mid-gap surface potential is the mid-gap voltage, V_{mg} , and the corresponding current is the mid-gap current, I_{mg} . (See Figure 5-9 for a schematic of the subthreshold swing and a summary of the equations derived below.)



Figure 5-9: Schematic comparing the ideal subthreshold swing with an actual measured subthreshold swing.

Therefore,

$$\frac{1}{S} = \frac{\log(I_{inv}) - \log(I_{mg})}{V_{inv} - V_{mg}} = \log\left(\frac{I_{inv}}{I_{mg}}\right) \cdot \frac{1}{V_{inv} - V_{mg}}$$
(5-7)

where I_{inv} and I_{mg} can then be re-written in terms of the subthreshold current expression given in (5-5) so that

$$\log\left(\frac{I_{inv}}{I_{mg}}\right) = \log\left(\frac{\mu_{n} \cdot \frac{Z}{L} \cdot \frac{\varepsilon_{s}}{\sqrt{2} \cdot L_{D} \cdot \beta^{2}} \cdot \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(1 - e^{-\beta \cdot V_{Ds}}\right) \cdot e^{\beta \cdot 2 \cdot \phi_{B}} \cdot \frac{1}{\sqrt{\beta \cdot 2 \cdot \phi_{B}}}}{\mu_{n} \cdot \frac{Z}{L} \cdot \frac{\varepsilon_{s}}{\sqrt{2} \cdot L_{D} \cdot \beta^{2}} \cdot \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(1 - e^{-\beta \cdot V_{Ds}}\right) \cdot e^{\beta \cdot \phi_{B}} \cdot \frac{1}{\sqrt{\beta \cdot \phi_{B}}}}\right)$$
(5-8)

which simplifies to

$$\log\left(\frac{I_{inv}}{I_{mg}}\right) = \log\left(\frac{e^{\beta \cdot 2\cdot\phi_{B}} \cdot \frac{1}{\sqrt{\beta \cdot 2\cdot\phi_{B}}}}{e^{\beta \cdot \phi_{B}} \cdot \frac{1}{\sqrt{\beta \cdot \phi_{B}}}}\right) = \log\left(\frac{e^{\beta \cdot \phi_{B}}}{\sqrt{2}}\right)$$
(5-9)

so that

$$\frac{1}{S} = \log\left(\frac{I_{inv}}{I_{mg}}\right) \cdot \frac{1}{V_{inv} - V_{mg}} = \log\left(\frac{e^{\beta \cdot \phi_B}}{\sqrt{2}}\right) \cdot \frac{1}{V_{inv} - V_{mg}}$$
(5-10)

Written in terms of the natural log, and given that $\beta = \frac{q \cdot \phi_B}{k \cdot T} > 50$ at room temperature,

and is >> $\frac{1}{2} \cdot \ln(2)$, this expression can then be written as: $\frac{1}{S} = \frac{1}{\ln(10)} \cdot \ln\left(\frac{e^{\beta \cdot \phi_B}}{\sqrt{2}}\right) \cdot \frac{1}{V_{inv} - V_{mg}} = \frac{1}{\ln(10)} \cdot \left[\ln(e^{\beta \cdot \phi_B}) - \ln(\sqrt{2})\right] \cdot \frac{1}{V_{inv} - V_{mg}}$ $= \frac{1}{\ln(10)} \cdot \left[\beta \cdot \phi_B - \frac{1}{2}\ln(2)\right] \cdot \frac{1}{V_{inv} - V_{mg}} \approx \frac{\beta \cdot \phi_B}{\ln(10)} \cdot \frac{1}{V_{inv} - V_{mg}}$ (5-11)

Therefore,

$$S \approx \frac{V_{inv} - V_{mg}}{\left(\frac{q}{k \cdot T} \cdot \phi_{B} / \ln(10)\right)}$$
(5-12)

The relationship given on p. 447 of Sze [117] for the subthreshold swing is defined in terms of the various capacitances present. For the idealized case of no interfacial trapped charge,

$$S \approx \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left[\frac{C_{OX} + C_D}{C_{OX}}\right]$$
(5-13)

where C_D is the depletion capacitance. When interface trap charge is present

$$S \approx \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left[\frac{C_{OX} + C_D + C_{IT}}{C_{OX}}\right]$$
(5-14)

In the case of SiC MOSFETs, there is likely an additional component due to changes in the oxide trap charge component with applied bias, leading to an additional capacitive term:

$$S \approx \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right]$$
(5-15)

Since both expressions (5-12) and (5-15) are equal to the subthreshold swing,

$$\frac{V_{inv} - V_{mg}}{\begin{pmatrix} \frac{q}{k \cdot T} \cdot \phi_B \\ / \ln(10) \end{pmatrix}} = \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right]$$
(5-16)

or

$$V_{inv} - V_{mg} = \phi_B \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right]$$
(5-17)

So for the non-idealized case that we can actually measure we can write

$$V_{inv_{meas}} - V_{mg_{meas}} = \phi_B \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right]$$
(5-18)

and for the idealized case we can write

$$V_{inv_ideal} - V_{mg_ideal} = \phi_B \cdot \left[\frac{C_{OX} + C_D}{C_{OX}}\right]$$
(5-19)

If we further assume mid-gap interface trap neutrality, such that the interface traps below mid-gap are positively charged when empty and neutral when filled and interface traps above mid-gap are neutral when empty and negatively charged when filled, then when the surface potential is at mid-gap there is no charge contribution from any of the interface traps. This assumption worked very successfully in the analysis of irradiated Si MOSFETs. If this is so for SiC MOSFETs as well, which is not an unreasonable assumption, then the deviation in the mid-gap voltage from the ideal value is a measure of the total number of oxide traps present. This is not an easy number to pin down since it changes with applied bias and time, as we have shown. Nonetheless, we can write

$$V_{mg_ideal} - V_{mg_meas} = \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-20)

or

$$V_{mg_meas} = V_{mg_ideal} - \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-21)

Therefore, (5-18) can be re-written using (5-21) as

$$V_{inv_meas} = \phi_B \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right] + V_{mg_ideal} - \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-22)

This can further be re-written using (5-19) and re-arranging terms to get

$$V_{inv_meas} = \phi_B \cdot \left[\frac{C_{OX} + C_D + C_{IT} + C_{OT}}{C_{OX}} \right]$$

$$+ \left\{ V_{inv_ideal} - \phi_B \cdot \left[\frac{C_{OX} + C_D}{C_{OX}} \right] \right\} - \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-23)

This reduces to

$$V_{inv_meas} - V_{inv_ideal} = \phi_B \cdot \left[\frac{C_{IT} + C_{OT}}{C_{OX}}\right] - \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-24)

 C_{IT} is often defined as $q \cdot D_{IT}$, where N_{IT} is the integral of D_{IT} over the relevant

portion of the bandgap (see Section 3.4). Therefore,

$$C_{IT} = q \cdot D_{IT} \cong \frac{q \cdot N_{IT}}{\phi_B}$$
(5-25)

Since the cause of the increase of the subthreshold swing, *S*, (equivalent to a stretch-out of the subthreshold portion of the I_D - V_{GS} characteristic) is the action of interface traps being filled, and in the case of SiC MOSFETs of oxide traps being filled as well when sweeping from negative gate bias to positive bias, we can similarly write the capacitance due to oxide traps charging during the gate sweep as

$$C_{OT} = \frac{q \cdot \Delta N_{OT}}{\phi_B} \tag{5-26}$$

Therefore, (5-24) can be re-written as

$$V_{inv_meas} - V_{inv_ideal} = \phi_B \cdot \left(\frac{\frac{q \cdot N_{IT}}{\phi_B} + \frac{q \cdot \Delta N_{OT}}{\phi_B}}{C_{OX}}\right) - \frac{q \cdot N_{OT}}{C_{OX}}$$
(5-27)

so that the difference between the ideal inversion voltage and the measured value is

$$V_{inv_ideal} - V_{inv_meas} = \frac{q \cdot N_{OT}}{C_{OX}} - q \cdot \left[\frac{N_{IT} + \Delta N_{OT}}{C_{OX}}\right]$$
(5-28)

In terms of changes in the subthreshold swing, the difference between the

measured and ideal values, (5-15) and (5-13) respectively, yields

$$S_{meas} - S_{ideal} = \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left[\frac{C_{IT} + C_{OT}}{C_{OX}}\right]$$
(5-29)

or

$$S_{meas} - S_{ideal} = \frac{k \cdot T}{q} \cdot \ln(10) \cdot \left(\frac{\frac{q \cdot N_{IT}}{\phi_B} + \frac{q \cdot \Delta N_{OT}}{\phi_B}}{C_{OX}}\right)$$

$$= \frac{k \cdot T}{q \cdot \phi_B} \cdot \ln(10) \cdot \left[\frac{q \cdot (N_{IT} + \Delta N_{OT})}{C_{OX}}\right]$$
(5-30)

The above results are summarized in the schematic shown in Figure 5-9.

If we define S_{meas_up} as the subthreshold swing obtained by sweeping the gate up from negative bias to positive bias following a negative-bias stress, and similarly define S_{meas_down} as the subthreshold swing obtained by sweeping the gate down from positive bias to negative bias following a positive-bias stress, then the difference in these two values yields

$$S_{meas_up} - S_{meas_down} = \frac{k \cdot T}{q \cdot \phi_B} \cdot \ln(10) \cdot \left[\frac{q \cdot (\Delta N_{OT_up} - \Delta N_{OT_down})}{C_{OX}}\right]$$
(5-31)

If we make the further simplifying assumption that there is not much change in the number of oxide traps in the case of S_{meas_down} where the bias is still positive in the subthreshold region following a positive-bias stress, then (5-31) simplifies to

$$S_{meas_up} - S_{meas_down} = \frac{k \cdot T}{q \cdot \phi_B} \cdot \ln(10) \cdot \left[\frac{q \cdot (\Delta N_{OT_up})}{C_{OX}}\right]$$
(5-32)

and re-arranging terms yields an expression for the amount of oxide charge changing charge state during the sweep

$$\Delta N_{OT_meas} = \frac{\phi_B}{k \cdot T} \cdot \frac{C_{OX}}{\ln(10)} \cdot (S_{meas_up} - S_{meas_down})$$
(5-33)

This is the expression given in Figure 5-4. Likewise, (5-30) will reduce to

$$S_{meas_down} - S_{ideal} = \frac{k \cdot T}{q \cdot \phi_B} \cdot \ln(10) \cdot \left[\frac{q \cdot N_{IT}}{C_{OX}} \right]$$
(5-34)

when the gate is swept down in voltage, providing a method for calculating the interface trapped charge between mid-gap and inversion

$$N_{IT} = \frac{\phi_B}{k \cdot T} \cdot \frac{C_{OX}}{\ln(10)} \cdot (S_{meas_down} - S_{ideal})$$
(5-35)

This is the equation that was used to calculate the number of interface traps in Figure 5-8.

The difficulty arises in calculating an accurate value for the subthreshold swing, since the inversion current and mid-gap currents vary from pA to twenty orders of magnitude smaller current at room temperature, and there are only a few decades at best of actual measured subthreshold current. The currents in Si MOS generally ranged from 10^{-7} to 10^{-14} A, allowing for much greater confidence in fitting the slope from measured data. Given the great amount of extrapolation required, it is wise to take the results with several grains of salt. However, in spite of this caveat, reasonable results can be obtained.

The difference in the mid-gap voltage measured following a positive-bias stress by a sweep down in gate voltage and the mid-gap voltage measured following a negativebias stress by a sweep up in gate voltage can be found using (5-20):

$$V_{mg_meas_down} - V_{mg_meas_up} = \left[V_{mg_ideal} - \frac{q \cdot N_{OT_down}}{C_{OX}} \right] - \left[V_{mg_ideal} - \frac{q \cdot N_{OT_up}}{C_{OX}} \right]$$

$$= \frac{q \cdot (N_{OT_up} - N_{OT_down})}{C_{OX}}$$
(5-36)

or

$$V_{mg_meas_down} - V_{mg_meas_up} = \frac{q \cdot \Delta N_{OT_stress}}{C_{OX}}$$
(5-37)

Re-arranging terms provides a method for calculating the change in the number of oxide traps occupied due to a bias stress

$$\Delta N_{OT_stress} = \frac{C_{OX}}{q} \cdot \left(V_{mg_meas_down} - V_{mg_meas_up} \right)$$
(5-38)

This is the equation that was used to extrapolate the number of oxide traps in Figure 5-5 and Figure 5-6.

In a similar manner, the difference in the inversion voltage measured following a positive-bias stress by a sweep down in gate voltage and the inversion voltage measured following a negative-bias stress by a sweep up in gate voltage can be found using (5-28):

$$V_{inv_meas_down} - V_{inv_meas_up} = \left\{ V_{inv_ideal} - \frac{q \cdot N_{OT_down}}{C_{OX}} + q \cdot \left[\frac{N_{IT} + \Delta N_{OT_down}}{C_{OX}} \right] \right\}$$

$$- \left\{ V_{inv_ideal} - \frac{q \cdot N_{OT_up}}{C_{OX}} + q \cdot \left[\frac{N_{IT} + \Delta N_{OT_up}}{C_{OX}} \right] \right\}$$
(5-39)

or

$$V_{inv_meas_down} - V_{inv_meas_up} = \frac{q \cdot (N_{OT_up} - N_{OT_down})}{C_{OX}}$$

$$-\frac{q \cdot (\Delta N_{OT_up} - \Delta N_{OT_down})}{C_{OX}}$$
(5-40)

that, using the simplifying expressions introduced in calculating (5-33) and (5-38) above, can be re-written as:

$$V_{inv_meas_down} - V_{inv_meas_up} = \frac{q \cdot [\Delta N_{OT_stress} - \Delta N_{OT_meas}]}{C_{OX}}$$
(5-41)

Equations (5-37) and (5-41) are a re-statement of (5-3) and (5-2), respectively, thus completing the derivation of the equations used to analyze the experimental subthreshold swing results of Section 5.3 and summarized in Figure 5-4.

5.5 Experimental observation of variation in subthreshold swing versus measurement speed

Not only does the speed of the measurement affect the magnitude of the V_T instability observed, as discussed above, it also affects the stretch-out of the subthreshold slope, which is the inverse of the subthreshold swing defined above. Figure 5-10 shows a series of back-and-forth instability measurements, with a 3,000 s gate-bias stress of ±15 V. The measurement times ranged from around 1 s to 1,200 s. Not surprisingly, the fastest measurement speeds resulted in the largest V_T instability and the slowest measurement speeds with the smallest instability. This narrowing of the envelope of the I_D - V_{GS} characteristics with slower measurement speeds is also depicted in Figure 5-11.



Figure 5-10: Variation in the subthreshold *I-V* characteristics of a 4H-SiC MOSFET from Sample Set I due to a back-and-forth gate-bias stressing sequence as a function of measurement speed. The width of the envelope decreases with increasing measurement speed, as does the stretch-out of the subthreshold *I-V* characteristics.



Figure 5-11: Variation in the threshold voltage, calculated using the inversion current method and the data shown in Figure 5-10, as a function of measurement time when sweeping up following a negative-bias stress and sweeping down following a positive-bias stress.

The number of switching oxide traps ranges from around 1.9×10^{11} traps per cm² for a 1-s sweep time to as little as 0.8×10^{11} traps per cm² for the slowest measurement time of 1 ks.

Accompanying this decrease in V_T instability is a change in the subthreshold swing during the sweep up following a negative gate-bias stress. Not only does the envelope of I_D - V_{GS} characteristics get closer together, but the slopes of these characteristics become more parallel as well. This is shown explicitly in Figure 5-12. The stretch-out decreases as the measurement takes longer to complete. This makes sense if most of the oxide traps change charge state before they can be measured. It is interesting to note that although the subthreshold swing does not change with measurement time when sweeping down following a positive-bias stress, some oxide traps have clearly changed charge state during the longer measurements since the I_D - V_{GS} characteristics have moved to the left. A similar translation occurs in the opposite direction during the longer measurements when sweeping up following a negative-bias stress. These ideas are discussed further in Section 6.7 regarding the modeling of this effect.



Figure 5-12: Variation in the subthreshold swing, *S*, calculated using the data shown in Figure 5-10, as a function of measurement time when sweeping up following a negative-bias stress and sweeping down following a positive-bias stress.

These results are consistent with the basic view that oxide traps can and do change charge state not only during the bias stress, but also during the measurement, and that faster measurements will result in larger observable shifts in V_T .

5.6 Charge separation analysis

As was discussed in previous chapters, there are potentially four different types of interfacial charge which can affect the threshold voltage: oxide traps, interface traps, mobile ions, and fixed charge. It is useful to be able to analytically separate out the different types of charge in order to fully model the device characteristics as well as to understand the full effects of processing variations. Since these different types of charge are likely due to different types of defects or impurities, it is likely that processing variations will affect them differently.

Charge separation analysis can be difficult, since there are a number of balancing or competing effects. For example, as discussed in Section 4.9, oxide trap filling or emptying can be balanced by mobile ion drift at elevated temperatures. Also, large numbers of negatively charged interface traps, as is the case for *n*-channel MOSFETs in inversion, may be balanced by large numbers of positively charged oxide traps and fixed charge. At very fast times, interface traps and oxide traps may both be changing charge state, thus making it difficult to determine whether a standard interface trap measurement is due to interface traps alone, or to a combination of trap types. In addition, if oxide trap filling and emptying requires a two-step process involving interface traps, as discussed in Chapter 6, then this charge separation analysis will be even more difficult.

126

The V_T -instability effect due to changes in oxide trap occupation can also affect other measurements designed to extract the interface-trap density. One such situation is for the dual-capacitor method proposed by Dr. Mrinal Das of Cree Research at an informal workshop at ARL several years ago. The basic idea is to compare the measured flat-band voltages for similarly processed *n*- and *p*-type capacitors. The flat-band voltage, V_{FB} , is the voltage at which a flat-band surface potential is achieved in an MOS capacitor. The flat-band capacitance, C_{FB} , can be defined in terms of C_{OX} and the depletion capacitance, C_D .

$$\frac{1}{C_{FB}} = \frac{1}{C_{OX}} + \frac{1}{C_D} \approx \frac{1}{\varepsilon_{OX}/t_{OX}} + \frac{1}{\varepsilon_s/L_D}$$
(5-42)

Values for C_{FB} as a ratio of C_{OX} are readily found as a function of oxide thickness and impurity doping [102].

When including the effects of the voltage shift due to oxide trap charging and discharging, the fuller analysis is as follows. Assuming that no mobile ionic charge is present, then the net charge is the sum of the N_{IT} , N_F , and N_{OT} , as discussed in Chapters 2 and 3, and will be the cause of the deviation of the measured flat-band capacitance compared with the ideal value, for both *n*-type and *p*-type capacitors:

$$N_{net_n} = \frac{C_{OX}}{q} \cdot (V_{FB_{ideal_n}} - V_{FB_{meas_n}}) = -N_{IT_n} + N_F + N_{OT_n}$$
(5-43)

$$N_{net_p} = \frac{C_{OX}}{q} \cdot (V_{FB_{ideal_p}} - V_{FB_{meas_p}}) = N_{IT_p} + N_F + N_{OT_p}$$
(5-44)

The ideal flat-band voltages are presented in Sze, p. 363 [102] for both *n*-type and *p*-type semiconductors:

$$V_{FB_ideal_n} = \phi_m - \left(\chi_{SiC} + \frac{E_{g_SiC}}{2 \cdot q} - \phi_{B_SiC}\right)$$
(5-45)

$$V_{FB_ideal_p} = \phi_m - \left(\chi_{SiC} + \frac{E_{g_SiC}}{2 \cdot q} + \phi_{B_SiC}\right)$$
(5-46)

Therefore, the difference in the ideal flat-band voltages is:

$$V_{FB_ideal_n} - V_{FB_ideal_p} = \phi_{B_SiC} - \left(-\phi_{B_SiC}\right) = 2 \cdot \phi_{B_SiC}$$
(5-47)

The difference in the net charge can be written in terms of measured and ideal flat-band voltages:

$$N_{net_p} - N_{net_n} = \frac{C_{OX}}{q} \cdot \left[(V_{FB_{ideal_p}} - V_{FB_{meas_p}}) - (V_{FB_{ideal_n}} - V_{FB_{meas_n}}) \right]$$
(5-48)

This can be re-written by grouping the measured and ideal terms:

$$N_{net_p} - N_{net_n} = \frac{C_{OX}}{q} \cdot \left[(V_{FB_meas_n} - V_{FB_meas_p}) - (V_{FB_ideal_n} - V_{FB_ideal_p}) \right]$$

$$= \frac{C_{OX}}{q} \cdot \left[(V_{FB_meas_n} - V_{FB_meas_p}) - 2 \cdot \phi_{B_sic} \right]$$
(5-49)

This difference in net charge can also be written in terms of the charge-trap densities:

$$N_{net_p} - N_{net_n} = (N_{IT_p} + N_F + N_{OT_p}) - (-N_{IT_n} + N_F + N_{OT_n})$$

$$= (N_{IT_p} + N_{IT_n}) + (N_{OT_p} - N_{OT_n})$$
(5-50)

Therefore, the difference in the measured flat-band capacitances of similarly processed *n*-type and *p*-type capacitors results in the following relation:

$$V_{FB_meas_n} - V_{FB_meas_p} = \frac{q}{C_{OX}} \cdot \left[(N_{IT_p} + N_{IT_n}) + (N_{OT_p} - N_{OT_n}) \right] + 2 \cdot \phi_{B_SiC} \quad (5-51)$$

So, if not for the presence of switching oxide traps, the number of interface traps could be readily determined. However, since near-interfacial oxide traps are also present, this

method yields the total number of interface traps along with the number of oxide traps that change charge state. This last term is, as demonstrated by the results of this chapter, highly time dependent. The faster the measurements, the larger the lower bound for the number of oxide traps present.

The results of Section 5.3 show how a lower bound can be calculated for the number of switching oxide traps. However, it is still difficult to determine the total number of oxide traps present. If the total number of interface traps can be independently calculated, then (5-51) would provide one method. *C-V* analysis on *p*-type capacitors is a standard method for determining the interface trap density [115]. Several methods also exist using the actual MOSFET. These include inferring the interface trap charge from the change in net charge with measurement temperature [118], and charge pumping [108].

Another question is what is the total number of fixed charge? The following analysis provides some insight. Using *n*-channel MOSFETs and *p*-type capacitors, which are readily available, a similar analysis to (5-43) and (5-44) can be performed:

$$V_{T_{ideal}} - V_{T_{meas}} \cong \frac{q}{C_{OX}} \cdot N_{net_{n}} = \frac{q}{C_{OX}} \cdot \left(-N_{IT_{n}} + N_{F} + N_{OT_{n}} \right)$$
(5-52)

$$V_{FB_ideal} - V_{FB_meas} \cong \frac{q}{C_{OX}} \cdot N_{net_p} = \frac{q}{C_{OX}} \cdot \left(N_{IT_p} + N_F + N_{OT_p}\right)$$
(5-53)

These equations can be rearranged to calculate the sum of fixed charge and oxide trap charge, if the interface trap charge can be calculated independently, as discussed above:

$$N_F + N_{OT_n} = N_{net_n} + N_{IT_n}$$
(5-54)

$$N_F + N_{OT_p} = N_{net_p} - N_{IT_p}$$
(5-55)

Recent results by Habersat and Lelis, et al. [105] using the sample sets described in Section 4.2, found that $N_F + N_{OT}$ is a large sum under negative bias (13-20×10¹¹ cm⁻³), but a relatively small sum under positive bias (2-5×10¹¹ cm⁻³), corresponding to (5-55) and (5-54), respectively. These results by themselves are ambiguous since they could be due either to a large number of fixed charge, which was found to be positively charged in Si MOSFETs [31], balanced by a large number of negatively charged oxide traps under positive bias which are uncharged under negative bias; or to a small number of fixed charge with a large number of positively charged oxide traps under negative bias and uncharged under positive bias.

Recent ESR results discussed in Chapter 2 suggest that E'-type defects are present in the oxide of SiC MOSFETs. No other electrically active defects in the oxide have been identified to date. E' defects were identified as the dominant oxide trap in irradiated Si MOSFETs with similar gate oxide thicknesses, as well as the cause of the V_T instability in those devices. Given that those E' defects acted as hole traps, it is not unreasonable that they behave in a similar fashion in SiC MOSFETs as well. If so, then the charge separation results of Habersat, Lelis, et al. suggest that there is not a large number of fixed charge. However, it has also been suggested that the E' center may under certain circumstances be able to trap a second electron, thus becoming negatively charged [100]. If this is the case, then the $N_F + N_{OT}$ results discussed above may yet be ambiguous regarding the total number of fixed charge.
5.7 Summary

Key experimental results presented in this chapter are consistent with a direct tunneling process, wherein electrons both fill and empty near-interfacial oxide traps distributed spatially over the first 40 to 50 Å of the oxide. These results include a linear increase in the V_T instability with the log of the stress time, the observation that extrapolating to zero instability occurs at stress times comparable to the first tunneling transition times of around 0.1 ps, and—most importantly—the strong time-dependence of the measurements. The fast I-V system at NIST allowed measurements as fast as 20 μ s to be performed, with the fastest measurements revealing the largest V_T instabilities. The number of switching oxide traps with the fastest measurements was as large as 3.6×10^{11} cm^{-2} . It is not surprising that the speed of the measurement would be so important since a gate bias is necessarily applied during the measurement which in fact applies a bias stress of its own. The slower the measurement, the longer the time for this alternate gate bias to affect the results of the previous bias stress. The standard 1-s measurements could also be slowed down, and measurements as slow as 1 ks were performed, with the slowest measurements resulting in the smallest V_T instabilities and oxide traps densities half as large as those at regular 1-s ramp speeds.

Clearly, the near-interfacial oxide traps can change charge state during the measurement. Further evidence of this is the larger subthreshold swing observed when sweeping up in gate bias following a negative-bias stress compared to that when sweeping down in gate bias following a positive-bias stress. Although some portion of the deviation in the ideal subthreshold swing is certainly due to interface traps changing charge state, the repeated difference in subthreshold swing between sweeping up and

131

sweeping down is likely due to the complex response of oxide traps, especially since slower measurements showed less of a difference. A detailed analysis of this effect was presented. It was also shown that calculating the instability of the subthreshold I_D - V_{GS} characteristics by extrapolating down to mid-gap current levels resulted in V_T instabilities comparable to those found using the fast *I*-*V* system. In some cases, extrapolated instability calculations have indicated switching oxide-trap densities as large as 5×10^{11} cm⁻². These are trap densities comparable to those of interface traps.

Given the strong time-dependence of the response of the oxide traps, it is very difficult to apply analytical techniques in an attempt to separate out the effects of oxide traps and interface traps. It is also clear that not all of the non-interface trap charge is "fixed." Finally, it is important to distinguish between these different types of interfacial charge traps since they are probably caused by different physical defects, and therefore different processing techniques will likely affect them differently.

6 Tunneling Modeling

6.1 Introduction

Given the likelihood of a tunneling mechanism being responsible for the experimental results presented in this work—as discussed in previous chapters, this chapter will discuss the derivation, application, and comparison of a two-way tunneling model with those experimental results. It is important that a two-way model is applied, since depending on the electric field in the oxide and the distribution of charged and uncharged traps at a given spatial depth, electrons are likely to be both tunneling in and out (an electron tunneling out of an oxide trap back to the semiconductor can be equivalently considered to be a hole tunneling into the trap) simultaneously during a given period of time. This model is applied to both the bias stress periods, as well as during the measurement periods, when the gate-to-source voltage, V_{GS} , is being ramped.

The present model can successfully account for the variation in threshold-voltage instability observed as a function of bias-stress time, bias-stress magnitude, and measurement time. However, further improvements are still needed since it does not yet account for the variation in subthreshold swing. As will be discussed, this improvement will likely require the incorporation of a two-step process, with tunneling between near-interfacial oxide traps and interface traps, rather than simply between the oxide traps and the SiC valence and/or conduction bands.

133

6.2 Background

The modeling of the quantum-mechanical tunneling process of electrons filling and emptying trap states in the insulator will be done using the semi-classical WKB Approximation to solve the Schrodinger Equation. This approximation is named after three physicists, G. Wentzel, H.A. Kramers, and L. Brillouin, who in 1926 independently applied the already known simplification of wave functions to the wave mechanics of Schrodinger's quantum physics. This must be solved self-consistently with Poisson's equation at the interface and into the near-oxide region. The calculations discussed in Chapter 3 present a straight forward way of calculating the potential and field in the oxide.

This study follows the previous work of McLean [113], who applied the wellknown WKB model for the direct tunneling of electrons into localized trap states in the insulating gate oxide, but expands upon it by allowing for the simultaneous tunneling of electrons in and out of the near-interfacial oxide traps.

The tunneling transition rate, which is the rate of change of the neutralized, uncharged trap density with respect to time, can be defined as the product of the probability of a tunneling transition and the charged trap density, for a given trap energy level and distance from the interface:

$$T(E, x, t) = \frac{\partial n_{OT_neut}(E, x, t)}{\partial t}$$

$$= g(E, x, t) \cdot [n_{OT_total}(E, x) - n_{OT_neut}(E, x, t)]$$
(6-1)

where $n_{OT_total}(E,x)$ is the total number of oxide traps for a given trap energy level and distance from the interface, $n_{OT_neut}(E,x,t)$ is the number of uncharged states where an

electron has neutralized the trapped hole, and g(E, x, t), which is the probability of a tunneling event based on the WKB approximation, is given by

$$g(E, x, t) = \alpha(E, x, t) \cdot \exp[-Z(E, x, t)]$$
(6-2)

where the pre-factor $\alpha(E,x,t)$ is the "supply function" of the tunneling electrons and the exponential term controls the tunneling depth, based on the scaled distance Z(E,x,t):

$$Z(E, x, t) = 2 \cdot \int_0^x \beta(E, x', t) \cdot dx'$$
(6-3)

with $\beta(E,x,t)$ related to the barrier height facing the tunneling electrons:

$$\beta(E, x, t) = \sqrt{\left(\frac{2 \cdot m_t^*}{\hbar^2}\right) \cdot \left[V_B(x, t) - E\right]}$$
(6-4)

where m_t^* is the effective tunneling mass, *h*-bar is Planck's constant divided by $2 \cdot \pi$, and $V_B(x,t)$ -*E* the potential barrier itself [113].

If the oxide traps are assumed to all exist at a single energy level, E_t , and a trapezoidal approximation is used for the effect of an applied bias [112, 119], then for small variations in time [113]:

$$\beta(x) = \sqrt{\left(\frac{2 \cdot m_t^*}{\hbar^2}\right) \cdot \left[E_t - q \cdot \boldsymbol{\mathcal{E}}_{OX} \cdot x\right]}$$
(6-5)

Since α varies much more slowly with *E* and *x* compared to the exponential term [113], the tunneling probability function can be re-written as

$$g(x) = \alpha \cdot \exp[-2 \cdot \beta(x) \cdot x] \tag{6-6}$$

Equation (6-1) can be solved for the neutralized trap concentration:

$$n_{OT_neut}(x,t) = n_{OT_total}(x) \cdot \left[1 - \exp(-\alpha \cdot t \cdot \exp[-2 \cdot \beta(x) \cdot x])\right]$$
(6-7)

Therefore, using (6-1), (6-6), and (6-7), the tunneling transition rate is given by

$$T(x,t) = n_{OT_total}(x) \cdot \alpha \cdot \exp[-2 \cdot \beta(x) \cdot x] \cdot \left[\exp(-\alpha \cdot t \cdot \exp[-2 \cdot \beta(x) \cdot x])\right]$$
(6-8)

The tunneling probability, g(x), tunneling transition rate, T(x), and a normalized distribution of charged traps, $[n_{OT_total}(x) - n_{OT_neut}(x)]$, are shown in Figure 6-1 for a given time, *t*. There is a clear peak in the tunneling transition rate, such that the tunneling is localized at a given depth into the oxide for a particular time.



Figure 6-1: The sharp exponential decay in the tunneling probability with distance into the oxide, coupled with the relatively sharp distribution of oxide traps heretofore unaffected by the tunneling and therefore yet to change charge state, results in a sharp peak in the tunneling transition rate, illustrating the concept of a "tunneling front" which moves deeper into the oxide at a linear-with-log-time rate.

If the distribution of traps is assumed to be uniform, then differentiating the tunneling transition rate with respect to distance and setting it equal to zero yields the distance at which the peak tunneling is occurring for a given time *t*:

$$x_{peak} = \frac{1}{2 \cdot \beta} \cdot \ln(\alpha \cdot t) \tag{6-9}$$

Equation (6-9) indicates that the location of this peak moves logarithmically with time into the oxide, which leads to the notion of a tunneling front [112, 113], where most trap states in its wake have changed charge state whereas those beyond remain unchanged. This linear-with-log-time response is what we generally observe for the threshold-voltage instability at room temperature as demonstrated in the experimental chapters of this study. In addition, as noted earlier, this tunneling front was found to move at a rate of about 2 Å per decade of time in irradiated Si MOS [112]. Furthermore, given that the initial tunneling transitions were calculated to occur at around 0.1 ps [120], this means at bias stress times typically applied, this tunneling front is somewhere between 30 and 40 Å into the oxide from the SiC interface. Typical values of β range between 0.55 and 0.75 Å⁻¹ so that a typical full-width half-max value of the tunneling transition rate is around 2 Å wide (see Figure 6-1). Clearly, this is a spatially localized phenomenon when considering a one-way tunneling mechanism to oxide traps with a relatively uniform distribution and a single energy level.

Since the tunneling front should be at the interface when the first tunneling transitions occur, the time scale can be set by letting

$$\alpha = \frac{1}{t_0} \tag{6-10}$$

where t_0 is the initial tunneling transition time described above [120]. For every decade of time, the tunneling front moves at a rate of $ln(10)/(2 \cdot \beta)$, which yields a tunneling rate of 1.5 to 2 Å per decade of time.

6.3 Two-way tunneling model

For a one-way tunneling process, with a constant bias applied, it may be enough to simply calculate whether the tunneling front has passed or not. However, since the experimental results presented in this study involve switching the bias polarity back and forth, as well as varying the measurement ramp speed, it is important to readily account for constantly varying bias conditions and the possibility that electrons that have previously tunneled in might under a different set of gate-bias conditions now tunnel out (see Figure 6-2 for a schematic of this process). A two-way tunneling model allows for this possibility.



oxide traps are positively charged

oxide traps are neutralized

Figure 6-2: Schematic of holes predominantly tunneling in under negative bias to uncover positive charge and electrons predominantly tunneling in under positive bias to neutralize the positive charge.

For such a case, the new distribution of neutralized states can be found from the present distribution as follows, using a similar procedure to that for calculating the time-dependent occupation of interface traps [104]:

$$n_{OT_neut}(x,t+\Delta t) = n_{OT_neut}(x,t) + \Delta n_{OT_neut}(x,t)$$
(6-11)

where the change in the neutralized trap distribution is found using (6-1) by taking the difference between the electrons tunneling in and those tunneling out (equivalent to holes tunneling in)

$$\Delta n_{OT_neut}(x,t) = \Delta t \cdot p_{tun_e}(x) \cdot \left[n_{OT_total}(x) - n_{OT_neut}(x,t) \right]$$

$$-\Delta t \cdot p_{tun_h}(x) \cdot n_{OT_neut}(x,t)$$
(6-12)

with the probability of electrons tunneling in given by

$$p_{tun_e}(x) = \alpha_e \cdot \exp[-2 \cdot \beta_e(x) \cdot x]$$
(6-13)

using (6-6), and the probability of holes tunneling in given by

$$p_{tun_h}(x) = \alpha_h \cdot \exp[-2 \cdot \beta_h(x) \cdot x]$$
(6-14)

The values for β are as given in (6-5):

$$\beta_{e}(x) = \sqrt{\left(\frac{2 \cdot m_{t_{e}}^{*}}{\hbar^{2}}\right) \cdot \left[E_{t_{e}} - q \cdot \boldsymbol{\mathcal{E}}_{OX} \cdot x\right]}$$
(6-15)

but with separate values for the effective tunneling mass of electrons and holes ($m_{t_e}^*$ and $m_{t_h}^*$, respectively), and for the trap level for electrons and holes, and the opposite sign for the field-modified barrier height for holes

$$\beta_{h}(x) = \sqrt{\left(\frac{2 \cdot m_{t_{-h}}^{*}}{\hbar^{2}}\right) \cdot \left[E_{t_{-h}} + q \cdot \boldsymbol{\mathcal{E}}_{OX} \cdot x\right]}$$
(6-16)

The effective electron tunneling mass is well know and is given by $0.42 \cdot m_0$ [112, 113, 120, 121], where m_0 is the electron rest mass. The effective tunneling mass for holes varies greatly in the literature [122] and appears to be generally used as a fitting parameter.

From (6-15) and (6-16) it is clear that the deeper in energy level the trap is (larger E_t), the less effect variations in the potential barrier due to the oxide field or trap depth will have. Therefore, the experimental results when varying the stress bias (and therefore the oxide field) presented in Chapter 4 can provide insight into the trap energy level. Under positive bias, where electrons tunneling in will dominate, a larger variation in V_T instability with bias-stress magnitude (as seen in Figure 4-11) implies a shallower energy trap level for electrons. Similarly under negative bias, where holes tunneling in will dominate, a smaller variation in V_T instability with bias-stress magnitude (also seen in Figure 4-11) implies a deeper energy trap level for holes. These results are consistent with a trap level slightly above the conduction band of SiC, which is one of the trap levels proposed for irradiated Si MOS [86, 112, 89, 90]. This trap level is illustrated in Figure 6-3.



tunnel in (electrons tunnel out)



A trap level across from well below the valence band edge of Si, which would be around the valence band edge of SiC, is another energy level proposed for irradiated Si MOS [120, 112, 89, 90]. This deeper energy level for electrons tunneling in (and shallower trap level for holes tunneling in) is consistent with the larger variation in V_T instability with negative bias-stress magnitude and smaller variation with positive biasstress magnitude for earlier results on deposited oxide samples (as seen in Figure 4-10). It is also consistent with the recent results on SiC MOS capacitor structures discussed in Section 4.8 [109]. These capacitor instability results may also suggest that both of these trap levels are present.

6.4 Steady-state conditions

As the charge tunnels in and out, it will modify the total amount of oxide charge versus distance from the interface, and therefore the spatial variation in oxide field—as described in Chapter 3. At some point in time, these two processes will achieve a steady state condition between electrons tunneling in and out. This steady state can be found by setting $\Delta n_{OT_neut}(x)$ in (6-12) to zero, such that:

$$p_{tun_h}(x) \cdot n_{OT_neut}(x) = p_{tun_e}(x) \cdot \left[n_{OT_total}(x) - n_{OT_neut}(x) \right]$$
(6-17)

or

$$\alpha_{h} \cdot \exp[-2 \cdot \beta_{h}(x) \cdot x] \cdot n_{OT_neut}(x) = \alpha_{e} \cdot \exp[-2 \cdot \beta_{e}(x) \cdot x]$$

$$\cdot \left[n_{OT_total}(x) - n_{OT_neut}(x)\right]$$
(6-18)

which reduces to

$$\frac{n_{OT_neut}(x)}{n_{OT_total}(x)} = \frac{1}{1 + \begin{pmatrix} \alpha_h \\ \alpha_e \end{pmatrix} \cdot \exp[-2 \cdot (\beta_h(x) - \beta_e(x)) \cdot x]}$$
(6-19)

If $\beta_h(x) = \beta_e(x)$, then the fraction of neutralized oxide traps is given by $1/[1+(\alpha_h/\alpha_e)]$. Letting $\alpha_h = \alpha_e$, then the fraction of neutralized oxide traps at a particular oxide depth is less than 0.5 when $\beta_h(x) < \beta_e(x)$, and greater than 0.5 when $\beta_h(x) > \beta_e(x)$. Therefore, an oxide trap is more likely to be neutralized when

$$m_{t_{-h}}^* \cdot \left[E_{t_{-h}} + q \cdot \boldsymbol{\mathcal{E}}_{OX} \cdot x \right] > m_{t_{-e}}^* \cdot \left[E_{t_{-e}} - q \cdot \boldsymbol{\mathcal{E}}_{OX} \cdot x \right]$$
(6-20)

This effect is demonstrated in the next set of figures, which plot the fraction of neutralized oxide traps, i.e., those for which an electron has tunneled in to neutralize the trapped positive charge (see Chapter 2 for a discussion of the E' defect and its charge states), versus distance into the oxide. This variation is given for constant oxide fields

ranging from -2 to +2 MV/cm, which covers the general range of the experimental results. Although very useful in understanding the relationship between the trap occupation density as a function of oxide trap depth and field for given values of energy trap level, etc., it does not show an actual distribution of neutralized traps versus oxide depth since the oxide field varies with oxide depth, especially when most of the positively charged oxide traps have not been neutralized (see Chapter 3 and Figure 6-12).

Figure 6-4 plots the variation in the fraction of neutralized oxide traps for a trap energy level adjacent to the semiconductor conduction band (see Figure 6-3). Similarly, Figure 6-5 plots this variation for a trap energy level well below the valence band edge of Si and approximately adjacent to the SiC valence band edge. These calculations demonstrate how trap levels either close to the conduction band or close to the valence band can lead to greater sensitivity to either a positive or negative oxide field, if an appropriate value for the effective hole tunneling mass is chosen.



Figure 6-4: Calculation of the steady-state fraction of neutralized (filled with electrons) oxide traps versus oxide depth as a function of constant oxide-field values for the following parameter values: $E_{t_e} = 2.7 \text{ eV}, E_{t_h} = 6.3 \text{ eV}, \text{ and } m_h^* = 0.162 \ (m_e^* = 0.42 \text{ and } \alpha_e = \alpha_h).$



Figure 6-5: Calculation of the steady-state fraction of neutralized (filled with electrons) oxide traps versus oxide depth as a function of constant oxide-field values for the following parameter values: $E_{t_e} = 6.0 \text{ eV}, E_{t_h} = 3.0 \text{ eV}$, and $m^*_h = 0.885$ ($m^*_e = 0.42$ and $\alpha_e = \alpha_h$).

There are some difficulties with this approach. The opposite sensitivity to either a positive or negative oxide field can be achieved for the same trap energy levels by changing the effective hole tunneling mass value. Also, for a trap level across from the conduction band, although the number of available holes is always large, the number of available of electrons is dependent on the surface potential. The opposite is true for a trap level across from the valence band. However, it is quite likely that the energy level will change depending on whether the oxide trap is positively charged or neutralized. For example, both Nicklaw, et al. and Goes, et al. showed using density functional theory (DFT) that the trap level could change by over an eV depending on the charge state due

to changes in relaxation or "puckering" of the defect structure [123, ¹²⁴]. If it is below the valence band when positively charged and above the conduction band when neutralized (although this change in trap energy is much less likely for a wide bandgap semiconductor such as SiC), then there would be no surface potential dependence for either tunneling transition. Figure 6-6 illustrates the variation in neutralized traps versus oxide depth and field for such a case, again using an appropriate value for the effective hole tunneling mass. (These are the values that were used to obtain reasonable fits to experimental results for variations in measurement time and gate bias stress—to be discussed later in this chapter.)

A more likely scenario is that the oxide trap levels lie within the bandgap for at least one of the trap states, and that this then involves a dependence not only on the surface potential, but also on the responsiveness of the interface traps. This possibility is beyond the scope of this work, but will be considered again when comparing the experimentally observed variation in subthreshold swing with the model.



Figure 6-6: Calculation of the steady-state fraction of neutralized (filled with electrons) oxide traps versus oxide depth as a function of constant oxide-field values for the following parameter values: $E_{t_e} = 6.0 \text{ eV}$, $E_{t_h} = 6.3 \text{ eV}$, and $m^*_h = 0.377 (m^*_e = 0.42 \text{ and } \alpha_e = \alpha_h)$.

6.5 **Tunneling time-step increments**

As discussed previously, the first tunneling transition rates have been calculated to occur at around 0.1 ps [120]. This then also provides a scale for the time increments used in (6-12). Figure 6-7 shows how the longer the tunneling occurs, the deeper into the oxide the tunneling front has moved. The net change in the charge state of those oxide traps closest to the interface is continuously decreasing. At the same time, the deeper into the oxide, the smaller the net change for a short, fixed time increment such as $t_0 = 1$ ps. This illustrates the need to take larger and larger time steps for longer tunneling times, since the net change per unit time keeps decreasing.



Figure 6-7: Calculation of the variation in net charge tunneling into oxide traps during a time increment of t_0 seconds as a function of total tunneling time.

There is, however, a difficulty with taking larger time steps. Although the occupancy of oxide traps closest to the interface quickly reaches a steady-state, as indicated in Figure 6-7, taking larger and larger time increments for longer and longer tunneling times necessitates taking the difference of very large numbers to calculate the net tunneling charge close to the interface —see (6-12). This difficulty can be avoided by setting the trap occupancy to the long-term steady state values (e.g., as shown in Figure 6-6) once this state has been achieved by the movement of the tunneling front with time.

6.6 Tunneling model mechanics

Figure 6-8 illustrates the basic algorithm for the tunneling model. First, the total number of active oxide traps is defined, along with their spatial distribution (see Figure 3-14 and accompanying discussion). An assumed interface trap distribution is also defined, as well as a schedule of different gate stress biases to apply (including the gate bias ramp applied during the measurement) and for how long. The initial bias stress is set for $V_{GS} = 0$ V, and is applied for 10^{15} s to ensure that the device begins with an unbiased initial condition.



Figure 6-8: Schematic of tunneling model algorithm.

With the application of a new gate bias, the surface potential is found selfconsistently with the interface trap and oxide trap occupations. Next, the details of the initial electric potential and electric field are calculated (see Chapter 3). This is followed by the tunneling-calculation loop. With every time step, the change in oxide trap occupancy is calculated. As the tunneling front proceeds deeper into the oxide, the trap states in its wake are set to the steady-state value for the reasons described above. The surface potential, electric field, etc., are recalculated several times per decade of time, and the time step is incremented as well. This is repeated for all the gate-bias stress steps, including those simulating a measurement.

Figure 6-9 depicts the variation in the positively charged oxide-trap distribution versus oxide depth, first for an initial very long-term unbiased steady-state condition, followed by a gate-bias sweep in the positive direction, and finally for a bias stress of 10^6 s and a bias stress of +15 V. The effect of gate-bias stress during the measurement is shown in 5-V steps, for a 1-s gate ramp.

Initially, under the unbiased condition, most of the oxide traps are positively charged (this is consistent with the neutralized trap distribution shown in Figure 6-6). The application of -5 V at the beginning of the gate-ramp for an initial measurement has the effect of slightly increasing the number of positively charged oxide traps over the first 25 Å of oxide. As the gate bias is swept up, the oxide traps over this range become more and more neutralized as electrons tunnel in. However, the traps deeper in are still unaffected. They are only affected during the long-term positive-bias stress, after which most of the oxide traps to a depth of around 40 Å have been neutralized.

150



Figure 6-9: Initial variations in the normalized positively charged oxide-trap distribution vs oxide depth as a function of gate bias during both a 1-s measurement sweeping up in gate bias and a subsequent $+15 \text{ V } 10^6$ -s bias stress.

Figure 6-10 shows the opposite process. Following the long-term positive-bias stress, the gate-bias is swept down, causing oxide traps over the first 25 Å to gradually become positively charged again as holes tunnel in (i.e., electrons tunnel out). Again, it is not until a long-term negative-bias stress that most of the oxide traps out to about 40 Å are positively charged. Figure 6-11 shows this bias stress and measurement sequence in the opposite order. This back-and-forth instability is essentially repeatable, as depicted in Figure 4-7. These calculations of the model show the importance of the measurement in affecting the charge state of the near-interfacial oxide traps. A faster measurement would not affect oxide traps as far from the interface.



Figure 6-10: Subsequent variations in the normalized positively charged oxide-trap distribution vs oxide depth as a function of gate bias following (1) an initial +15 V bias stress, (2) a 1-s measurements sweeping down in gate bias, and (3) a final -15 V 10^6 -s bias stress.



Figure 6-11: Further subsequent variations in the normalized positively charged oxide-trap distribution vs oxide depth as a function of gate bias following (1) an initial -15 V bias stress, (2) a 1-s measurements sweeping up in gate bias, and (3) a final +15 V 10^{6} -s bias stress.

Figure 6-12 depicts the change in the electric field versus oxide depth for the variations in the positively charged oxide-trap distribution given in Figure 6-11, when the gate bias is swept up during the measurement. As the amount of positively charged oxide traps decreases with electrons tunneling in with the gate bias increasing, the change in the oxide field diminishes. Following a long-term negative-bias stress wherein most of the oxide traps even deep in have been neutralized, the change in oxide field with oxide depth is minimal.



Figure 6-12: Variations in the change in oxide field vs oxide depth due to the variations in positively charge oxide-trap distribution as shown in Figure 6-11.

6.7 Comparison of model with experimental results

Figure 6-13 shows the simulated back-and-forth bias stressing and measurement results of the model, plotting the variation in the V_{OT} component of threshold voltage, defined in Chapter 3 as the voltage shift due to the presence of oxide trapped charge, versus each bias-stress sequence. The first two points show the before-and-after effects of a -15 V gate-bias stress. V_{OT} becomes more negative due to the increase in the number of positively charged oxide traps—see Figure 6-11 and (3-33). This is followed by a sweep up in gate bias, from -5 V to + 15V, in step increments of 0.1 V.



Individual Bias-Stress Sequences

Figure 6-13: Variation in V_{OT} , the voltage shift due to the presence of oxide trapped charge, is plotted versus individual bias stress sequences simulating both the long-term bias stresses, as well as the shorter bias stresses that comprise the measurement ramp for several different measurement times.

The change in V_{OT} following each simulated bias increment of the gate ramp is shown for several different measurement speeds. Since each bias stress step takes longer for the longer measurement speeds, it is not surprising that the simulated 1-s sweep leads to a larger change in V_{OT} during the measurement than does the simulated 20-µs sweep, since more electrons can tunnel in and increase the number of charge neutralized states. This is followed by a +15 V gate-bias stress and 0.1 V decrements in the gate bias to simulate the downward sweep in gate bias for the next measurement. Again, the longer sweep time leads to a larger shift during the simulated measurement. Since the drain current in the subthreshold region is given as [117]:

$$I_{D} = \mu_{n} \cdot \left(\frac{Z}{L}\right) \cdot \frac{\varepsilon_{s}}{\sqrt{2} \cdot L_{D} \cdot \beta^{2}} \cdot \left(\frac{n_{i}}{N_{a}}\right)^{2} \cdot \left(1 - e^{-\beta \cdot V_{Ds}}\right) \cdot e^{\beta \cdot \psi_{s}} \cdot \frac{1}{\sqrt{\beta \cdot \psi_{s}}}$$
(6-21)

Therefore, the log of the drain current is

$$\log(I_D) = \log[A] + \frac{\beta \cdot \psi_s}{\ln(10)} - \frac{\log(\beta \cdot \psi_s)}{2}$$
(6-22)

so that

$$\log(I_D) \propto \psi_s \tag{6-23}$$

This means that a plot of the surface potential versus V_{GS} will look similar in form to that of the log of the drain current versus gate bias. This is shown in Figure 6-14, where the results from Figure 6-13 are re-plotted, matching experimental results from Figure 5-2. The simulated increase in instability with increasing measurement speed is as expected, as is the larger difference under positive bias.



Figure 6-14: Simulated plots of surface potential (in terms of the bulk potential) versus gate bias are similar in form to plots of $log(I_D)$ vs V_{GS} . Model calculations predict larger instabilities for faster measurement ramp speeds, matching experimental.

The normalized variation of instability with ramp speed is explicitly plotted in Figure 6-15 for a bias stress of ± 15 V and a bias stress time of 10^6 s. This compares reasonably well with the experimental results of Figure 5-2. For this fit, the following model parameters were used: $E_{t_e} = 3.1$ eV, $E_{t_h} = 3.5$ eV, $m^*_h = 0.34$, $m^*_e = 0.42$, $\lambda =$ 0.10, and $\alpha_e = \alpha_h$, which are the same as those used in Figure 6-6. The value for the oxide trap distribution decay parameter, $\lambda = 0.10$ —described in Figure 3-14, and Equations (3-84) and (3-85)—produces a sharper exponential decay than is likely, but was used to better fit the strong upturn in the experimental data for the fastest measurement times. However, this is based on a linear extrapolation of the experimental data to slightly longer stress times, which is subject to a degree of uncertainty. Using a value of $\lambda = 0.03$ instead, which defines a milder exponential decay of the total oxide trap distribution, leads to a factor of 2 increase in ΔV_{OT} between 1-s and 20-µs simulated measurements instead of the factor of 3 found experimentally. However, the 1-s experimental measurement led to a calculation of only 1.2×10^{11} cm⁻² oxide traps changing charge state, which is on the low side of the observed range of results. This may be due to the presence of mobile ions in these slower, room-temperature measurements, given that evidence of mobile ions was observed at elevated temperature for devices from this same sample set (see Figure 4-22 and Figure 4-24), and evidence of room-temperature mobile-ion drift has been observed in other samples. Therefore, the more uniform distribution of oxide traps might actually produce a better fit, if a time-dependent mobile-ion drift component were added to the model.



Figure 6-15: Plot of the normalized back-and-forth threshold-voltage instability versus measurement ramp speed, comparing model calculations and experimental results.

The model was also used to simulate the variation in instability for different bias stress magnitudes. Figure 6-16 shows these results, as a function of stress time for simulated 1-s measurements, with significant differences due to the magnitude of the positive-bias stress, but no real difference seen under negative-bias stress. These results are quite consistent with the results of Figure 4-11. It should be noted that the total instability is dependent on the total number of active oxide traps assumed, which in this case was 1×10^{12} cm⁻². The same model parameters were used as in the above results, and the oxide trap distribution was set at λ =0.03. This is the more mild exponential decay and gives a fairly linear-with-log-stress-time response.



Figure 6-16: Plot of the changes in V_{OT} , the voltage shift due to the presence of oxide trapped charge, versus both positive and negative-bias stress magnitude with a simulated measurement speed of 1 s.

One thing that stands out about the comparison of experimental results with those of the model is that the model does not predict any variation in subthreshold swing between sweeping up following a negative-bias stress and sweeping down following a positive-bias stress—see Figure 6-14. This is contrary to the experimental results presented in Chapter 5 (other than those with very long measurement times).

This larger stretch-out of the subthreshold *I-V* characteristics is assumed to be due to the discharging of some oxide trap states during the measurement that were previously charged during the negative-bias stress. Since the bias applied during the measurement of the subthreshold region is of opposite polarity in this case—instead of being of the same bias polarity for a sweep down following a positive-bias stress—the number of oxide traps changing charge state is likely greater, thus causing a larger stretch-out in this case.

Figure 6-17 presents several possible measurement scenarios. If the measurement is made fast enough, then very few oxide traps will have had time to change charge state and there will be no additional stretch-out. On the other hand, if the measurement is slow enough, then most of the oxide traps will already have changed charge state by the time the subthreshold current rises above the experimental noise floor, representing about the last 20 percent of the change in surface potential from mid-gap to inversion. The fact that we do in fact observe experimentally a stretch-out suggests that the middle case is what actually happens. The problem is that the model does not predict this.

160



Figure 6-17: Schematic indicating several possible responses of the drain current versus gate voltage in the subthreshold region, depending on the speed of the measurement and the response time of the oxide traps.

What causes this significant change in the charge states over this range of surface potential? Figure 3-2 shows the variation in oxide field versus surface potential. The polarity of the field changes down at flat band. The magnitude of the field remains relatively small until the onset of strong inversion. The inversion carrier concentration is also quite insignificant until just below strong inversion. The only parameter that is changing significantly from mid-gap to inversion is the occupation of interface traps, although for wide band gaps such as SiC, the trap time constant for states near mid-gap is quite large [20, 102]. Another point to consider is the difference in the band gaps of Si and SiC, as shown in Figure 6-18.



Figure 6-18: Schematic showing a comparison of the band gaps of Si and 4H-SiC and how oxide traps that fall below the valence band edge of Si may actually be within the bandgap of SiC.

Even oxide trap states that lie well below the Si valence band edge may very well lie within the bandgap of SiC. This means that in order to tunnel to or from such states, it will require some sort of two-step process. The most likely scenario for such a two-step process is, for example, where an electron first fills an interface-trap state and then tunnels into the oxide to neutralize a positively charged oxide trap. Since the filling of interface traps is dependent on the surface potential (as well as the interface trap response time), this may very well explain why we experimentally observe this stretch-out of the subthreshold *I-V* characteristics. The model will need to be modified to account for this proposed two-step process. This in turn will require the calculation of the timedependent occupational density of the interface traps, since this time dependence will determine the number of available filled or empty states from which electrons can tunnel to or from near-interfacial oxide traps.

6.8 Summary

In summary, a well-established tunneling model based on the WKB Approximation was employed to simulate the charging and uncharging of near-interfacial oxide traps. This work extends that model by allowing for a simultaneous two-way process. The calculation of a long-term steady-state level of trap occupation dependent on the distance from the interface and the local oxide field, as well as the assumed energy level of the trap, is of great importance, since it allows for a simple incorporation of the tunneling front concept. Traps in the wake of this front have changed charge state—or more accurately—have achieved their final steady state value. Those traps still ahead of this front have not yet been affected, thus accounting for the time dependence of both the bias stress and of the measurement. Since the probability of tunneling decreases with increasing oxide depth, it is necessary to take increasingly larger time steps at later times in order to facilitate the calculation process.

Examples of the simulated changes in the occupied oxide trap level for a longterm bias stress as well as the bias applied during the measurement were presented, as was the outline of the tunneling model algorithm.

The results of the model were compared with experimental results presented in earlier chapters, and were found to be quite good for the most part in explaining the dependence on measurement time, stress time, and stress bias. However, the model does not account for the additional stretch-out of the subthreshold *I-V* characteristics during a sweep up in bias following a negative-bias stress. It was suggestd that it is likely that at least some of the oxide-trap states lie within the bandgap of SiC, and therefore a two-step

163

process involving the filling and emptying of interface traps will probably need to be incorporated to improve the present model.

7 Effect of V_T Instability on SiC Power MOSFET Reliability

7.1 Introduction

Given the voltage instability described in the previous chapters, and given the Army's interest in the development of reliable, efficient, voltage-controlled, normallyoff, high-power, high-voltage, and high-temperature SiC switches, it is important to ask whether these effects are also observed in the vertical SiC power DMOSFET; and if so, what are the implications for device operation in circuits and systems of interest.

Therefore, following a description of the power devices, experimental results are presented showing that the effects of gate-bias stressing in SiC power DMOSFETs are very similar to those in lateral test structures, and that *ON*-state current stressing leads to an increase in V_T instability. Furthermore, this increase in instability appears to be due to an increase in device temperature from self-heating but the presence of an elevated temperature is significant only when a bias is applied. The V_T instability effect is likely due to electrons tunneling into and out of near-interfacial oxide traps that extend spatially into the gate oxide form the SiC interface, and the increase in V_T instability due to hightemperature bias stressing is quite possibly due to the activation of additional nearinterfacial oxide traps. This activation also leads to an increase in stretch-out in the subthreshold *I-V* characteristics following a negative gate-bias stress. Finally, the complex time, temperature, and bias dependence of the I_D - V_{GS} characteristics leads to the conclusion that present reliability test standards for high-temperature gate bias (HTGB) stressing based on Si technology are inadequate for SiC power MOSFETs.

165

The SiC DMOSFET is formed by first implanting a *p*-well into an *n*-type epi, followed by a second implantation into the *p*-well to form the n^+ -source region (see Figure 7-1). This creates a lateral channel followed by a vertical drift region, whose thickness is determined by how much voltage it is required to block, with the drain located at the bottom n^+ -substrate. Devices anywhere from 1.2 kV to 10 kV have been demonstrated [11, 12, 125]. This chapter will focus on the effects of the voltage instability on 1.2kV 4H SiC DMOSFETs with a drift layer thickness of about 10 µm. The devices used in this study range in rating (and therefore in size) from 5-A all the way up to 50-A, and were manufactured between 2005 and 2010 and were considered stateof-the-art at the time of manufacture. The older, smaller area devices were used in the initial comparisons of gate-bias instability, whereas the larger, more recent devices were used to assess the more complicated effects of bias and temperature.



Figure 7-1: Schematic of the power DMOSFET which has a double implant to form the source and channel region [11].
Ideally, power switches such as the SiC DMOSFET will have a very low *ON*-state resistance and very low leakage current in the *OFF* or blocking state. To achieve this, the effective channel mobility should be as high as possible and the threshold voltage should be positive with enough margin so that no subthreshold current flows in the *OFF* state. As discussed in Chapter 2, the presence of interfacial charge can decrease the effective mobility by decreasing the number of free carriers and increasing coulombic scattering. And, of course, it can also shift the threshold-voltage.

It is also very important to keep in mind the results of the fast *I-V* measurements described in Chapter 5, which showed that the gate-voltage ramp speed is directly related to how much instability is recorded. Figure 5-2 implies that the actual V_T shift may be as much as a volt or more, even when a more standard 1-s measurement indicates a V_T shift of no more than a quarter or third of a volt. This greater-than-realized instability can potentially affect the performance and reliability of a power SiC DMOSFET in at least two ways.

First, as is indicated in Figure 7-2, if the threshold voltage is not set high enough, then a negative shift of the I_D - V_{GS} characteristic can lead to a substantial increase in the subthreshold leakage current. For a power DMOSFET, this would mean an increase in drain leakage in the *OFF* state and therefore a decrease in its functional blocking voltage—even when V_T using the linear *I*-V extrapolation appears to be fine. For example, Figure 7-2 shows how a shift of just 0.5 V could lead to a two order-ofmagnitude increase in leakage current. The larger actual V_T -instability shift, combined with the well understood phenomenon of decreasing threshold voltage with increasing temperature because of fewer filled interface traps [118] and an increase in thermal

carriers, means that even more margin must be built into the design of the room temperature threshold voltage to remain reliable at elevated temperature.



Figure 7-2: Illustration of potential reliability issues due to V_T instability effect. Subthreshold leakage current may increase dramatically in the *OFF* state if the threshold voltage is not set properly.

Second, a positive V_T shift can decrease performance by increasing the resistance in the ON state by increasing the threshold voltage and thus reducing the value of V_{GS} - V_T . While it is critical not to increase the drain leakage in the blocking state, it is also important not to set the threshold too high, especially since increased threshold voltage may be coupled with lower effective mobility [81]. Thus, it is important to accurately estimate the magnitude of the actual threshold-voltage instability so as to find the optimum threshold voltage to achieve the best combination of low *OFF*-state leakage with low *ON*-state resistance. As long as the threshold voltage is set high enough to preclude the onset of subthreshold drain leakage current in the blocking state, then the primary effect of this instability is to increase the *ON*-state resistance. For well-behaved power DMOSFETs, this would increase the power loss by no more than a few percent.

7.2 Gate-bias stress-induced threshold-voltage instability in SiC power DMOSFETs

This section presents some basic bias-stress induced threshold-voltage instability measurements on fully processed 4-H SiC power DMOSFETs as a function of bias-stress and time—at room temperature. The effects of elevated temperature will be addressed in subsequent sections. The basic experimental procedure was previously described in Chapter 4. In essence, a gate-bias stress was applied for a certain bias-stress time, the effect was measured by ramping the gate bias, and then a gate-bias stress of opposite polarity was applied for the same bias-stress time before re-measuring (see Figure 7-3). The measurements of the I_D - V_{GS} characteristic were all made using an Agilent 4155 parameter analyzer, with a gate-ramp speed of about 1 s and with 50 mV applied to the drain. The source and drain of these three-terminal devices were both grounded during the gate-bias stress. The average V_T instability versus stress time was calculated using two different methods: linear extrapolation to zero current; and the voltage corresponding to a set current.

Under normal operating conditions, when the device is in the ON state, a positive gate bias of 15 or 20 V is applied. In the blocking state, although the gate bias is zero, the large voltage on the drain leads to a negative field across the gate oxide. This effect has been simulated by applying a negative bias on the gate while grounding the drain.



Figure 7-3: Schematic of the test-and-measurement sequence used for the roomtemperature gate-bias stress.

Figure 7-4 shows a comparison of the gate-bias stress-induced V_T instability for 5-A DMOSFETs and for lateral MOSFETs with similar gate oxides. The as-processed epi lateral MOSFET has a slightly lower average instability, but the lateral MOSFET with an implanted epi region that simulates the implanted channel of the DMOSFET shows an average instability versus bias-stress time that is right in line with the response of the DMOSFETs. This result gives added confidence that we are looking at the same effect in the fully processed high-power vertical DMOSFETs as in the low-power lateral test structures whose results were presented in Chapters 4 and 5. In particular, these results are consistent with those of Figure 4-14, where the middle curve of that figure shows that the V_T instability of an implanted channel device with an NO anneal, such as might be found in the DMOSFET, is marginally worse than that for an as-grown epi channel device. It should be noted that the more recent state-of-the-art DMOSFETs have much improved V_T instability characteristics from those of just a few years back where gatebias stressing at room temperature produced instabilities greater than 1 V even with relatively slow 1-s *I-V* measurements [36].



Figure 7-4: Comparison of the average threshold-voltage instability of several 5-A SiC DMOSFETs with that of lateral MOSFETs with and without an implanted channel vs bias-stress time, calculated by finding the shift in the linearly extrapolated V_T .

Figure 7-5 shows the effect of the magnitude of the gate-bias stress on 20-A devices. The bottom curve is of a device stressed with ± 15 V on the gate (± 2 MV/cm). Stressing with ± 23 V (top diamond curve, ± 3 MV/cm) applied to the gate leads to an increase in the instability of between 0.1 and 0.2 V. This response is similar to that presented on lateral test structures (see Figure 4-9). Similar variations of V_T shift with stress time on newer 50-A devices have also recently been reported [53]. These similar

results on fully processed SiC power MOSFETs indicate that the same charge trapping and de-trapping effect via a direct tunneling mechanism is in all likelihood occurring.



Figure 7-5: Average threshold-voltage instability for the 20-A SiC DMOSFET vs stress time as a function of gate-bias stress, calculated by finding the voltage shift for a constant current of 90 μ A.

7.3 Effect of *ON*-state current stress on the threshold-voltage instability in SiC power DMOSFETs

The work in the previous section showed that a simple gate-bias stress at room temperature with no current flowing can cause significant shifts in the threshold voltage of SiC DMOSFETs. Since the actual device stress condition under positive bias is in the *ON* state, it is important to investigate the magnitude of the instability with drain current flowing and compare it to a simple gate-bias stress.

An initial investigation of *ON*-state current stressing was reported several years back, and researchers did observe a slight degradation of the *ON*-state I_D - V_{DS} characteristics following several hundred hours of stress time [126]. However, their measurement procedure was such that they likely mitigated a significant part of the response by following a positive bias *ON*-state stress with the measurement of a family of I_D - V_{DS} curves starting with $V_{GS} = 0$ V. For similar reasons, they did not observe a significant shift in the I_D - V_{GS} characteristics. As discussed in Chapters 4 and 5, the bias applied during the measurement can significantly affect the result of the bias during stress.

Figure 7-6 shows the experimental procedure used for the *ON*-state bias-stress testing. For the *ON*-state stress, in addition to the positive gate bias, a voltage is applied to the drain (typically 1.5 to 2.0 V) of the DMOSFET to allow the rated current to flow from source to drain. The remainder of the sequence is the same for both tests: the gate is swept down in voltage to measure the new I_D - V_{GS} characteristic; a negative gate-bias stress of equal duration to the positive-bias stress is then applied—this time with the source and drain grounded, followed by a final measurement with the gate swept up in voltage. As before, all the measurements were performed at room temperature with an Agilent 4155 using a high power measurement unit, with a small V_{DS} of 50 mV applied. It took approximately 1 second to perform the measurement.



Figure 7-6: Stress-and-measurement sequence employed to investigate the effect of *ON*-state stressing on the threshold-voltage instability of SiC power DMOSFETs (compare with Figure 7-3 for the gate-bias stress-and-measurement sequence).

Figure 7-7 plots the instability due to this *ON*-state stress for a range of stress times, from 100 to 10,000 seconds, and compares it to the instability due to a simple gatebias stress with no current flowing. The *ON*-state stress does cause a larger instability, increasing by about 20 to 25 percent. The difference between the two stresses increases with increasing stress time; the simple gate-bias stress results show the typical linearwith-log-time response whereas the *ON*-state stress results appear to be accelerating somewhat with increased stress time. This effect is due primarily to an increasing positive shift of the I_D - V_{GS} characteristic during the *ON*-state or positive-bias stress with stress time, indicating that an additional mechanism may be coming into play. On the other hand, there is very little shift to the left compared with the initial values. This last result is consistent with the effect of bias stress polarity for modern thermal oxides shown in Figure 4-11. In this case, the threshold-voltage instability was calculated by comparing the shift in V_{GS} when $I_D = 0.1$ A.



Figure 7-7: Comparison of V_T instability in a DMOSFET due to *ON*-state stressing, with 20 A flowing, and that due to simple gate-bias stressing, with $V_{DS} = 0$ V. In each case $V_{GS} = +15$ V was applied during the first half of the stress, and a simple negative gate-bias stress with $V_{GS} = -15$ V and $V_{DS} = 0$ V was applied during the second half of the stress.

The threshold-voltage instability can be much larger when calculated by comparing the shift in V_{GS} when $I_D = 1 \times 10^{-5}$ A, to quantify the effect of the stress on the subthreshold *I-V* characteristics. This is shown in Figure 7-8. A direct comparison shows that the instability of the *I-V* characteristic due to *ON*-state stressing is about twice as large at the much lower current level. This is due to a significant stretch-out of the subthreshold *I-V* characteristics following the negative gate bias stress at longer stress times. This result suggests that under actual operating conditions, an even larger positive threshold voltage margin will be needed to prevent leakage in the *OFF* state.



Figure 7-8: Comparison of *ON*-state stress induced voltage instability when measured using three different methods: for $I_D = 1 \times 10^{-5}$ A, for $I_D = 0.1$ A, and finding V_T by extrapolating the linear *I-V* curve to zero current.

Figure 7-9 shows a comparison of the back-and-forth instability of the actual subthreshold *I-V* characteristics for a state-of-the-art 50-A SiC MOSFET which has been subjected to a very long 10,000-s *ON*-state stress (see Figure 7-6 for the *ON*-state stress experimental sequence), along with 10,000-s gate-bias stress instability measurements before and after the *ON*-state stress (see Figure 7-3 for those gate-bias stress experimental sequences).

It is clear that the application of the *ON*-state stress causes an increased instability in both directions and a significant increase in the stretch-out of the *I-V* characteristic under negative bias. As discussed in Chapter 5, this larger stretch-out when sweeping up in gate bias is likely due to oxide traps filling during the measurement. The increase in this stretch-out following the *ON*-state stress may be due to an increase in the number of active oxide traps, to be discussed more fully in a later section below.

It is also interesting to note that for the gate-bias stress only sequence (see again Figure 7-3) following the *ON*-state stress sequence, the *I-V* characteristic following positive-bias stress has moved significantly back to the left; and the *I-V* characteristic following long negative bias-stresses moves even further to the left.



Figure 7-9: Gate-bias stress subthreshold I_D - V_{GS} instability characteristics of a 50-A SiC power MOSFET as the result of an *ON*-state stress test. Pre- and post-*ON*-state-stress measurements are shown as well.

Figure 7-10 shows similar results for a 20-A device, with the gate voltage required to achieve a drain current of 1×10^{-5} A plotted for a sequence of gate-bias instability measurements: before (Pre), immediately after (During), and a day after (Post) the application of an *ON*-state current stress. In this case, the stress time is plotted on a linear time scale and is not re-started at the beginning of each of the three segments, thus

representing cumulative stress time. As a result, the initial increase in the back-and-forth instability of the *I-V* characteristic with increasing gate-bias stress time (first of the three segments) does not look linear. However, the instability clearly does increase with stress time, and that instability increases dramatically with the application of the *ON*-state stress (middle segment).

But as was the case for the results shown in Figure 7-9 above, when the gate-bias stress is repeated a day later (third segment), the *I-V* characteristic following a positive-bias stress has initially moved significantly back to the left before once again moving to the right with increasing bias-stress time. This overnight shift occurred without bias being applied to the gate.



Figure 7-10: Gate voltage required for $I_D = 1 \times 10^{-5}$ A (with $V_{DS} = 0.1$ V) of a 20-A SiC DMOSFET during a measurement following both positive and negative-bias stress: (a) before (Pre) an *ON*-state stress, (b) immediately after (During) an *ON*-state stress, and (c) a day after (Post) an *ON*-state stress. The cumulative stress time is plotted on a linear scale.

To further investigate this curious effect, the instability in the gate voltage required to obtain a drain current of 1×10^{-5} A for a 50-A MOSFET is plotted in Figure 7-11 for both an ON-state current stress sequence, and for a series of separate 10,000-s regular gate-bias stress sequences immediately following. Again, note the sharp increase in instability under the longest ON-state current stress, due to a greater shift of the I-Vcharacteristic in both directions, but especially under negative bias. The magnitude of this back-and-forth instability begins to rapidly decrease during the subsequent sequence of 10,000-s gate-bias-only stresses. This is primarily due to a negative shift in the I-Vcharacteristic following the positive gate-bias stress, as was seen in Figure 7-9 and Figure 7-10. This type of behavior is only observed following the termination of an ON-state stress, suggesting that a different or additional mechanism is affecting the instability in this case. Although a large current flows during the ON-state, the drain-to-source bias is no more than 2 V and the field in the gate oxide is no more than 2 MV/cm. This effect may be due either to the deactivation of some oxide traps over time once the device is back at room temperature, or to the possible annealing of some interface traps.



Figure 7-11: Subthreshold instability of a 50-A SiC DMOSFET versus variation in stress time for an *ON*-state stress, followed immediately by several 10,000-s gate-bias stress sequences plotted versus cumulative stress time.

A comparison of the *ON*-state stress-induced instability in the subthreshold region $(I_D = 1 \times 10^{-5} \text{ A})$ between the 20-A and 50-A MOSFETs discussed above is shown in Figure 7-12. The 50-A device has a smaller instability. Judging by the results of an infra red (IR) thermal camera [127], these devices clearly underwent self-heating and required active cooling during the *ON*-state testing, with the 20-A device running much cooler (65 °C vs 145 °C) even though the current densities were similar. The results of a second 50-A device, which ran at an intermediate temperature (100 °C), are also shown in Figure

7-12. This suggests that temperature during the stress (all the measurements were made at 30 $^{\circ}$ C) is likely an important factor.



Figure 7-12: Subthreshold instability of various SiC DMOSFETs during an *ON*-state stress. Temperature variation is due to variations in self-heating, measured using an IR thermal camera.

These results suggest that not only does the *ON*-state stressing of a SiC DMOSFET increase the threshold-voltage instability due to increased charge trapping, but that the long-term effect is to shift the whole envelope of back-and-forth subthreshold *I-V* characteristics to the left. Whether the overall effect is to increase the amount of positive oxide trap charge or to decrease the amount of negative interface trap charge, the margin for *OFF*-state leakage current is clearly less following *ON*-state stressing.

7.4 High-temperature emulation of *ON*-state stress

It was shown in the previous section that ON-state stressing, where the rated current flows through the channel during the stress, resulted in a larger V_T instability than that due to gate-bias stressing alone. It was also shown that device temperatures increased during the ON-state stress and that at the highest device temperature this instability increased noticeably at longer total stress times. In order to confirm that it is the increase in device temperature that is the primary cause of the increased back-andforth instability effect, a high-temperature emulation of the ON-state stress, with no channel current, was performed.

Similar 50-A 4H SiC power DMOSFETs as were used for the *ON*-state stress tests discussed in the previous section were subjected to the following high-temperature and bias-stress testing sequence, as shown in Figure 7-13. The test method used to perform *ON*-state stress measurements was closely emulated (compare with Figure 7-6), except that instead of allowing the rated current to flow through the device, the device under test (DUT) simply had a positive gate-bias stress applied and was heated to 150 °C to simulate the self-heating effect of the *ON*-state stress. Following the positive-bias stress, the DUT was quickly cooled down to room temperature in about one minute and the I_D - V_{GS} characteristic was measured, with the gate swept down in voltage. This was then followed by a negative gate-bias stress at room temperature of equal duration to the positive-bias stress previously applied, with –15 V on the gate and the source and drain again grounded; followed by a second measurement of I_D - V_{GS} with the gate swept up in voltage. Room temperature gate bias stress-and-measure bias instability sequences (see

Figure 7-3) were also performed both before and after the high-temperature-stress test method.

Once again, all the measurements were performed with an Agilent 4155, with a small V_{DS} of 50 mV applied. The threshold-voltage instability in this study was calculated both by comparing the shift in V_T using the standard linear *I*-*V* extrapolation method to zero current (" V_T method"), as well as by the shift in V_{GS} when $I_D = 1 \times 10^{-5}$ A (" V_{GS} method") in order to quantify the effect of the stress on the subthreshold *I*-*V* characteristics.



Figure 7-13: Stress-and-measurement sequence employed to investigate the effect of an emulated *ON*-state stressing on the threshold-voltage instability of SiC power DMOSFETs (compare with Figure 7-6).

Figure 7-14 shows the back-and-forth instability of the I_D - V_{GS} characteristics, plotted on a log current scale, when subjected to a 10,000-s high-temperature emulation of the *ON*-state stress. The curve for the room-temperature measurement following the positive gate-bias stress at 150 °C is at the far right (lightly shaded circular symbols) and the curve following the negative-bias stress is at the far left (darkly shaded triangular symbols). Also shown are room-temperature stress-and-measure bias instability curves for measurements performed both before (solid lines) and long after (dashed lines) the emulated *ON*-state stress sequence, also with stress periods of 10,000 s.

Clearly, the emulated *ON*-state stress causes a significant shift to the right following positive-bias stress as well as a significant shift to the left following negativebias stress compared to the pre-stress curves. In addition, the subthreshold *I-V* characteristic is significantly stretched-out following the negative-bias stress. In this case, the subthreshold *I-V* characteristic is also clearly stretched-out to some extent following the positive gate bias stress when compared to the pre-emulated *ON*-state stress values. This may be evidence either for a very large increase in the number of active oxide traps, or to the creation of additional interface traps during the high-temperature stress [89, 128].

The post emulated *ON*-state stress measurements, taken long after the hightemperature stress, show a significant shift back to the left following a positive-bias stress, but no discernible change following a negative-bias stress, including the permanent looking stretch-out of the subthreshold *I-V* characteristic. These results are very reminiscent of those for the actual *ON*-state stress.



Figure 7-14: Example of the instability in the I_D - V_{GS} characteristics of a power MOSFET as the result of a high-temperature-stress emulation of an *ON*-state stress test. Pre- and posthigh-temperature-stress measurements are shown as well.

Figure 7-15 and Figure 7-16 depicts the back-and-forth instability due to the emulated *ON*-state stress depicted in Figure 7-13 as a function of stress time. The total instability is shown in Figure 7-15, and is seen to follow the typical high-temperature result—that the total instability increases faster than the linear-with-log-stress-time rate seen at room temperature (compare with Figure 7-11 and Figure 4-22).



Figure 7-15: Comparison of the total back-and-forth instability of the I_D - V_{GS} characteristics versus stress time for a high-temperature-stress emulation of the *ON*-state stress calculated using both the V_T and V_{GS} analysis methods.

Figure 7-16 shows that the majority of this increased instability is due to a shift to the right under positive bias when measured on a linear current scale, but a shift to the left under negative bias when measured on a log scale. This implies that both the *ON*-state resistance and *OFF*-state leakage current are increasing with increasing high-temperature stress time. It is also interesting to note that although the envelopes of the back-and-forth instability shift differently for the two methods of analysis, the total instabilities increase at the same rate in this case. Figure 7-15 also shows the post high-temperature stress results of a simple room temperature gate-bias stress as described in

Figure 7-3. In this case, there is a noticeable difference, with the back-and-forth instability decreasing sharply when measured on a linear current scale (V_T method).



Figure 7-16: Comparison of the shift under both positive and negative-bias stress of the I_D - V_{GS} characteristics versus stress time for a high-temperature-stress emulation of the ON-state stress calculated using both the V_T and V_{GS} analysis methods.

In summary, the results depicted in Figure 7-14 through Figure 7-16 are in general very similar to those reported previously for devices subjected to an *ON*-state stress, strongly suggesting that the increased instability observed is due primarily to self-heating and not to the presence of a large current flowing from source to drain.

7.5 Degradation of ON-state characteristics

Figure 7-17 shows the *ON*-state I_D - V_{DS} characteristics for a 20-A DMOSFET. The arrow points to the gap between the $V_{GS} = 14$ V and $V_{GS} = 16$ V curves. The results reported here show DMOSFET V_T instabilities generally between 0.25 and 0.5 V. This correlates very well with the instabilities observed in similarly processed lateral MOSFETs. We know that the actual instability (difficult to measure because of the counter-acting effect of the gate bias during the measurement) can be three or more times larger. But suppose that the actual shift in V_T is about 1 V. If V_T was originally set to preclude the chance of turning on the subthreshold current in the blocking state, then the practical effect of a threshold-voltage shift would be to reduce the effective gate voltage applied in the *ON* state. But Figure 7-17 shows that even a shift of 1 V would only lead to an increase in V_{DS} of about 0.1 V, from about 1.9 to 2.0 V. This would increase the *ON*-state losses about 5 percent, which should be manageable for power converter applications.

Figure 7-18 in fact shows the change in the I_D - V_{DS} characteristic with $V_{GS} = 15$ V before and after a 1-hour *ON*-state stress. V_{DS} must increase about 0.1 V following the stress to maintain 20 A of drain current. This is the same result as derived from Figure 7-17, which results in about a five percent increase in $R_{DS,ON}$.



Figure 7-17: *ON*-state I_D - V_{DS} as a function of V_{GS} . The effect of the V_T instability would be to increase the *ON*-state resistance only by a few percent.



Figure 7-18: A 1-hour *ON*-state stress leads to an increase in *ON*-state resistance. V_{DS} must increase about 0.1 V following the stress to maintain 20 A of drain current.

7.6 High-temperature bias-stress mechanisms

With the significant and super-linear increase with log time of the V_T instability that has been observed when stressing at elevated temperature in both power DMOSFETs (see Figure 7-8, Figure 7-11, and Figure 7-15) and lateral test structures (see Figure 4-22), it is critical to understand and limit the underlying basic mechanism(s). Especially since SiC power MOSFETs are expected to operate reliably at elevated temperatures, and an increased V_T instability can lead to increased leakage current in the *OFF* state and possibly to device failure.

At first glance, this large increase in V_T instability at elevated temperature is unexpected, since the direct tunneling mechanism described in Chapter 6 that successfully explains the V_T -instability effect at room temperature should not be very sensitive to temperature. In this model, electrons are simultaneously tunneling to and from near-interfacial oxide traps that vary spatially with distance from the SiC interface, and with the local oxide field and the fraction of filled or empty traps determining which mechanism dominates.

However, very recent results in the literature may provide key clues to explaining this strong temperature sensitivity. First, it has been reported that E'-type oxide defects that are associated with an O-vacancy have been observed in SiC MOSFETs using electrically detected magnetic resonance techniques [61]. No other type of oxide defect has yet to be found using any ESR technique. Since this was the main type of oxide defect in irradiated Si MOSFETs, which showed a similar type of V_T -instability effect that was successfully explained using a tunneling model [88, 89, 90], it is likely that this is the same general mechanism that is occurring in SiC MOSFETs. Especially given the

experimental evidence described in Chapters 4 and 5, including: the linear-with-log-time increase in V_T instability with stress time, the extrapolation to zero V_T instability to stress times comparable with the first tunneling transition times, and the larger observed V_T instability with faster measurement times.

Second, it has also been recently reported using on-the-fly electron spin resonance techniques in the study of NBTI in Si MOS that the number of these E' centers increased dramatically with bias at an elevated temperature of 100 °C, but not with temperature stress alone [96]. In addition, they found that the number of oxide defects markedly decreased when the sample was returned to room temperature. This result may in fact explain the unexpected increase in the V_T instability at elevated temperature.

To further test this theory, the following experiment was very recently performed [51], consisting of a two-step stress sequence (see Figure 7-19). First, the device was subjected to a 150 °C high-temperature stress for varying periods of time, with or without a positive gate bias applied. Second, a simple gate-bias stress-and-measure instability sequence was applied—except that the whole sequence was performed at 150 °C. The positive and negative-bias stress periods were a constant 60-s long, regardless of how long the previous temperature stress was. Room temperature stress-and-measure bias instability sequences were also performed, both before and after the high-temperature-stress test method.



Time

Figure 7-19: Short stress-and-measurement instability sequence to determine effect of hightemperature stress with or without applied gate-bias stress.

Figure 7-20 and Figure 7-21 present the results of this test method, depicted in Figure 7-19, for determining the effects of bias stress at elevated temperature. Figure 7-20 shows that when bias is applied during the time-dependent high-temperature stress, the total back-and-forth instability once again increases super-linearly with log time, as in Figure 7-15 for example, although for this test method it takes one hundred times longer to see a similar increase in the magnitude of the instability since the 60-s stress-andmeasurement sequence used to measure the effect is so much shorter. Interestingly, there is practically no increase in the total instability when no bias is applied during the 150 °C time-dependent stress. Figure 7-21 shows that once again there is an increase in both the positive shift following positive bias and the negative shift following negative bias during the stress-and-measurement sequence used to measure the effect of the stress, although in this case there is no noticeable difference regardless of whether the V_T instability is measured on a linear current scale (using the V_T method) or on a logarithmic current scale (using the V_{GS} method) for these high temperature measurements.



Figure 7-20: Short stress-and-measurement instability sequence to determine effect of high-temperature stress with or without applied gate-bias stress.



Figure 7-21: Short stress-and-measurement instability sequence to determine effect of hightemperature stress with or without applied gate-bias stress.

These experimental results are consistent with the recent results in the literature. The V_T instability only increases at high temperature when bias is applied, and the subsequent room temperature instability is much smaller, especially when measured using the V_T method. Since the V_T instability is likely a measure of the number of active near-interfacial oxide traps—keeping in mind that faster *I-V* measurements reveal more total traps, these results suggest that this increase in threshold-voltage instability at elevated temperature is due to the activation of additional near-interfacial oxide traps related to the *E'*-center defect. If this is so, then it is important to develop improved processing methods to either decrease the number of precursor oxide defect sites, by somehow increasing the O concentration in the interfacial layer or tying up the dangling bonds as the NO anneal may do, or to reduce the bond strain in this region of the oxide so that pre-cursor Si-Si bonds are less likely to break.

Otherwise, an increased negative shift can give rise to increased leakage current in the *OFF*-state and potential device failure if proper precautions are not met to provide an adequate margin for the threshold voltage. Of course, increasing V_T too much will lead to an unnecessary increase in *ON*-state resistance and reduced device performance. Therefore, it is important to also develop improved test methods that will more accurately ensure device reliability.

If the activation of additional oxide traps can be properly quantified with respect to temperature and oxide field (see Chapter 8: Future work), and if the critical leakage current causing device failure is known, then the time-to-failure due to negative-bias temperature stress could be determined.

The change in the log of the drain current can be found from the change in the threshold voltage in the subthreshold region using Equations (3-56) and (5-4), dependent of course upon the initial threshold voltage and how much voltage shift is required to raise the drain current above the noise floor. The change in threshold voltage due to a change in oxide trap charge is given by (4-1), yielding the following relationship between the change in oxide-trap charge and drain current.

$$\Delta \log(I_D) = \frac{-\Delta V_T}{S} = \frac{q \cdot \Delta N_{OT}}{S \cdot C_{OX}}$$
(7-1)

For a uniform spatial distribution of oxide traps, the change in drain current as a function of time is, using (6-9) and (6-10), given by

$$\Delta \log(I_D) = \frac{-\Delta V_T}{S} = \frac{q \cdot \Delta N_{OT}}{S \cdot C_{OX}} = \frac{q}{S \cdot C_{OX}} \cdot \frac{N(t)}{2 \cdot \beta} \cdot \ln\left(\frac{t}{t_0}\right)$$
(7-2)

where N(t) is the volumetric active oxide-trap density which increases with time at temperature, as a function of applied bias.

A schematic of the increase in the number of positively charged oxide traps with time under negative bias at high temperature, due to both the tunneling front extending deeper into the oxide and an increase in the number of active traps, is shown in Figure 7-23.



Figure 7-22: Schematic showing the increase in the number of positively charged oxide traps with time under negative bias at high temperature, due to both the tunneling front extending deeper into the oxide and an increase in the number of active traps.

7.7 Reliability testing

The feasibility of using SiC MOSFET devices for power electronics applications has been recently demonstrated with the development of several power converter modules [29, 12]. However, before these devices can be properly qualified, device reliability issues, including the effect of threshold-voltage instability, must first be resolved. Although there exist industrial and military standards for stress test qualification of semiconductor devices (e.g Joint Electron Devices Engineering Council (JEDEC) JESD22-A108C [129], MIL-STD-750E [130], and Automotive Electronics Council (AEC) AEC-Q101 [131]), these standards are not self-consistent and are based on Si device technology, and therefore do not take into account the complex time, bias, and temperature dependence of the V_T instability in SiC MOSFETs. The consequence of this complex response, the evidence for which has been reported in Chapters 4, 5, and 7, is that how you test and measure will significantly affect your results. Furthermore, it is critical that the testing be relevant to the intended application. This means that the effects of AC switching must be addressed, as well as the effects of operating at elevated temperatures, and that the worst-case operating stress conditions need to be identified. This includes determining whether recent improvements regarding smaller V_T instabilities are being achieved by balancing mobile ion drift with charge trapping effects, and if so, are there any conditions for which this would cause a reliability issue.

Green, Lelis, and Habersat [53] have very recently presented an initial investigation of the effects of HTGB stress on the reliability of 4H-SiC power MOSFET devices within the guidelines of the accepted industrial and military standards for stresstest qualification mentioned above. They showed that a large variation occurs in the I_D - V_{GS} characteristics of a SiC power MOSFET following a one-hour gate-bias stress at 150 °C with $V_{GS} = +15$ V, depending on whether the measurement is made immediately after rapidly cooling back to room temperature, or after a delay of one hour (see Figure 7-23). To further dramatize the difference, the immediate measurement was made by sweeping down in gate bias, from +15 to 0 V, whereas the later measurement was made by sweeping up in gate bias in the conventional manner. They found that the initial poststress measurement would result in device failure based on the AEC standard which requires that post burn-in measurement parameters remain within \pm 20 percent of their pre-stress values, whereas the second, later measurement with the one-hour delay would

remain within the specifications. The important point is that both measurement procedures are acceptable within the existing standards. This clearly illustrates the effect of the time-dependent nature of the charge trapping on HTGB stress-testing results.



Figure 7-23: Large variation in the device room-temperature I_D - V_{GS} characteristic curves following a positive bias HTGB stress test at 150 °C, depending on measurement delay time.

Figure 7-24(a) shows a variation in response following a one-hour gate-bias stress at 150 °C with $V_{GS} = -15$ V. Although the variation in V_T between the immediate and one-hour delayed measurements is not as great in this case, the immediate measurement does result in an increase of the *OFF*-state leakage current due to the larger negative shift in V_T . More dramatic is the effect on V_T shift and increased leakage current when measuring at temperature, as shown in Figure 7-24 (b), since V_T naturally decreases with increasing temperature. This underlines the fact that not only do the present test standards allow for a large variation in HTGB test results for SiC MOSFETs, they also do not address the large temperature effect observed.



Figure 7-24: Drain current plotted on a linear scale as a function of V_{GS} following a negative bias HTGB stress test comparing (a) immediate room temperature I_D - V_{GS} characteristics with a one-hour delay time and (b) immediate high temperature I_D - V_{GS} characteristics versus a room-temperature measurement with a one-hour delay time.

These results illustrate the need for more stringent test and measurement standards to avoid inconsistent pass/fail results for SiC MOSFETs. For example, the present JEDEC standard [129] requires that post burn-in electrical measurements be completed as soon as possible, but no longer than 96 hours after removal of the bias. The above results of Green, Lelis, et al. [53] show the inadequacies of allowing a full one-hour window for the post-stress measurements when qualifying SiC MOSFET devices. In addition, the complex time, temperature, and bias dependence of the V_T -instability, likely due to the charging and discharging of near-interfacial oxide traps through a direct

tunneling mechanism, causes the drain current versus gate-to-source voltage (I_D - V_{GS}) characteristics to be sensitive to both the measurement sweep speed and direction, neither of which are addressed at all by the present standards.

Although the present standards call for pre- and post-bias temperature stress measurements such as HTGB to be performed at room temperature, it is important to measure at the stress temperature as well since SiC power MOSFETs are expected to operate at these temperatures and the leakage current under negative-bias stress is worse at elevated temperatures. It is also critical that the bias -stress times versus temperature be long enough to allow for the activation of all the elevated temperature mechanisms which may occur under actual operational conditions. Charge trapping effects will get worse, although for some devices this effect may be countered by mobile ion drift. It is also important to determine appropriate accelerated test conditions so that nonoperational failure mechanisms are not introduced. Elevated temperature in the range of 150 °C cannot be a test accelerant if the devices are expected to operate at this elevated temperature.

Finally, it is important that tests relevant to the device application are employed, such as AC switching tests under the correct worst-case conditions. For a power MOSFET in a typical power conversion circuit, the device will either be switching *ON* and *OFF*, at a frequency of around 20 kHz, or be held in the *OFF* or blocking state [18]. The most appropriate AC stress test would be one where both the gate bias and drain bias switches as in an actual power conversion circuit: for example, from $V_{GS} = +15$ V and V_{DS} = +2 V to $V_{GS} = 0$ V and $V_{DS} = +600$ V. In lieu of that, it is appropriate to switch V_{GS}

from +15 V to $V_{GS} = -5$ or -10 V to simulate the negative field that develops across the gate oxide in the blocking mode.

The most appropriate blocking or HTRB stress test would be one that first switches the device *ON* and *OFF* to allow for the greatest activation of additional nearinterfacial oxide traps at high temperature under the higher oxide fields in the *ON* state before applying a long-term lower oxide field under negative-bias stress conditions. This should then be followed by another AC stress to see if the stressed device can still withstand transient bias conditions once a large number of oxide traps have been charged.

If the activation rate of additional oxide traps at high temperature and bias can be formulated, then the oxide charge trapping under HTGB and HTRB conditions could be calculated versus time, leading to a prediction of device reliability versus time.

7.8 Summary

The gate-bias stress-induced threshold-voltage instability observed in lateral SiC MOSFETs is also present in fully processed SiC DMOSFETs, and the shifts are comparable in magnitude, and similar in their response to bias-stress time, gate-oxide field, and temperature. Therefore, it is highly likely that the same near-interfacial oxide trapping mechanisms are the cause of the observed instability in both the I_D - V_{GS} and I_D - V_{DS} characteristics of SiC power MOSFETs.

Bias stressing, while also allowing the rated current to flow through the channel of the power MOSFET, results in an increase in the V_T -instability effect compared to gate-bias stressing alone. Although the effect of this *ON*-state stressing is only slightly worse when measured at high-current levels on a linear scale, it is significantly worse when measured at lower current levels on a logarithmic scale. There occurs both a significant additional positive shift of the I_D - V_{GS} characteristic following a positive-bias stress and a negative shift and increased stretch-out of the subthreshold *I*-*V* characteristics following a negative-bias stress. The *ON*-state stress results in self-heating, so it is not surprising that similar room-temperature measurement results are obtained when bias stressing at elevated temperature. The initial increase in the positive shift following positive-bias stress goes away over time after the device has returned back to room temperature, and eventually results in a smaller positive shift than before the initial *ON*-state or high-temperature stress. It is not clear whether this is due to the annealing of interface traps or to some more complicated oxide-trap mechanism.

High-temperature stress alone without an applied bias does not increase V_T instability. These results, coupled with recent results in the literature, suggest that this increased instability caused by high-temperature bias stressing is due to the activation of additional near-interfacial oxide traps, which are then free to change charge state with a change in gate bias. Many of these newly activated oxide trap defects likely become inactive again following a return to room temperature—especially over time.

This charge trapping will obviously affect the operation of the power SiC DMOSFET. Even though it is difficult to measure the full effect of any gate bias stress due to the short times needed to change the charge state of those traps closest to the interface, the effect will nevertheless occur. Switching oxide-trap densities as large as 8×10^{11} were calculated when measuring with ramp speeds of around 1 s.

Therefore it is imperative that the threshold voltage be set high enough to preclude increasing leakage current in the *OFF* state, when a negative field will be
established across the gate oxide. This is in addition to any margin designed in for negative shifts at elevated temperature due to fewer charged interface traps as well as the increases in the intrinsic carrier concentration.

While it is critical not to increase the drain leakage in the blocking state, it is also important not to set the threshold too high, especially since increased threshold voltage may be coupled with lower effective mobility. Thus, it is important to accurately estimate the magnitude of the actual threshold-voltage instability so as to find the optimum threshold voltage to achieve the best combination of low *OFF*-state leakage with low *ON*-state resistance.

The application of various existing standard high-temperature bias-stress reliability tests such as HTGB and HTRB, which are designed to assess the long-term effect of these stress conditions on the operating characteristics of a device under test, will likely need to be modified, given the great sensitivity of the SiC MOSFET V_T instability response to bias stress and temperature. Furthermore, it is important to determine the appropriate worst-case test and measurement conditions for a given circuit application, such as power conversion, and also to ensure that mobile ion drift is not masking potential reliability problems caused by charge trapping effects in nearinterfacial oxide traps.

8 Summary, Conclusions, and Future Work

8.1 Summary and conclusions of present work

My research led to the initial discovery and reporting of the basic threshold voltage (V_T) instability phenomenon in SiC MOSFET devices, first in lateral test structures and later in fully processed vertical power devices. Although this effect has been primarily studied in MOSFETs on 4H-SiC and with a thermally grown gate oxide, it has also been observed in MOSFETs made on 6H-SiC, MOSFETs with a deposited oxide, and in SiC MOS capacitors. This V_T -instability phenomenon has also been observed in SiC MOSFETs irrespective of manufacturer.

A positive gate-bias stress will cause a positive shift of the I_D - V_{GS} characteristics and a negative gate-bias stress will cause a negative shift. This effect is very repeatable. The larger the gate bias, the larger the shift, although the effect of bias magnitude tends to saturate at higher oxide fields. The longer the gate-bias stress is applied, the larger the instability observed. This increase is generally observed to occur at a linear-with-logstress-time rate. Extrapolating this linear rate back to zero instability indicates that stress times would have to be reduced to below 1 ps to prevent any V_T shift from occurring.

These effects are consistent with electrons directly tunneling to and from localized near-interfacial oxide traps which extend several nm into the gate oxide. Given the similarity of this V_T -instability phenomenon with that observed previously in irradiated Si MOSFETs with similar gate oxide thicknesses and the successful modeling of that effect with a tunneling model (wherein the first tunneling transition times into the

oxide were calculated to occur at around 0.1 ps), this work applied that model and extended it by allowing for a simultaneous two-way tunneling process. This two-way tunneling model calculates a long-term steady-state level of trap occupation, dependent on oxide depth, localized field, and assumed trap energy level, incorporating the concept of a tunneling front. Traps in the wake of this front have settled to a new steady-state trap occupation level, whereas traps deeper into the oxide and still ahead of this front have yet to be affected by the application of a new gate bias. This mechanism accounts for the time dependence of the stress bias. It also accounts for the sensitivity of the measured V_T instability with measurement time.

It is not surprising that the speed of the measurement would be so important since a gate bias is necessarily applied during the measurement, which in fact applies a bias stress of its own. The slower the measurement, the longer the time for this alternate gate bias to affect the results of the previous bias stress. Standard measurement times are on the order of 1 to 2 s. Faster measurements using a fast *I-V* system show a larger V_T instability effect, with a 20-µs gate ramp showing the largest shift of all. Conversely, slowing down a conventional parameter analyzer to a 1-ks gate ramp showed the smallest shift of all. This difference is caused by oxide traps changing charge state during the measurement.

Another effect likely caused by oxide traps changing charge state during the measurement is the larger subthreshold swing (or stretch-out of the subthreshold I_D - V_{GS} characteristic) when sweeping up in gate bias following a negative-bias stress compared to sweeping down following a positive-bias stress. This difference was also measurement-time dependent, with the much slower 1-ks long measurements showing no

difference at all. Analysis was also presented which showed that the extrapolated *I-V* instability at mid-gap current levels was comparable to the instability observed in fast *I-V* measurements. The magnitude of this V_T instability can be used to calculate a lower bound for the number of oxide traps, based on the number that switch charge state. Standard-room temperature gate-bias stressed back-and-forth V_T -instability results yield switching oxide-trap densities of 1×10^{11} to 2.5×10^{11} cm⁻². Fast *I-V* measurements and extrapolated instability have revealed switching oxide-trap densities as large as 3×10^{11} to 5×10^{11} cm⁻². These are trap densities comparable to those of interface traps. Very slow 1ks-ramp speed measurements exhibit switching oxide-trap densities half as large as those at regular 1-s ramp speeds.

The only physical defect that has been observed to date in the gate oxides of SiC MOSFETs is referred to in the literature as an E' center. The pre-cursor site is a weak Si-Si bond owing to an O vacancy, with each Si back-bonded to three different O atoms. If this Si-Si bond is broken during processing or by operational stress, it can become an active oxide trap. Studies of irradiated Si MOS test structures found that this defect behaves as a hole trap which becomes a neutral dipole if it subsequently traps an electron. It has also been suggested that this defect in its neutral charge state may also behave as an electron trap. The large super-linear-with-log-time increase in the V_T instability observed at device temperatures of 100 °C and above can be explained in terms of a bias-dependent high-temperature increase in the number of active E' center oxide defects. Switching oxide-trap densities as large as 8×10^{11} have been calculated with measurement speeds of around 1 s.

If the threshold voltage is set too low, then a negative V_T shift could cause an increase in *OFF*-state leakage current, leading to potential failure of the power MOSFET when trying to block 600 V or more in a typical application. This negative V_T shift can become much greater at elevated temperature, given both the potential increase in the number of active oxide traps and fewer charged interface traps, as well as the decrease in V_T due to the increase in the intrinsic carrier concentration. This identifies hightemperature reverse bias (HTRB) as a potential failure mode, with the significant increase in the number of positively-charged near-interfacial oxide traps the accompanying failure mechanism. If the activation rate of additional oxide traps at high temperature and bias can be formulated, then the oxide charge trapping under HTGB and HTRB conditions could be calculated versus time, leading to a prediction of device reliability versus time.

Based upon a large variation in high-temperature V_T -instability results, it is quite possible that the flat response of some recent SiC MOSFETs versus temperature is due to a balance of oxide charge trapping effects and mobile-ion drift.

A positive V_T shift under positive-bias stress will increase the *ON*-state resistance. An *ON*-state stress, during which the rated current is flowing through the channel, will increase this positive shift, due primarily to elevated-temperature effects caused by the self-heating of the device.

Given the complex time, temperature, and bias dependence of the V_T -instability effect described in this work, existing reliability standards based on Si device technology are likely inadequate for SiC MOSFETs and need to be revised. Particular attention needs to be paid to the measurement conditions. Measurements should be made quickly, without delay, and with a gate bias that is minimally disruptive. In addition, elevated

temperature is problematic as a stress accelerant, since SiC power MOSFETs are expected to operate at temperatures of 150 °C, and even higher-temperature stressing will likely lead to additional defect creation which may not necessarily occur at the operating temperature.

Processing variations have a noticeable effect on the V_T instability. Lateral SiC MOSFETs with an implanted epi channel (which mimic the implanted channel of a power DMOSFET) show a slightly larger instability than devices with an un-implanted channel. SiC MOSFETs that did not receive the standard post-oxidation NO anneal have a much larger V_T instability, with switching oxide-trap densities as large as 6×10^{11} calculated with standard 1-s measurement speeds. The reduction in oxide traps following a post-oxidation NO anneal mirrors the reduction in interface-trap density, which also decreases significantly. One possible explanation for this is that many test procedures designed to measure the number of interface traps are, at least in part, actually measuring the response of near-interfacial oxide traps. Making a distinction between the different types of interfacial charge is important, since they are likely associated with different physical defects, which in turn may respond differently to processing variations.

8.2 Future work

The work presented here can naturally lead to additional future work in a number of different areas. For example:

Validate high-temperature oxide-trap model

Confirm that it is in fact the activation of additional E' center-type oxide trap defects that is causing the large increase in V_T -instability at temperatures above 100 °C. This should be done by correlating a larger array of ESR results—both on Si and SiC MOS type structures, as a function of bias stress magnitude, polarity, and temperature with a similar array of electrical measurements.

Develop two-step oxide-trap tunneling model

Although the two-way tunneling model can accurately describe the timedependence of the V_T instability results—as well as the bias magnitude and bias polarity dependence—it does not predict the larger subthreshold swing when sweeping up in gate bias following a negative-bias stress. Therefore, the next step in the development of this simultaneous two-way oxide-trap tunneling model is to include the very likely possibility that interface traps are involved in a two-way process, such that electrons must first fill an interface trap before they can tunnel to a near-interfacial oxide trap, or conversely, that an electron tunneling out from a near-interfacial oxide trap will first fill an available interface trap.

Model transient device behavior for circuit applications

Incorporate this oxide-trap tunneling model into the existing SiC power MOSFET simulator at the University of Maryland to enable accurate prediction of device behavior in circuit applications of great importance, such as that of various power converter systems.

Establish new reliability test standards for SiC power MOSFETs

Given the strong time and bias dependence of the V_T instability effect, and given the lax and inconsistent existing standards for power electronics in terms of test time and procedure, it is important to update these standards for SiC power MOSFETs. This also includes ensuring that reliability tests relevant to the device application are applied, and that the actual worst-case conditions are identified.

In addition, methods need to be developed and applied to verify whether mobile ion drift is in some cases compensating for oxide trapping effects to reduce the V_T instability in some devices at high temperature. Furthermore, if this is, in fact, happening, determine whether this balancing of charge fails under certain conditions, leading to potential reliability issues.

Develop improved processing methods

Given that near-interfacial oxide traps play a critical role in the V_T -instability effect in SiC MOSFETs, which can in turn cause serious reliability issues, it is important to develop processing methods that decrease this defect, or the effect of this defect. Possible ways may include incorporating more O at the interface to prevent the original creation of this defect, reducing the near-interfacial bond strain, or identifying other methods similar to the NO post-oxidation anneal to help prevent the activation of this defect, or possibly to further reduce the interface trap density if these do, in fact, play an integral part in the charging and/or discharging of the oxide-trap defect.

9 References

- [1] Gary L. Harris, "Properties of Silicon Carbide," INSPEC United Kingdom 1995.
- [2] O. Kordina and S.E. Saddow, "Silicon Carbide Overview," from "Advances in Silcon Carbide Processing and Applications,"ed. S.E. Saddow and A. Agarwal, Artech House, Inc., Boston, 2004.
- J.B. Casady, A. Agarwal, S. Seshadri, R.R. Siergiej, L.B. Rowland, M.F.
 MacMillan, D.C. Sheridan, P.A. Sanger, and C.D. Brandt, "4H-SiC Power
 Devices for Use in Power Electronic Motor Control," *Solid-State Elec.*, vol. 42:12, p. 2165 (1998).
- [4] R. Schorner, P. Friedrichs, D. Peters, and D. Stephani, "Significantly Improved Performance of MOSFET's on Silicon Carbide Using the 15R-SiC Polytype," *IEEE Elec. Dev. Lett.*, vol. 20:5, p. 241 (1999).
- [5] G.Y. Chung, C.C. Tin, J.R. Williams, K. McDonald, R.K. Chanana, A. Weller,
 S.T. Pantelides, L.C. Feldman, O.W. Holland, M.K. Das, and J.W. Palmour,
 "Improved Inversion Channel Mobility for 4H-SiC MOSFETs Following High Temperature Anneals in Nitric Oxide," *IEEE Elec. Dev. Lett.*, vol. 22:4, p. 176 (2001).
- [6] B. J. Baliga, "Power Semiconductor Device Figure of Merit for High Frequency Applications," *IEEE Electron Device Lett.*, vol. 10, p. 455 (Oct. 1989).
- [7] J. A. Cooper, Jr. and A. K. Agarwal: "SiC Power-Switching Devices—The Second Electronics Revolution?" *Proc. IEEE*, vol. 90:6, p. 956 (2002).
- [8] N.R. Jankowski, L. Everhart, B.R. Geil, C.W. Tipton, J. Chaney, T. Heil, and W.Zimbeck, "Stereolithographically Fabricated Aluminum Nitride Microchannel

Substrates for Integrated Power Electronics Cooling," *IEEE 11th Intersociety Conference on Thermal and Thermomechanical Phenomena in Electronic Systems (ITHERM)*, p. 180 (2008).

- [9] J.W. Palmour, M. Das, S-H. Ryu, B. Hull, Q. Zhang, R. Callanan, A. Burk, M. O'Loughlin, Y. Khlebnikov, J. Sumakeris, and A. Agarwal, "SiC Power Devices for Energy Efficiency," plenary talk presented at the *European SiC and Related Materials Conference (ECSCRM)*, Oslo, Norway (2010).
- [10] M. Treu, R. Rupp, P. Blaschitz, K. Ruschenschmidt, T. Sekinger, P. Friedrichs, R. Elpelt, and D. Peters, "Strategic Considerations for Unipolar SiC Switch Options: JFET vs. MOSFET," *Conference Record of the 2007 IEEE Industry Applications Conference*, p. 324 (2007).
- B.A. Hull, C. Jonas, S.-H. Ryu, M. Das, M. O'Loughlin, F. Husna, R. Callanan, J. Richmond, A.Agarwal, J. Palmour, and C. Scozzie," Performance of 60 A, 1200
 V 4H-SiC DMOSFETs," *Mat. Sci. Forum*, vols. 615-617, p. 749, (2009).
- K. Matocha, P. Losee, A. Gowda, E. Delgado, G. Dunne, R. Beaupre, and L.
 Stevanovic, "Performance and Reliability of SiC MOSFETs for High-Current Power Modules," *Mat. Sci. Forum*, vols. 645-648, p. 1123 (2010).
- [13] Rohm web site: <u>http://www.rohm.com/news/101221.html</u>, "The World's First Low On-Resistance High-Speed SiC Transistor (DMOSFET)," (21 December 2010).
- [14] Cree web site: <u>http://www.cree.com/press/press_detail.asp?i=1295272745318</u>,
 "Cree Launches Industry's First Commercial Silicon Carbide Power MOSFET;

Destined to Replace Silicon Devices in High-Voltage (≥ 1200 -V) Power Electronics," (17 January 2011).

- [15] Cree data sheet from Digi-Key web site :
 <u>http://www.digikey.com/us/en/ph/Cree/cmf20120d-mosfet.html</u>.: "CMF20120D-Silicon Carbide Power MOSFET 1200V 80 mΩ Z-FeTTM MOSFET N-Channel Enhancement Mode," (2011).
- T.E. Salem, D. P. Urciuoli, R. Green, and G. K. Ovrebo, "High-Temperature High-Power Operation of a 100 A SiC DMOSFET Module," in *Proc. 2009 IEEE Applied Power Electronics Conf.*, p. 653-657 (2009).
- [17] R.A. Wood and T.E. Salem, "Characteristics of a 1200 V, 550 A SiC DMOSFET
 Dual Module," presented at 2010 PCIM (Power Electronics/Intelligent
 Motion/Power Quality) Nuremberg, Germany (2010).
- [18] D. Urciuoli, R. Green, A. Lelis, and D. Ibitayo, "Performance of a Dual, 1200 V,
 400 A, Silicon-Carbide Power MOSFET Module," *Proc. of 2010 ECCE* (2010).
- [19] J. Scofield, N. Merrett, J. Richmond, A. Agarwal, and S. Leslie, "Electrical and Thermal Performance of 1200 V, 100 A, 200 °C 4H-SiC MOSFET based Power Switch Modules," *Mat. Sci. Forum*, vols. 645-648, p. 1119 (2010).
- [20] J. A. Cooper, Jr.: "Advances in SiC MOS Technology," *Phys. Status Solidi (a)*, vol. 162, p. 305 (1997).
- [21 V. V. Afanas'ev, M. Bassler, G. Pensl, and M. Schulz, "Intrinsic SiC/SiO interface states," *Phys. Stat. Sol.* (a), vol. 162, p. 321 (1997).

- [22] M. K. Das, B. S. Um, and J. A. Cooper, Jr., "Anomalously high density of interface states near the conduction band in SiO /4H-SiC MOS devices," *Mater. Sci. Forum*, vols. 338–342, p. 1069 (2000).
- [23] A. J. Lelis, D. Habersat, R. Green, A. Ogunniyi, M. Gurfinkel, J. Suehle, and N. Goldsman, "Time Dependence of Bias-Stress-Induced SiC MOSFET Threshold-Voltage Instability Measurements," *IEEE Trans. Elec. Dev.*, vol. 55:8, p. 1835 (2008).
- [24] L.A. Lipkin and J.W. Palmour, "Insulator Investigation on SiC for Improved Reliability," *IEEE Trans. Elec. Dev.*, vol. 46:3, p 525 (1999).
- [25] M.M. Maranowski and J.A. Cooper, Jr., "Time-Dependent-Dielectric-Breakdown Measurements of Thermal Oxides on N-Type 6H-SiC," *IEEE Trans. Elec. Dev.*, vol. 46:3, p 520 (1999).
- [26] M. Treu, R. Rupp, and G. Sölkner, "Reliability of SiC Power Devices and its Influence on their Commercialization – Review, Status, and Remaining Issues," *IEEE IRPS*, p. 156 (2010).
- [27] K. Matocha, G. Dunne, S. Soloviev, and R. Beaupre, "Time-Dependent Dielectric Breakdown of 4H-SiC MOS Capacitors and DMOSFETs," *IEEE Trans. Elec. Dev.*, vol. 55:8, p. 1830 (2008).
- [28] L. Yu, K.P. Cheung, J. Campbell, J.S. Suehle, and K. Sheng, "Oxide Reliability of SiC MOS Devices," *IEEE IIRW Final Report*, p. 141 (2008).
- [29] S.-H. Ryu, B. Hull, S. Dhar, L. Cheng, Q. Zhang, J. Richmond, M. Das, A. Agarwal, J. Palmour, A. Lelis, B. Geil, and C. Scozzie, "Performance, Reliability,"

and Robustness of 4H-SiC Power DMOSFETs," *Mat. Sci. Forum*, vols. 645-648, p. 969 (2010).

- [30] T. Kimoto, G. Feng, T. Hiyoshi, K. Kawahara, M. Noborio, and J. Suda, "Defect Control on Growth and Processing of 4H-SiC for Power Device Applications," opening invited plenary talk of the *International Conference on SiC and Related Materials (ICSCRM)*, Nuremberg, Germany (2009).
- [31] B.E. Deal, "Standardized Terminology for Oxide Charges Associated with Thermally Oxidized Silicon," *IEEE Trans. Elec. Dev.*, vol. 27, p. 606 (1980).
- [32] S. Potbhare, N. Goldsman, G. Pennington, A. Lelis, and J. McGarrity, "Numerical and experimental characterization of 4H-silicon carbide lateral metal-oxide-semiconductor field-effect transistor," *J. Appl. Phys.*, vol. 100, 044515 (2006).
- [33] S. Potbhare, N. Goldsman, G. Pennington, A. Lelis, and J. McGarrity, "A quasitwo-dimensional depth-dependent mobility model suitable for device simulation for Coulombic scattering due to interface trapped charges," *J. Appl. Phys.*, vol. 100, 044516 (2006).
- [34] N. S. Saks, S. S. Mani, A. K. Agarwal, and V. S. Hegde, "Hall Mobility of the Electron Inversion Layer in 6H-SiC MOSFETs," *Mater. Sci. Forum*, vols. 338-342, p. 737 (2000).
- [35] N. S. Saks "Hall Effect Studies of Electron Mobility and Trapping," from "Part III: Surface and Interface Properties" of <u>Silicon Carbide—Recent Major</u> <u>Advances</u>, ed. By W.J. Choyke, H. Matsunami, and G. Pensl, Springer, Berlin, p. 387 (2004).

- [36] A. Lelis, W. Tipton, S. Bayne, B. Geil, C. Scozzie, R. Stahlbush, N. Saks, and A. Hefner, "Government HPE Device Evaluation," *DARPA HPE Program Rev. Meeting*, Monterey, CA (May 2004).
- [37] A.J. Lelis, D. Habersat, G. Lopez, J.M. McGarrity, F.B. McLean, and N.
 Goldsman, "Bias Stress-Induced Threshold-Voltage Instability of SiC
 MOSFETs," *Mater. Sci. Forum*, vols. 527-529, p. 1317 (2006).
- [38] A.J. Lelis, D. Habersat, F. Olaniran, B. Simons, J.M. McGarrity, F.B. McLean, and N. Goldsman, "Time-Dependent Bias Stress-Induced Instability of SiC MOS Devices" *Mater. Res. Soc. Symp. Proc.*, vol. 911, 0911-B13-05 (2006).
- [39] A.J. Lelis, D. Habersat, R. Green, A. Ogunniyi, M. Gurfinkel, J. Suehle, and N.
 Goldsman, "Time Dependence of Bias-Stress Induced Threshold-Voltage
 Instability Measurements," *Proc. 2007 IEEE International Semiconductor Device Research Symposium* (2007).
- [40] M. Gurfinkel, J. Suehle, J.B. Bernstein, Y. Shapira, A.J. Lelis, D. Habersat, and N. Goldsman, "Ultra-Fast Characterization of Transient Gate Oxide Trapping in SiC MOSFETs," *Intl. Rel. Physics Symposium Proceedings*, vol. 45, p. 462 (2007).
- [41] M. Gurfinkel, H.D. Xiong, K.P. Cheung, J.S. Suehle, J.B. Bernstein, Y. Shapira,
 A.J. Lelis, D. Habersat, and N. Goldsman, "Characterization of Transient Gate
 Oxide Trapping in SiC MOSFETs Using Fast I–V Techniques," *IEEE Trans. Elec. Dev.*, vol. 55:8, p. 2004 (2008).

- [42] A.J. Lelis, D. Habersat, R. Green, A. Ogunniyi, M. Gurfinkel, J. Suehle, and N. Goldsman, "Effect of SiC Power DMOSFET Threshold-Voltage Instability"
 Mater. Res. Soc. Symp. Proc., vol. 1069, 1069-D11-04 (2008).
- [43] A.J. Lelis, R. Green, D. Habersat, and N. Goldsman, "Effect of Threshold-Voltage Instability on SiC DMOSFET Reliability," *IEEE IIRW Final Report*, p. 72 (2008).
- [44] T. Okayama, S.D, Arthur, J.L. Garrett, and M.V. Rao, "Bias-stress induced threshold voltage and drain current instability in 4H–SiC DMOSFETs," *Solid State Electronics*, vol. 52, p. 164 (2008).
- [45] A. Lelis, D. Habersat, R. Green, and N. Goldsman, "Temperature-Dependence of SiC MOSFET Thershold-Voltage Instability," *Mater. Sci. Forum*, vols. 600-603, p. 807 (2009).
- [46] M. J. Tadjer, K. D. Hobart, E. Imhoff, and F. J. Kub, "Temperature and Time Dependent Threshold Voltage Instability in 4H-SiC Power DMOSFET Devices," *Mater. Sci. Forum*, vols. 600-603, p. 1147 (2009).
- [47] A. Lelis, D. Habersat, R. Green, and N. Goldsman, "Implications of Threshold-Voltage Instability on SiC DMOSFET Operation," *Mater. Sci. Forum*, vols. 615-617, p. 809 (2009).
- [48] A. Lelis, R. Green, D. Habersat, and N. Goldsman, "Effect of ON-state Stress on SiC DMOSFET Subthreshold I-V Characteristics," *Mater. Sci. Forum*, vols. 645-648, p. 983 (2010).
- [49] M. Grieb, M. Noborio, D. Peters, A. J. Bauer, P. Friedrichs, T. Kimoto, and H.Ryssel, "Comparison of the threshold-voltage stability of SiC MOSFETs with

thermally grown and deposited gate oxides," *Mater. Sci. Forum*, vols. 645-648, p. 681 (2010).

- [50] A. Lelis, D. Habersat, R. Green, and N. Goldsman, "Time-Dependence of SiC MOSFET Subthreshold-Voltage Instability Measurements and Implications for Reliability Testing," presented at 2010 MRS Spring Meeting, San Francisco, CA (2010).
- [51] A.J. Lelis, R. Green, and D. Habersat, "High-Temperature Reliability of SiC Power MOSFETs," *Mater. Sci. Forum*, vols. 679-680, p. 599 (2011).
- [52] H. Yano, Y. Oshiro, D. Okamoto, T. Hatayama, and T. Fuyuki, "Instability of 4H-SiC MOSFET characteristics due to interface traps with long time constant," presented at 2010 ECSCRM, Oslo, Norway (2010).
- [53] R. Green, A. Lelis, and D. Habersat, "Application of Reliability Test Standards to SiC Power MOSFETs," presented *at 2011 Intl. Rel. Physics Symposium* (April 2011).
- [54] K.-C. Chang, J. Bentley, and L.M. Porter, "Nanoscale characterization of the silicon dioxide–silicon carbide interface using elemental mapping by energyfiltered transmission electron microscopy," *J. Electron. Mater.*, vol. 32:5, p. 464 (2003).
- [55] T. Zheleva, A. Lelis, U. Lee, G. Duscher, F. Liu, M. Das, and J.
 Scofield, "Transition layers at the SiO₂/SiC interface," presented at the 2007 Intl.
 Conf. Silicon Carbide and Related Materials (ICSCRM), Otsu, Japan (2007).
- [56] T. Zheleva, A. Lelis, G. Duscher, F. Liu, I. Levin, and M. Das, "Transition layers at the SiO₂/SiC interface," *Appl. Phys. Lett.*, vol. 93, 022108 (2008).

- [57] S. T. Pantelides, G. Duscher, M. Di Ventra, R. Buczko, K. McDonald, M.B. Huang, R.A. Weller, I. Baumvol, F.C. Stedile, C. Radtke, S.J. Pennycook, G. Chung, C.C. Tin, J.R. Williams, J.H. Won, and L.C. Feldman, "Atomic-scale engineering of the SiC–SiO₂ interface," *Mater. Sci. Forum*, vols. 338–342, p. 1133 (2000).
- [58] S.A. Corrêa, C. Radtke, G.V. Soares, L. Miotti, I.J.R. Baumvol, S. Dimitrijev, J. Han,L. Hold, F. Kong, and F. C. Stedile, "Effects of nitrogen incorporation on the interfacial layer between thermally grown dielectric films and SiC," *Appl. Phys. Lett.*, vol. 94, 251909 (2009).
- [59] K.C. Chang, N.T. Nuhfer, L.M. Porter, and Q. Wahab," High-carbon concentrations at the silicon dioxide–silicon carbide interface identified by electron energy loss spectroscopy", *Appl. Phys. Lett.*, vol. 77:14, p. 2186 (2000).
- [60] M.S. Dautrich, P.M. Lenahan, and A.J. Lelis, "Identification of trapping defects in 4H-silicon carbide metal-insulatorsemiconductor field-effect transistors by electrically detected magnetic resonance," *Appl. Phys. Lett.*, vol. 89, 223502 (2006).
- [61] C. J. Cochrane, P. M. Lenahan, and A. J. Lelis, "An electrically detected magnetic resonance study of performance limiting defects in SiC metal oxide semiconductor field effect transistors," J. Appl. Phys., vol. 109, 014506 (2011).
- [62] D.J. Meyer, P.M. Lenahan, and A.J. Lelis, "Observation of trapping defects in 4H-silicon carbide metal-oxide-semiconductor field-effect transistors by spindependent recombination," *Appl. Phys. Lett.*, vol. 86, 023503 (2005).

- [63] R. Buczko, S. J. Pennycook, and S. T. Pantelides, "Bonding Arrangements at the Si-SiO₂ and SiC-SiO₂ Interfaces and a Possible Origin of their Contrasting Properties" *Phys. Rev. Lett.*, vol. 84:5, p. 943 (2000).
- [64] G.G. Jernigan, R.E. Stahlbush, M.K. Das, J.A. Cooper, Jr., and L.A. Lipkin,
 "Interfacial differences between SiO₂ grown on 6H-SiC and on Si(100)," *Appl. Phys. Lett.*, vol. 74:10, p 1448 (1999).
- [65] F.J. Feigl, W.B. Fowler, and K.L.Yip, "Oxygen vacancy model for the E_1 ' center in SiO₂," *Sol. State Comm.*, vol. 14:3, p. 225 (1974).
- [66] J. Rozen, S. Dhar, M.E. Zvanut, J.R. Williams, and L.C. Feldman, "Density of interface states, electron traps, and hole traps as a function of the nitrogen density in SiO₂ on SiC," J. Appl. Phys., vol. 105, 124506 (2009).
- [67] S.K. Powell, N. Goldsman, J.M. McGarrity, J. Bernstein, C.J. Scozzie, and A. Lelis, "Physics-based numerical modeling and characterization of 6H-silicon-carbon metal–oxide–semiconductor field-effect transistors," *J. Appl. Phys.*, vol. 92, p 4053 (2002).
- [68] V.V. Afanas'ev, F. Ciobanu, G. Pensl, and A. Stesmans, "Contributions to the Density of Interface States in SiC MOS Structures," from <u>Silicon Carbide</u> <u>Recent Major Advances</u>, ed. By W.J. Choyke, H. Matsunami, and G. Pensl, Springer, Berlin, p. 343 (2004).
- [69] H. Li, S. Dimitrijev, H. B. Harrison, and D. Sweatman, "Interfacial characteristics of N₂O and NO nitrided SiO₂ grown on SiC by rapid thermal processing," *Appl. Phys. Lett.*, vol. 70, p. 2028 (1997).

- [70] G. Y. Chung, C. C. Tin, J. R. Williams, K. McDonald, M. Di Ventra, S. T.
 Pantelides, L. C. Feldman, and R. A. Weller, "Effect of nitric oxide annealing on the interface trap densities near the band edges in the 4H polytype of silicon carbide," *Appl. Phys. Lett.*, vol. 76, p. 1713 (2000).
- [71] L. A. Lipkin, M. K. Das, and J. W. Palmour, "N₂O Processing Improves the 4H-SiC:SiO₂ Interface" *Mater. Sci. Forum*, vols. 389-393, p. 985 (2002).
- [72] D. Habersat, A.J. Lelis, G. Lopez, J. McGarrity, and F.B. McLean, "On Separating Oxide Charges and Interface Charges in 4H-SiC Metal-Oxide-Semiconductor Devices" *Mater. Sci. Forum*, vols. 527-529, p. 1007 (2006).
- [73] E.O. Sveinbjornsson, G. Gudjonsson, F. Allerstam, H.O. Olafsson, P-A. Nilsson,
 H. Zirath, T. Rodle, and R. Jos "High channel mobility 4H-SiC MOSFETs," *Mater. Sci. Forum*, vols. 527-529, p. 961 (2006).
- [74] M.K. Das, B.A. Hull, S. Krishnaswami, F. Husna, S. Haney, A. Lelis, C.J.
 Scozzie, and J.D. Scofield, "Improved 4H-SiC MOS Interfaces Produced via Two
 Independent Processes: Metal Enhanced Oxidation and 1300oC NO Anneal,"
 Mater. Sci. Forum, vols. 527-529, p. 967 (2006).
- [75] A. Kerber, E. Cartier, L. Pantisano, R. Degraeve, T. Kauerauf, Y. Kim, A. Hou,
 G. Groeseneken, H.E. Maes, and U. Schwalke, "Origin of the Threshold voltage instability in SiO₂/HfO₂ Dual Layer Gate Dielectrics," *IEEE Elec. Dev. Lett.*, vol. 24, p. 87 (2003).
- [76] C. Shen, M.-F. Li, X.P. Wang, Y.-C. Yeo, and D.-L. Kwong, "A Fast Measurement Technique of MOSFET I_D-V_G Characteristics," *IEEE Elec. Dev. Lett.*, vol. 27:1, p. 55 (2006).

- [77] V.V. Afanas'ev, A. Stesmans, F. Ciobanu, G. Pensl, K.Y. Cheong, and S.
 Dimitrijev, "Mechanisms responsible for improvement of 4H–SiC/SiO₂ interface properties by nitridation," *Appl. Phys. Lett.*, vol. 82:4, p. 568 (2003).
- [78] F.C. J. Kong, S. Dimitrijev, and J. Han, "Electrical Characteristics of Near-Interface Traps in 3C-SiC Metal–Oxide–Semiconductor Capacitors," *IEEE Elec. Dev. Lett.*, vol. 29:9, p. 1021 (2008).
- [79] T.L. Tewksbury and H.-S. Lee, "Characterization, Mcdeling, and Minimization of Transient Threshold Voltage Shifts in MOSFET' s," *IEEE J. Sol.-State Circ.*, vol. 29:3, p. 239 (1994).
- [80] A.F. Basile, J. Rozen, X.D. Chen, S. Dhar, J.R. Williams, L.C. Feldman, and P.M.
 Mooney, "Passivation of Deep Levels at the SiO₂/SiC Interface," *Elec. Chem. Soc. Trans.*, vol. 28:4, p. 95 (2010).
- [81] A. Agarwal and S. Haney, "Some Critical Materials and Processing Issues in SiC Power Devices," *J. Elec. Mater.*, vol. 37:5, p. 646 (2008).
- [82] T. Ohshima, H. Itoh, and M. Yoshikawa, "Effect of gamma-ray irradiation on the characteristics of 6H silicon carbide metal-oxide-semiconductor field effect transistor with hydrogen-annealed gate oxide," *J. Appl. Phys.*, vol. 90:6, p. 3038 (2001).
- [83] K.K. Lee, T. Ohshima, and H. Itoh, "Performance of Gamma Irradiated P-Channel 6H-SiC MOSFETs: High Total Dose," *IEEE Trans. Nucl. Sci.*, vol. 50:1, p. 194 (2003).

- [84] A.C. Ahyi, S.R. Wang, and J.R. Williams "Gamma irradiation effects on 4H-SiC MOS capacitors and MOSFETs," *Mater. Sci. Forum*, vols. 527-529, p. 1063 (2006).
- [85] Brad Bittel, Penn State Univ., private communication.
- [86] J.R. Schwank, P.S. Winokur, P.J. McWhorter, F.W. Sexton, P.V. Dressendorfer, and D.C. Turpin, "Physical Mechanisms Contributing to Device "Rebound"," *IEEE Trans. Nucl. Sci.*, vol. 31:6, p. 1434 (1984).
- [87] C.M. Dozier, D.B. Brown, J.L. Throckmorton, and D.I. Ma, "Defect Production in SiO₂ by X-Ray and Co-60 Radiations," *IEEE Trans. Nucl. Sci.*, vol. 32:6, p. 4363 (1985).
- [88] A.J. Lelis, H.E. Boesch, Jr., T.R. Oldham, and F.B. McLean, *IEEE Trans. Nucl. Sci.*, vol. 35:6, p. 1186 (1988).
- [89] A.J. Lelis, T.R. Oldham, H.E. Boesch, Jr., and F.B. McLean, "The Nature of the Trapped Hole Annealing Process," *IEEE Trans. Nucl. Sci.*, vol. 36:6, p. 1808 (1989).
- [90] A.J. Lelis and T.R. Oldham, "Time Dependence of Switching Oxide Traps," *IEEE Trans. Nucl. Sci.*, vol. 41:6, p. 1835 (1994).
- [91] F.B. McLean and T.R. Oldham, "Basic Mechanisms of Radiation Effects in Electronic Materials and Devices," *Harry Diamond Laboratories HDL-TR-2129* (September 1987).
- [92] T.R. Oldham "Ionizing Radiation Effects in MOS Oxides," World Scientific, London (1999).

- [93] P.M. Lenahan and P.V. Dressendorfer, "Hole traps and trivalent silicon centers in metal/oxide/silicon devices," J. Appl. Phys., vol. 55, p. 3495 (1984).
- [94] C.L. Marquardt and G.H. Sigel, Jr., "Radiation-Induced Defect Centers in Thermally Grown Oxide Films," *IEEE Trans. Nucl. Sci.*, vol. 22, p. 2234 (1975).
- [95] B. E. Deal, M. Sklar, A. S. Grove, and E. H. Snow, "Characteristics of the Surface-State Charge (Q_{SS}) of Thermally Oxidized Silicon," *J. Electrochem. Soc.: Solid State Sci.*, p. 266 (March 1967).
- [96] J.T. Ryan, P.M. Lenahan, T. Grasser, and H. Enichlmair, "Observations of negative bias temperature instability defect generation via on the fly electron spin resonance," *Appl. Phys. Lett.*, vol. 96, 223509 (2010).
- [97] J.F. Conley, Jr., P.M. Lenahan, A.J. Lelis, and T.R. Oldham, "Electron spin resonance evidence for the structure of a switching oxide trap: Long term structural change at silicon dangling bond sites in SiO₂," *Appl. Phys. Lett.*, vol. 67, p. 2179 (1995).
- [98] J.F. Conley, Jr., P.M. Lenahan, A.J. Lelis, and T.R. Oldham, "Electron Spin Resonance Evidence that E'_{γ} Centers Can Behave as Switching Oxide Traps," *IEEE Trans. Nucl. Sci.*, vol. 42:6, p. 1744 (1995).
- [99] T. Grasser, B. Kaczer, W. Goes, T. Aichinger, P. Hehenberger, and M. Nelhiebel,
 "A Two-Stage Model for Negative Bias Temperature Instability," *IEEE 47th Intl. Rel. Phys. Symp.*, p. 33 (2009).
- [100] M. Walters and A. Reisman, Radiation-Induced Neutral Electron Trap Generation in Electrically Biased Insulated Gate Field Effect Transistor Gate Insulators, J. *Electrochem. Soc.*, vol. 138, p. 2756 (1991).

- [101] F.J. Grunthaner, P.J. Grunthaner, and J. Maserjian, "Radiation-Induced Defects in SiO₂ as Determined with XPS," *IEEE Trans. Nucl. Sci.*, vol. 29:6, p. 1462 (1982).
- [102] S.M. Sze, *Physics of Semiconductor Devices*, 2nd ed. Wiley, New York, Chap. 7 (1981).
- [103] B.G. Streetman and S.K. Banerjee, *Solid State Electronic Devices*, 6th ed.Prentice Hall, Upper Saddle River, New Jersey, Chap. 6 (2006).
- [104] S. Potbhare, N. Goldsman, A. Akturk, M. Gurfinkel, A. Lelis, and J.S. Suehle,
 "Energy- and Time-Dependent Dynamics of Trap Occupation in 4H-SiC
 MOSFETs," *IEEE Trans. Elec. Dev.*, vol. 55:8, p. 2061 (2008).
- [105] D.B. Habersat, A.J. Lelis, S. Potbhare, and N. Goldsman, "Improvements In SiC MOS Processing as Revealed by Studies of Fixed and Oxide Trap Charge,"
 Mater. Sci. Forum, vols. 615-617, p. 769 (2009).
- [106] M. Gurfinkel, S. Potbhare, H.D. Xiong, J.S. Suehle, Y. Shapira, A.J. Lelis, D. Habersat, and N. Goldsman, "Ion implantation and SiC transistor performance," *J. Appl. Phys.*, vol. 105, 084511 (2009).
- [107] J. T. Yount, P. M. Lenahan, and J. T. Krick, "Comparison of defect structure in N₂O- and NH₃-nitrided oxide dielectrics," *J. Appl. Phys.*, vol. 76:3, p. 1754 (1994).
- [108] D.B. Habersat, A.J. Lelis, J.M. McGarrity, F.B. McLean, and S. Potbhare, "The effect of nitridation on SiC MOS oxides as evaluated by charge pumping," *Mater. Sci. Forum*, vols. 600-603, p. 743 (2009).
- [109] D. Habersat and A.J. Lelis, "Improved Observation of SiC/SiO₂ Oxide Charge Traps Using MOS *C-V*," *Mater. Sci. Forum*, vols. 679-680, p. 366 (2011).

- [110] P.J. McWhorter, S.L. Miller, and W.M. Miller, "Modeling the Anneal of Radiation-Induced Trapped Holes in a Varying Thermal Environment," *IEEE Trans. Nucl. Sci.*, vol. 37:6, p. 1682 (1990).
- [111] S. Dimitrijev, "SiC as a Material for Mainstream Electronics," tutorial presentation at *2009 ICSCRM*, Nuremberg, Germany (2009).
- [112] T. R. Oldham, A. J. Lelis, and F. B. McLean, "Spatial dependence of trapped holes determined from tunneling analysis and measured annealing," *IEEE Trans. Nucl. Sci.*, vol. 33:6, p. 1203 (1986).
- [113] F. B. McLean, "A direct tunneling model of charge transfer at the insulator– semiconductor interface in MIS devices," *Harry Diamond Lab., Adelphi, MD, HDL-TR-1765* (1976).
- [114] C. C. Chang and W. C. Johnson, "Frequency and Temperature Tests for Lateral Nonuniformities in MIS Capacitors," *IEEE Trans. Elec. Dev.*, vol. 24:10, p. 1249 (1977).
- [115] J.M. Benedetto and H.E. Boesch, Jr., "MOSFET and MOS Capacitor Responses to Ionizing Radiation," *IEEE Trans. Nucl. Sci.*, vol. 31:6, p. 1461 (1984).
- [116] P.J. McWhorter and P.S. Winokur, "Simple technique for separating the effects of interface traps and trapped oxide charge in metal-oxide-semiconductor transistors," *Appl. Phys. Lett.*, vol. 48, p. 133 (1986).
- [117] S.M. Sze, *Physics of Semiconductor Devices*, 2nd ed. Wiley, New York, Chap. 8, pp. 446-7 (1981).

- [118] S. Potbhare, N. Goldsman, A. Lelis, J.M. McGarrity, F.B. McLean, and D.
 Habersat, "A Physical Model of High Temperature 4H-SiC MOSFETs," *IEEE Trans. Elec. Dev.*, vol. 55:8, p. 2029 (2008).
- [119] X.C.Ma, Q.Q.Shu, S.Meng, W.G.Ma, "Image force effects on trapezoidal barrier parameters in metal-insulator-metal tunnel junctions," *Thin Solid Films*, vol. 436, p. 292 (2003).
- [120] S. Manzini and A. Modelli, "Tunneling Discharge of Trapped Holes in Silicon Dioxide," from "Insulating Films onSemiconductors," J. F. Verweij and D. R.
 Wolters, editors, Elsevier Science Publishers B.V. (North Holland), p.112 (1983).
- [121] A. Schenk and G. Heiser, "Modeling and simulation of tunneling through ultrathin gate dielectrics," J. Appl. Phys., vol. 81:12, p. 7900 (1997).
- [122] Y.T. Hou, M.F. Li, Y. Jin, and W.H. Lai, "Direct tunneling hole currents through ultrathin gate oxides in metal-oxide-semiconductor devices," *J. Appl. Phys.*, vol. 91:1, p. 258 (2002).
- [123] C.J. Nicklaw, Z.-Y. Lu, D.M. Fleetwood, R. D. Schrimpf, and S.T. Pantelides,
 "The Structure, Properties, and Dynamics of Oxygen Vacancies in Amorphous SiO₂," *IEEE Trans. Nucl. Sci.*, vol. 49:6, p. 2667 (2002).
- [124] W. Goes, M. Karner, V. Sverdlov, and T. Grasser, "Charging and Discharging of Oxide Defects in Reliability Issues," *IEEE Trans. Dev. and Mater. Rel.*, vol. 8:3, p. 491 (2008).
- [125] M.K. Das, R. Callanan, D.C. Capell, B. Hull, F. Husna, J. Richmond, M.O'loughlin, M.J. Paisley, A. Powell, and Q. Zhang, "State of the Art 10 kV

NMOS Transistors," *IEEE 20th Int. Symp. On Power Semiconductor Devices and IC's*, p. 253 (2008).

- [126] S. Krishnaswami, S.-H. Ryu, B. Heath, A. Agarwal, J. Palmour, B. Geil, A. Lelis, and C. Scozzie, "A Study on the Reliability and Stability of High Voltage 4H-SiC MOSFET Devices," *Mater. Sci. Forum*, vols. 527-529, p. 1313 (2006).
- [127] T. E. Salem, D. Ibitayo, and B. R. Geil, "Validation of Infrared Camera Thermal Measurements on High-Voltage Power Electronic Components," *IEEE Trans. Instrumentation and Measurement*, vol. 56:5, p. 1973 (2007).
- [128] C.J. Scozzie and J.M. McGarrity, "Charge Pumping Measurements on SiC MOSFETs," *Mater. Sci. Forum*, vols. 264-268, p. 985 (1998).
- [129] "Temperature, Bias, and Operating Life Standard," JESD22-A108C (2005).
- [130] "Test Methods for Semiconductor Devices," *MIL-STD-750E* (2006).
- [131] "Stress Test Qualification for Automotive Grade Discrete Semiconductors," AEC-Q101-Rev-C (2005).